# <u>Time series modelling of Gompertz-Makeham</u> <u>mortality curves: historical analysis, forecasting</u> <u>and life insurance applications</u>

### Oliver Lockwood March 2009

## <u>Abstract</u>

The Continuous Mortality Investigation (CMI) of the Institute and Faculty of Actuaries has for a number of years based its graduated tables of assured life, annuitant and pensioner mortality on Gompertz-Makeham formulae. In this thesis, we consider two-dimensional data sets consisting of the number of deaths and the exposed to risk at a range of ages in a range of calendar years. Having fitted a Gompertz-Makeham model to the data for each calendar year, we fit univariate time series models to represent the behaviour over time of the Gompertz-Makeham parameters. Cohort effects are allowed for by applying a multiplicative factor depending on year of birth to the fitted force of mortality. Prediction intervals for the future parameters of the model are calculated. Sample values of immediate and deferred annuities are presented, based on stochastic mortality simulations and a deterministic interest rate. An application to risk-based capital calculations, under the Individual Capital Assessment (ICA) regime of the Financial Services Authority (FSA), is presented.

### Key words

Mortality; Gompertz-Makeham Models; Time Series; Cohort Effects; Prediction Intervals; Annuities; Individual Capital Assessment

# **Contents**

| 1  | Introduction   | 11   |
|--|--|--|
| <b>2</b><br>2.1<br>2.2<br>2.3<br>2.3.1<br>2.3.2<br>2.3.3<br>2.4<br>2.5<br>2.6<br>2.7<br>2.8<br>2.9 | Fitting a Gompertz-Makeham model for each calendar year to CMI male assured lives data and to England and Wales population data<br>Model<br>Parameter estimation methodology<br>Parameter estimation results<br>CMI data<br>England and Wales male data<br>England and Wales female data<br>Results for the force of mortality<br>Akaike Information Criterion (AIC) and Bayesian Information Criterion (BIC)<br>Standardised residuals<br>Coefficient of determination, $R^2$<br>Residual plots<br>Conclusion | 24<br>24<br>26<br>26<br>27<br>28<br>29<br>29<br>32<br>34<br>36<br>41 |
| <b>3</b><br>3.1<br>3.2<br>3.3<br>3.4<br>3.5<br>3.6<br>3.7  | Adding a cohort effect to the modelIntroductionRestricted data setsγ parameters (cohort parameters)AIC and BICStandardised residualsResidual plotsConclusion   | 68<br>68<br>69<br>70<br>71<br>71<br>72                               |
| <b>4</b><br>4.1<br>4.2<br>4.3<br>4.4<br>4.5<br>4.6<br>4.7<br>4.8                                   | Fitting time series models to the parameter estimates<br>Introduction<br>$\kappa^{(0)}$ parameters<br>$\kappa^{(3)}$ parameters<br>$\kappa^{(4)}$ parameters<br>$\kappa^{(5)}$ parameters<br>$\gamma$ parameters (cohort parameters)<br>Correlations<br>Conclusion   | 83<br>83<br>84<br>85<br>87<br>88<br>90<br>90                         |
| <b>5</b><br>5.1<br>5.2<br>5.3<br>5.4<br>5.5<br>5.6<br>5.7  | Forecasting and life insurance applications<br>Introduction<br>Prediction intervals for the future parameter values<br>Prediction intervals for projected forces of mortality<br>Ages above the highest age in the data<br>Annuity values<br>Individual Capital Assessments (ICAs)<br>Conclusion   | <b>106</b><br>106<br>107<br>109<br>111<br>117<br>127                 |
| 6  | Conclusions  | 190  |
|  | References   | 192  |
|  | Appendices:  |  |
| Α  | Iterative scheme for estimating the parameters of Gompertz-Makeham models  | 193  |
| В  | Investigations into estimating both the kappa and gamma parameters by a single iterative procedure   | 196  |
| С  | Time series analysis   | 200  |

| C.1 | First-order autoregressive and moving average processes | 200 |
|-----|---|-----|
| C.2 | Sample autocorrelation coefficient                      | 201 |
| C.3 | Prediction intervals                                    | 201 |
| D   | The value of an annuity after one year                  | 203 |

# List of figures

| 1.1<br>1.2<br>1.3 | Logarithms of the exposed to risk for the three data sets<br>Logarithms of the crude force of mortality for the three data sets<br>Fitted mortality curves under various GM models, with exaggeration to | 18<br>19 |
|-------------------|--|----------|
| 1.4               | highlight certain features<br>CMI crude force of mortality as a function of age for various calendar years   | 20<br>21 |
| 1.5               | England and Wales male crude force of mortality as a function of age for various calendar vears  | 22       |
| 1.6               | England and Wales female crude force of mortality as a function of age for various calendar years  | 23       |
| 2.1               | Maximum likelihood parameter estimates for the GM(0,2) model fitted to CMI data  | 42       |
| 2.2               | Maximum likelihood parameter estimates for the GM(1,2) model fitted to CMI data  | 43       |
| 2.3               | Maximum likelihood parameter estimates for the GM(1,3) model fitted to CMI data  | 44       |
| 2.4               | Maximum likelihood parameter estimates for the GM(2,3) model fitted to CMI data  | 45       |
| 2.5               | Maximum likelihood parameter estimates for the GM(3,3) model fitted to CMI data  | 46       |
| 2.6               | Maximum likelihood parameter estimates for the GM(2,4) model fitted to CMI data  | 47       |
| 2.7               | Maximum likelihood parameter estimates for the GM(0,2) model fitted to<br>England and Wales male data  | 48       |
| 2.8               | Maximum likelihood parameter estimates for the GM(1,2) model fitted to<br>England and Wales male data  | 49       |
| 2.9               | Maximum likelihood parameter estimates for the GM(1,3) model fitted to<br>England and Wales male data  | 50       |
| 2.10              | Maximum likelihood parameter estimates for the GM(2,3) model fitted to<br>England and Wales male data  | 51       |
| 2.11              | Maximum likelihood parameter estimates for the GM(0,2) model fitted to<br>England and Wales female data  | 52       |
| 2.12              | Maximum likelihood parameter estimates for the GM(1,2) model fitted to<br>England and Wales female data  | 53       |
| 2.13              | Maximum likelihood parameter estimates for the GM(1,3) model fitted to<br>England and Wales female data  | 54       |
| 2.14              | Maximum likelihood parameter estimates for the GM(2,3) model fitted to<br>England and Wales female data  | 55       |
| 2.15              | 2000 mortality curves fitted to CMI data – differences from the GM(0,2) model  | 56       |
| 2.16              | 2000 mortality curves fitted to England and Wales male data – differences from the GM(0,2) model   | 57       |
| 2.17              | 2000 mortality curves fitted to England and Wales female data –<br>differences from the GM(0,2) model  | 58       |
| 2.18              | Plots of standardised residuals for a number of possible GM models fitted to CMI data  | 59       |
| 2.19              | Plots of standardised residuals for a number of possible GM models fitted to England and Wales male data   | 60       |
| 2.20              | Plots of standardised residuals for a number of possible GM models fitted to England and Wales female data   | 61       |
| 2.21              | Scatter diagrams of standardised residuals plotted against age for CMI data  | 62       |
| 2.22              | Scatter diagrams of standardised residuals plotted against year of birth for CMI data  | 63       |
| 2.23              | Scatter diagrams of standardised residuals plotted against age for England and Wales male data   | 64       |
| 2.24              | Scatter diagrams of standardised residuals plotted against year of birth for<br>England and Wales male data  | 65       |

| 2.25       | Scatter diagrams of standardised residuals plotted against age for England and Wales female data   | 66  |
|------------|--|-----|
| 2.26       | Scatter diagrams of standardised residuals plotted against year of birth for<br>England and Wales female data  | 67  |
| 3.1        | Maximum likelihood parameter estimates for the GM(1,3) model fitted to the restricted set of CMI data, before the introduction of gamma parameters   | 73  |
| 3.2        | Maximum likelihood parameter estimates for the GM(1,3) model fitted to<br>the restricted set of England and Wales male data, before the introduction<br>of gamma parameters                                      | 74  |
| 3.3        | Maximum likelihood parameter estimates for the GM(1,3) model fitted to<br>the restricted set of England and Wales female data, before the  |     |
| 3.4        | Introduction of gamma parameters<br>Plots of standardised residuals for the GM(1,3) model fitted to the restricted   | 75  |
| 3.5        | data sets, before the introduction of gamma parameters<br>Scatter diagrams of standardised residuals plotted against age for the<br>GM(1,3) model fitted to the restricted data sets, before the introduction of | 76  |
| 3.6        | gamma parameters<br>Scatter diagrams of standardised residuals plotted against year of birth for<br>the GM(1.3) model fitted to the restricted data sets, before the introduction                                | 77  |
| 3.7        | of gamma parameters<br>Estimates of the gamma parameters (or equivalently, A/Es for each year of   | 78  |
| 3.8        | birth)<br>Plots of standardised residuals for the GM(1,3) model extended to  | 79  |
| 3.9        | Scatter diagrams of standardised residuals plotted against age for the   | 80  |
| 3.10       | Scatter diagrams of standardised residuals plotted against year of birth for<br>the GM(1.3) model extended to incorporate a cohort effect  | 81  |
| 4 1        | Sample autocorrelation functions of the $\kappa^{(0)}$ parameters  | 01  |
| 4.2        | Residuals of the AR(1) processes fitted to the $\kappa^{(0)}$ series   | 92  |
| 4.3        | Sample autocorrelation functions of the residuals in Figure 4.2  | 93  |
| 4.4        | Differences between the $\kappa^{(3)}$ parameters for successive years   | 94  |
| 4.5        | Sample autocorrelation functions of the differences in Figure 4.4  | 95  |
| 4.6        | Residuals of the MA(1) processes fitted to the differences in Figure 4.4   | 96  |
| 4.7        | Sample autocorrelation functions of the residuals in Figure 4.6  | 97  |
| 4.8        | Residuals of the AR(1) processes fitted to the $\kappa^{(4)}$ series   | 98  |
| 4.9        | Sample autocorrelation functions of the residuals in Figure 4.8  | 99  |
| 4.10       | Residuals of the AR(1) processes fitted to the $\kappa^{(3)}$ series   | 100 |
| 4.11       | Sample autocorrelation functions of the residuals in Figure 4.10   | 101 |
| 4.1Z       | Logarithms of the $\Lambda P(1)$ processes fitted to the logged v series   | 102 |
| 4.13       | Sample autocorrelation functions of the residuals in Figure 4.13   | 103 |
| 5.1        | Projection of the future $\kappa^{(0)}_{(3)}$ parameters   | 151 |
| 5.2        | Projection of the future $\kappa^{(3)}_{(4)}$ parameters   | 152 |
| 5.3        | Projection of the future $\kappa^{(5)}$ parameters   | 153 |
| 5.4<br>5.5 | Projection of the future k parameters  | 154 |
| 5.5<br>5.6 | Extensions of the graphs in Figure 1.2 to future years, using a deterministic  | 100 |
| 5.0        | projection with all the future inpovation terms set to zero  | 156 |
| 5.7        | Projected forces of mortality for CMI data compared with previously published projection bases – age (i) 30. (ii) 50. (iii) 70. (iv) 90.   | 157 |
| 5.8        | Projected forces of mortality for England and Wales male data compared<br>with previously published projection bases – age (i) 30 (ii) 50 (iii) 70 (iv)  | 157 |
|            | 89   | 158 |
| 5.9        | Projected forces of mortality for England and Wales female data compared with previously published projection bases – age (i) 30, (ii) 50, (iii) 70, (iv)  |     |
|            | 89   | 159 |

| 5.10 | Extrapolation of historical mortality to ages above 90 (89 for England and Wales data) using the log-linear extrapolation method, before the |                  |
|------|--|------------------|
|      | introduction of gamma parameters   | 160              |
| 5.11 | Extrapolation of historical mortality to ages above 90 (89 for England and   |                  |
|      | Wales data) using the LifeMetrics extrapolation method, before the   |                  |
|      | introduction of gamma parameters   | 161              |
| 5.12 | Comparison of methods of extrapolating the 2005 mortality curve to ages  |                  |
|      | above 90 (89 for England and Wales data), before the introduction of   |                  |
|      | gamma parameters   | 162              |
| 5 13 | Projected forces of mortality for CMI data under the low improvement   |                  |
| 0.10 | assumption compared with previously published projection bases – age in  |                  |
|      | 2005 (i) $35$ (ii) $45$ (iii) $55$ (iv) $65$ (v) $75$  | 163              |
| E 14 | Droinstad farage of martality for CML data under the high improvement  | 105              |
| 5.14 | Projected forces of montality for Civil data under the high improvement  |                  |
|      |  | 404              |
|      | 2005 (I) 35, (II) 45, (III) 55, (IV) 65, (V) 75  | 164              |
| 5.15 | Projected forces of mortality for England and Wales male data under the  |                  |
|      | low improvement assumption compared with GAD projection bases – age  |                  |
|      | in 2005 (i) 35, (ii) 45, (iii) 55, (iv) 65, (v) 75   | 165              |
| 5.16 | Projected forces of mortality for England and Wales male data under the  |                  |
|      | high improvement assumption compared with GAD projection bases – age   |                  |
|      | in 2005 (i) 35, (ii) 45, (iii) 55, (iv) 65, (v) 75   | 166              |
| 5.17 | Projected forces of mortality for England and Wales female data under the  |                  |
| -    | low improvement assumption compared with GAD projection bases – age  |                  |
|      | in 2005 (i) 35 (ii) 45 (iii) 55 (iv) 65 (v) 75   | 167              |
| 5 18 | Projected forces of mortality for England and Wales female data under the  | 107              |
| 5.10 | high improvement assumption compared with GAD projection bases and   |                  |
|      |  | 100              |
| = 10 | In 2005 (1) 35, (11) 45, (11) 55, (1V) 65, (V) 75  | 168              |
| 5.19 | Projected probabilities of survival for CMI data under the low improvement   |                  |
|      | assumption compared with previously published projection bases   | 169              |
| 5.20 | Projected probabilities of survival for CMI data under the high improvement  |                  |
|      | assumption compared with previously published projection bases   | 170              |
| 5.21 | Projected probabilities of survival for England and Wales male data under  |                  |
|      | the low improvement assumption compared with GAD projection bases  | 171              |
| 5.22 | Projected probabilities of survival for England and Wales male data under  |                  |
|      | the high improvement assumption compared with GAD projection bases   | 172              |
| 5.23 | Projected probabilities of survival for England and Wales female data under  |                  |
| 0.20 | the low improvement assumption compared with GAD projection bases  | 173              |
| 5 24 | Projected probabilities of survival for England and Wales female data under  | 170              |
| 0.24 | the high improvement assumption compared with GAD projection bases   | 174              |
| E 0E | Breighted future sizes of a fund act up in 2005 agual to the mean annuity  | 1/4              |
| 5.25 | Projected future sizes of a fund set up in 2005 equal to the mean annuity  | 475              |
| - 00 | Value plus ICA capital for CMI data  | 175              |
| 5.26 | Projected future sizes of a fund set up in 2005 equal to the mean annuity  |                  |
|      | value plus ICA capital for England and Wales male data   | 176              |
| 5.27 | Projected future sizes of a fund set up in 2005 equal to the mean annuity  |                  |
|      | value plus ICA capital for England and Wales female data   | 177              |
| 5.28 | Scatter diagrams to illustrate the correlation between the values of $C(x)$ at   |                  |
|      | different ages x in 2005 for CMI data  | 178              |
| 5.29 | Scatter diagrams to illustrate the correlation between the values of $C(x)$ at   |                  |
|      | different ages x in 2005 for England and Wales male data   | 180              |
| 5 30 | Scatter diagrams to illustrate the correlation between the values of $C(x)$ at   |                  |
| 5.50 | different ages v in 2005 for England and Wales female data   | 182              |
| E 21 | Illustration of the accuracy of Approximation 1 for CMI data   | 102              |
| 5.31 | Illustration of the accuracy of Approximation 1 for England and Wales male   | 104              |
| 5.32 |  | 405              |
| - 00 |  | 185              |
| 5.33 | illustration of the accuracy of Approximation 1 for England and Wales  | 4 <del>-</del> - |
|      | temale data  | 186              |
| 5.34 | Illustration of the accuracy of Approximation 2 for CMI data   | 187              |
| 5.35 | Illustration of the accuracy of Approximation 2 for England and Wales male   |                  |
|      | data   | 188              |

| 5.36 | Illustration of the accuracy of Approximation 2 for England and Wales female data   | 189 |
|------|---|-----|
| B.1  | Maximum likelihood parameter estimates for the GM(1,3) model extended to incorporate a cohort effect with the kappa parameters re-estimated by Method A – England and Wales male data | 108 |
| B.2  | Maximum likelihood parameter estimates for the GM(1,3) model extended<br>to incorporate a cohort effect with the kappa parameters re-estimated by                                     | 190 |
|      | Method B – England and Wales male data  | 199 |

# List of tables

| 2.1  | Starting points for the iterations to estimate the parameters of the GM  | 25                               |
|--|--|----------------------------------|
| 2.2<br>2.3   | Fitted values of the parameters <i>a</i> and <i>b</i> for the static GM(0,2) model AIC and BIC values for a number of possible GM models fitted to CMI data  | 20<br>30<br>31                   |
| 2.4  | AIC and BIC values for a number of possible GM models fitted to England<br>and Wales male data   | 31                               |
| 2.5  | and Wales female data  | 32                               |
| 2.6  | Sample variances of the standardised residuals for a number of possible GM models fitted to CMI data   | 33                               |
| 2.7  | Sample variances of the standardised residuals for a number of possible GM models fitted to England and Wales male data  | 33                               |
| 2.8  | Sample variances of the standardised residuals for a number of possible GM models fitted to England and Wales female data Values of $P^2$ for a number of possible GM models fitted to CMI data  | 34                               |
| 2.9<br>2.10  | Values of $R^2$ for a number of possible GM models fitted to England and Wales male data   | 35                               |
| 2.11   | Values of $R^2$ for a number of possible GM models fitted to England and Wales female data   | 36                               |
| 3.1<br>3.2   | Impact on the AIC and BIC of introducing gamma parameters  | 70                               |
| 0.2  | gamma parameters   | 71                               |
| 4.1<br>4.2<br>4.3<br>4.4<br>4.5<br>4.6<br>4.7<br>4.8 | Parameter estimates of the AR(1) processes fitted to the $\kappa^{(0)}$ series<br>Parameter estimates of the AR(1) processes fitted to the $\kappa^{(3)}$ differences<br>Parameter estimates of the AR(1) processes fitted to the $\kappa^{(4)}$ series<br>Parameter estimates of the AR(1) processes fitted to the $\kappa^{(5)}$ series<br>Parameter estimates of the AR(1) processes fitted to the logged $\gamma$ series<br>Sample correlation coefficients between the residuals of the time series<br>models fitted to the kappa parameter graphs for CMI data<br>Sample correlation coefficients between the residuals of the time series<br>models fitted to the kappa parameter graphs for England and Wales male<br>data<br>Sample correlation coefficients between the residuals of the time series | 84<br>87<br>88<br>89<br>90<br>90 |
|  | models fitted to the kappa parameter graphs for England and Wales female data  | 90                               |
| 5.1<br>5.2   | Key to Tables 5.2-5.14<br>Means, standard deviations and key percentiles of the empirical distributions<br>of deferred/immediate annuity values for CMI data under the low<br>improvement assumption and the log-linear extrapolation method – 500   | 128                              |
| 5.3  | scenarios<br>Means, standard deviations and key percentiles of the empirical distributions<br>of deferred/immediate annuity values for CMI data under the high<br>improvement assumption and the log-linear extrapolation method – 500<br>scenarios  | 129                              |
| 5.4  | Means, standard deviations and key percentiles of the empirical distributions<br>of deferred/immediate annuity values for England and Wales male data<br>under the low improvement assumption and the log-linear extrapolation<br>method 500 scenarios   | 129                              |
| 5.5  | Means, standard deviations and key percentiles of the empirical distributions<br>of deferred/immediate annuity values for England and Wales male data<br>under the high improvement assumption and the log-linear extrapolation  | 150                              |
|  | method – 500 scenarios   | 130                              |

| 5.6  | Means, standard deviations and key percentiles of the empirical distributions<br>of deferred/immediate annuity values for England and Wales female data<br>under the low improvement assumption and the log-linear extrapolation  |     |
|------|---|-----|
| 5.7  | method – 500 scenarios<br>Means, standard deviations and key percentiles of the empirical distributions<br>of deferred/immediate annuity values for England and Wales female data<br>under the high improvement assumption and the log-linear extrapolation                           | 131 |
| 5.8  | method – 500 scenarios<br>Means, standard deviations and key percentiles of the empirical distributions<br>of deferred/immediate annuity values for England and Wales male data<br>under the high improvement assumption and the log-linear extrapolation<br>method – 5 000 scenarios | 131 |
| 5.9  | Means, standard deviations and key percentiles of the empirical distributions<br>of deferred/immediate annuity values for CMI data under the low<br>improvement assumption and the LifeMetrics extrapolation method – 500<br>scenarios  | 133 |
| 5.10 | Means, standard deviations and key percentiles of the empirical distributions of deferred/immediate annuity values for CMI data under the high improvement assumption and the LifeMetrics extrapolation method – 500  | 100 |
| 5.11 | Means, standard deviations and key percentiles of the empirical distributions<br>of deferred/immediate annuity values for England and Wales male data<br>under the low improvement assumption and the LifeMetrics extrapolation   | 100 |
| 5.12 | Means, standard deviations and key percentiles of the empirical distributions<br>of deferred/immediate annuity values for England and Wales male data<br>under the high improvement assumption and the LifeMetrics extrapolation  | 104 |
| 5.13 | Means, standard deviations and key percentiles of the empirical distributions<br>of deferred/immediate annuity values for England and Wales female data<br>under the low improvement assumption and the LifeMetrics extrapolation   | 134 |
| 5.14 | method – 500 scenarios<br>Means, standard deviations and key percentiles of the empirical distributions<br>of deferred/immediate annuity values for England and Wales female data<br>under the high improvement assumption and the LifeMetrics extrapolation<br>method 500 scenarios  | 135 |
| 5.15 | Deferred/immediate annuity values based on previously published<br>projections applied to the 2005 CMI mortality curve fitted in this thesis, using<br>the log-linear extrapolation method  | 136 |
| 5.16 | Deferred/immediate annuity values based on previously published<br>projections applied to the 2005 England and Wales male mortality curve<br>fitted in this thesis, using the log-linear extrapolation method   | 136 |
| 5.17 | Deferred/immediate annuity values based on previously published<br>projections applied to the 2005 England and Wales female mortality curve<br>fitted in this thesis, using the log-linear extrapolation method   | 136 |
| 5.18 | ICA capital calculations for deferred/immediate annuities and comparisons<br>with the annuity value distributions obtained in Section 5.5, as shown in<br>Table 5.3 for CMI data, Table 5.5 for England and Wales male data and<br>Table 5.7 for England and Wales female data        | 130 |
| 5.19 | Coefficients of the $Z_{2006}^{(i)}$ , for <i>i</i> = 0,3,4,5, in the expressions for $C(x)$ for each   | -   |
| 5.20 | age <i>x</i> and each data set<br>Diversification benefits from a portfolio of deferred/immediate annuities with<br>equal numbers of annuitants aged 35, 45, 55, 65 and 75 in 2005, compared  | 137 |
| 5.21 | with each age separately<br>Correlation coefficients between $C(x)$ and $C(y)$ for different ages x and y, for  | 137 |
| 5.22 | CMI data<br>Correlation coefficients between $C(x)$ and $C(y)$ for different ages x and y, for  | 138 |
| 5.23 | England and Wales male data<br>Correlation coefficients between $C(x)$ and $C(y)$ for different ages x and y, for<br>England and Wales female data  | 138 |
|      | בווטומווע מווע אאמוכא וכווומוכ עמנמ   | 130 |

| 5.24 | Investigation of the accuracy of Approximation 1 for CMI data          | 139 |
|------|--|-----|
| 5.25 | Investigation of the accuracy of Approximation 1 for England and Wales |     |
|      | male data  | 141 |
| 5.26 | Investigation of the accuracy of Approximation 1 for England and Wales |     |
|      | female data  | 143 |
| 5.27 | Investigation of the accuracy of Approximation 2 for CMI data          | 145 |
| 5.28 | Investigation of the accuracy of Approximation 2 for England and Wales |     |
|      | male data  | 147 |
| 5.29 | Investigation of the accuracy of Approximation 2 for England and Wales |     |
|      | female data  | 149 |

### 1: Introduction

It is necessary to model future mortality in order to assess the level of reserves and capital required for a portfolio of immediate or deferred annuities held by a life insurance company or by a pension fund, or for a portfolio of assurance contracts held by a life insurance company. Such modelling is particularly important for life insurance companies with significant guaranteed annuity option liabilities, as improvements in longevity have the potential to bring about much larger percentage increases in the value of these options than in the value of the underlying annuities. It is also necessary to model future mortality to price annuity and life insurance contracts, and to value securities whose payoffs are contingent on future mortality.

Figure 1.2 shows historical plots of the force of mortality as a function of age and calendar year. The data sets these plots relate to will be described in the 'Data' section below. We can see in all cases that the bands of colour in the plots slope upwards, indicating a persistent improvement in longevity over the period of the data. This is a key feature of the data that needs to be captured in a mortality model, otherwise the reserves and prices calculated by the model for annuity contracts cannot be expected to be adequate and those for life insurance contracts may be excessive. In order to capture this feature, the model needs to consider mortality as a two-dimensional function of calendar year as well as of age.

The rate of improvement in mortality shown in the plots in Figure 1.2 is not uniform but varies significantly by age, by calendar year and between the three data sets. Given the high values shown in Figure 1.1 for the exposed to risk for each of the data sets, it seems unlikely that these variations result solely from random fluctuations. As a result of the wide range of factors that have been observed to affect mortality, including standards of health care and lifestyle factors such as diet and smoking, it is unlikely that we will be able to obtain a definitive explanation for all these variations. It is still less likely that we will be able to make a reliable deterministic forecast of how these underlying factors will evolve in the future and hence derive a reliable deterministic mortality projection. We shall therefore develop a stochastic mortality model, explicitly recognising the uncertainty of future mortality. This is particularly important in reserving and capital assessment applications, as an institution with liabilities dependent on future mortality needs to consider this uncertainty to ensure that it has sufficient capital to meet the liabilities with a high level of confidence. Furthermore, there may be a regulatory requirement on the institution to hold sufficient capital to meet its liabilities with a defined level of confidence, and we shall consider an example of this in Section 5.6.

A number of stochastic mortality models considering mortality as a two-dimensional function of both age and calendar year have been fitted in previously published papers and some of the main examples are discussed below.

### P-spline model

Currie et al. (2004) propose a non-parametric model, the P-spline model, of the form:

$$\log \mu_{xt} = \sum_{i} \sum_{j} \theta_{ij} B_i^a(x) B_j^y(t) ,$$

where  $\mu_{xt}$  is the force of mortality at age *x* in calendar year *t*, the  $\theta_{ij}$  are parameters to be estimated and the functions  $B_i^a$  and  $B_j^y$  are cubic basis splines, as defined in, for example, de Boor (2001). A penalty is imposed on parameter estimates  $\{\theta_{ij}\}$  which do not vary smoothly either with *i* and *j* (an *age-period penalty*) or with *i* and *j* – *i* (an *age-cohort penalty*). The level of this penalty determines the balance between smoothness and goodness of fit. The estimation of the  $\theta_{ij}$  is then by maximising the log-likelihood less the penalty.

This model is considered by the CMI in CMI (2005), and further in CMI (2006) where it is fitted to the CMI's male assured lives data set and to England and Wales population data for both males and females. Being non-parametric, it has the advantage of flexibility, in that it can fit any pattern of mortality as a function of age and of calendar year that is in evidence in the data. However, the model has been criticised for producing projections of future mortality that

are unduly sensitive to the last year in the data set. In addition, as the model does not impose any structure on the dependence of mortality on age, there can be no certainty that the projected curves of the logarithm of mortality as a function of age (hereafter simply referred to simply as *mortality curves*) in future years will be reasonable. Finally, the procedure for fitting the model only produces percentiles of the distribution of mortality rates at each age in each future year. For stochastic modelling, we are more likely to be interested in generating a number of sample paths containing projected mortality rates at all ages in all future years.

### Lee-Carter model

Lee and Carter (1992) propose the following model:

$$\log \mu_{xt} = \alpha_x + \beta_x \kappa_t,$$

where the  $\alpha_x$ ,  $\beta_x$  and  $\kappa_t$  are parameters to be estimated. This model is considered by the CMI in CMI (2005), and further in CMI (2007) where it is fitted to the CMI's male assured lives data set and to England and Wales population data for both males and females. A time series model is then fitted to the kappa parameters, and projected into the future to calculate future mortality rates and corresponding sample annuity values.

A key advantage of this model over the P-spline model is that, as future projections can be calculated simply by projecting the time series of kappa parameters, both sample paths and percentiles of the distribution of future mortality rates can readily be calculated. Another advantage of this model compared with the P-spline model is that the parameters have a clear interpretation - the alpha parameters represent the variation of mortality with age, the kappa parameters represent the improvement of mortality over time and the beta parameters provide for the possibility that mortality may improve more rapidly at some ages than at others. However, CMI (2007) concludes that this model does not fit United Kingdom data well because it cannot incorporate cohort effects. This refers to the fact that certain generations have consistently exhibited either particularly high or particularly low mortality improvements compared with the previous generation. For example, Willets et al. (2004) identify the generation centred on year of birth 1931 for England and Wales data for both males and females, and the generation centred on year of birth 1926 for CMI data, as exhibiting particularly high mortality improvements compared with the previous generation. Willets (2004) suggests that this phenomenon is mainly caused by a decline in smoking prevalence and by changes in diet in early life.

Lee and Carter (1992) estimated the parameters of their model by minimising the following sum of squares:

$$\sum_{x}\sum_{t}(\log \hat{\mu}_{xt} - \alpha_x - \beta_x \kappa_t)^2,$$

where  $\hat{\mu}_{xt} = \frac{D_{xt}}{E_{xt}}$  and  $D_{xt}$  and  $E_{xt}$  are, respectively, the number of deaths and the central

exposed to risk in the data at age x in calendar year t. They then adjusted the kappa parameters so as to match the actual and expected total numbers of deaths in each calendar year. However, Brouhns *et al.* (2002) criticise this procedure for placing undue weight on regions of the data set where there are few deaths and where the standard errors of the force of mortality are therefore large. Brouhns *et al.* instead model the number of deaths at age x in calendar year t as a Poisson random variable with parameter equal to  $E_{xt}$  multiplied by the Lee-Carter force of mortality, *i.e.* 

$$E_{xt} \exp(\alpha_x + \beta_x \kappa_t)$$

The alpha, beta and kappa parameters are then estimated by maximum likelihood.

Lee and Carter (1992), Brouhns *et al.* (2002) and CMI (2007) all treat the alpha, beta and kappa parameters as fixed quantities estimated from the data. The stochastic variation in future projections then comes entirely from the innovation terms in the time series process fitted to the kappa parameters. However, this does not allow for the fact that the parameter estimates are subject to uncertainty. It also does not allow for the fact that the new data which emerge over time will in practice lead us to update our estimates of the alpha and beta parameters, and of the parameters of the time series process governing the kappa parameters. To address these issues, Czado *et al.* (2005) develop a Bayesian approach. Together with an assumed prior distribution, the data are used to derive a joint posterior distribution which is updated as new data emerge and gives interval estimates of the authors use the technique of *Markov chain Monte Carlo* to construct a Markov chain whose stationary distribution is the required posterior distribution.

### Extensions to the Lee-Carter model incorporating cohort effects

Renshaw and Haberman (2006) propose an extension to the Lee-Carter model incorporating an explicit allowance for cohort effects:

$$\log \mu_{xt} = \alpha_x + \beta_x^{(1)} \kappa_t + \beta_x^{(2)} \gamma_{t-x}.$$

The authors fit this model to England and Wales population data, for both males and females. This addresses the criticism of the Lee-Carter model as not allowing for cohort effects. However, CMI (2007), which includes a limited investigation of this model, reports some problems with its convergence and robustness. The iterative procedure used by Renshaw and Haberman to estimate the parameters was found to arrive at different solutions according to the parameter values that were taken as the starting point for the iterations. For certain subsets of CMI male assured lives data and of England and Wales male data, the model completely failed to converge. One possible reason for these problems is that convergence is extremely slow as a result of the likelihood function being relatively flat in some directions, because it is relatively difficult to distinguish between a cohort effect and a combination of age and period effects, and much steeper in others. Thus the iterations may simply not have been allowed to run for long enough. However, Cairns et al. (2008) provide some evidence to suggest that the problems with the model are more fundamental than this, in that the likelihood function has multiple maxima. Cairns et al. successively added additional calendar years to the data and found that in some cases, adding an additional year's data caused the parameter estimates to jump to a solution with qualitatively different behaviour. Cairns et al. also identify issues with the plausibility of future projections produced by the model in that there is more uncertainty in the projections at ages around 65 than at the oldest ages in the data, which does not seem reasonable given that there is more data at ages around 65.

Cairns *et al.* (2008) consider simplifying the Renshaw-Haberman model by making the beta parameters independent of age, leading to the so-called *age-period-cohort model*, which the authors refer to as model M3 whereas the Renshaw-Haberman model is referred to as model M2. The authors find that making the beta parameters independent of age resolves the issues with the convergence and robustness of the Renshaw-Haberman model. However, under this simplified model, any period effect and any cohort effect must have the same percentage impact on the force of mortality at all ages, and this might be considered too limiting.

### Cairns-Blake-Dowd (CBD) model

Cairns et al. (2006) propose the following model:

$$q_{xt} = \frac{\exp(\kappa_t^{(0)} + \kappa_t^{(1)}x)}{1 + \exp(\kappa_t^{(0)} + \kappa_t^{(1)}x)}$$

Note that this model is fitted to initial mortality rates  $q_{xt}$  rather than to forces of mortality  $\mu_{xt}$ . The authors fit it to England and Wales male data for ages 60 and above. This model is more flexible than the Lee-Carter model in terms of the ways in which mortality can evolve over time, with trends in the  $\kappa^{(0)}$  parameters having proportionately more impact on mortality at the younger ages in the data set and trends in the  $\kappa^{(1)}$  parameters having more impact at the older ages. The model is considerably less flexible than the Lee-Carter model in terms of the ways in which mortality can vary with age. The authors find no evidence that the model does not allow a sufficiently wide range of shapes of the mortality curve as a function of age for the data set they consider, but this would not necessarily be the case for other data sets, particularly those extending to ages below 60. The authors find that the shapes of the  $\kappa^{(0)}$  and  $\kappa^{(1)}$  parameter graphs they obtain implicitly reflect cohort effects, and this motivates the development of extensions of the CBD model that incorporate an explicit allowance for cohort effects.

#### Extensions to the CBD model incorporating cohort effects

Cairns *et al.* (2007) propose three different extensions to the CBD model incorporating cohort effects explicitly, referring to the CBD model as model M5 and to the extensions as models M6, M7 and M8. The extensions are as follows:

$$M6: q_{xt} = \frac{\exp(\kappa_{t}^{(0)} + \kappa_{t}^{(1)}x + \gamma_{t-x})}{1 + \exp(\kappa_{t}^{(0)} + \kappa_{t}^{(1)}x + \gamma_{t-x})},$$

$$M7: q_{xt} = \frac{\exp(\kappa_{t}^{(0)} + \kappa_{t}^{(1)}x + \kappa_{t}^{(2)}x^{2} + \gamma_{t-x})}{1 + \exp(\kappa_{t}^{(0)} + \kappa_{t}^{(1)}x + \kappa_{t}^{(2)}x^{2} + \gamma_{t-x})},$$

$$M8: q_{xt} = \frac{\exp(\kappa_{t}^{(0)} + \kappa_{t}^{(1)}x + \gamma_{t-x}(x_{c} - x))}{1 + \exp(\kappa_{t}^{(0)} + \kappa_{t}^{(1)}x + \gamma_{t-x}(x_{c} - x))},$$

where  $x_c$  is a parameter to be determined. The authors find that all three of these models achieve a very significant improvement over M5 in goodness of fit to both England and Wales and United States male population data, with M7 giving a sufficient improvement over M6 to justify the introduction of the  $\kappa^{(2)}$  parameters. The criterion used to measure goodness of fit here is the Bayes Information Criterion (BIC), which is discussed further in Section 2.5 of this paper. The authors find that models M7 and M6 (which is a special case of M7) are adequately robust but raise concerns about the robustness of model M8 in the context of US data. This is developed further in Cairns *et al.* (2008) where it is found that model M8 fails to produce plausible projections of future mortality at ages over 65 for US data.

### Gompertz-Makeham (GM) models

The graduated one-dimensional tables of assured life, annuitant and pensioner mortality produced by the CMI extend to ages below 60, with the annuitant and pensioner tables including early as well as normal retirements. In most cases, it is found that a model similar to the CBD model cannot adequately capture the age structure of the data. As a result, for a number of years, the CMI have based most of their graduations on *Gompertz-Makeham* models of order (*r*,*s*), or *GM*(*r*,*s*) models, for various non-negative integers *r* and positive integers *s*. This family of models was proposed by Forfar *et al.* (1988). The GM(*r*,*s*) model is defined by:

$$\mu_x = \sum_{i=0}^{r-1} \kappa^{(i)} x^i + \exp\left(\sum_{j=0}^{s-1} \kappa^{(r+j)} x^j\right),$$

where  $\mu_x$  is the force of mortality at age *x* and  $\kappa^{(0)}, \dots, \kappa^{(r+s-1)}$  are parameters to be estimated. In the case where r = 0, we define the empty sum  $\sum_{i=0}^{r-1} \kappa^{(i)} x^i$  to have the value 0. Typically

mortality at younger ages is driven primarily by the parameters  $\kappa^{(0)}, ..., \kappa^{(r-1)}$  and mortality at older ages is driven primarily by  $\kappa^{(r)}, ..., \kappa^{(r+s-1)}$ .

Figure 1.3 shows fitted mortality curves under several different GM models. The data set considered is the CMI data set, which will be described in the 'Data' section below, and the calendar year considered is 2000. The changes for the GM(1,3) model compared with the GM(1,2) model, and for the GM(2,3) model compared with the GM(1,3) model, have been exaggerated to emphasise the features that have changed. We see from Figure 1.3(i) that under the GM(0,2) model, the logarithm of the force of mortality is a linear function of age. It is found that this understates mortality at the youngest ages of the data set, and Figure 1.3(ii) shows that the main effect of introducing  $\kappa^{(0)}$  parameters is to rectify this by making mortality increase more slowly than exponentially with age at the youngest ages of the data set. Similarly, Figure 1.3(ii) shows that under the GM(2,3) model, the graph of the logarithm of mortality is broadly linear below age 45 (with a lower slope than at higher ages), rather than for the slope of the graph to continue to decrease as the age approaches the youngest age of the data set (30).

In this paper, we shall extend the GM family of models to the two-dimensional case by making the kappa parameters functions of time. Thus we shall consider models of the form:

$$\mu_{xt} = \sum_{i=0}^{r-1} \kappa_t^{(i)} x^i + \exp\left(\sum_{j=0}^{s-1} \kappa_t^{(r+j)} x^j\right),$$

where  $\mu_{xt}$  is the force of mortality at age *x* in calendar year *t*. We shall fit models of this form to the three data sets described below.

#### <u>Data</u>

The majority of the papers referred to above consider only population data. We, however, will also consider CMI male data, which can be expected to be more relevant to a typical life insurance company's liabilities. The population data we shall consider will be England and Wales data for both males and females.

The CMI male data we shall use represent the mortality experience of male assured lives holding endowment or whole life assurance policies with UK life insurance companies that contributed to the CMI's investigation over the period concerned. The data consist of the number of deaths and the central exposed to risk at each of the ages 30-90 nearest birthday in each of the calendar years 1947-2005, a total of 3,599 data cells. We shall not consider CMI data for females, or for pensioner or annuitant data sets, here as these data sets have either insufficient volume or an insufficiently long history to draw reliable conclusions.

The England and Wales data, for both males and females, were taken from the Human Mortality Database (www.mortality.org) maintained by the University of California, Berkeley (USA) and by the Max Planck Institute for Demographic Research (Germany). The data were originally provided by the Office for National Statistics (ONS). The data for each gender consist of the number of deaths and the central exposed to risk at each of the ages 30-89 last birthday in each of the calendar years 1962-2005, a total of 2,640 data cells. The data were downloaded on 20 March 2008.

In what follows, we shall denote the number of deaths in the data at age x in calendar year t by  $D_{xt}$ , where x is defined as the age nearest birthday in the case of CMI data and as the age last birthday in the case of England and Wales data. We shall denote the central exposed to risk at age x in calendar year t by  $E_{xt}$ .

Figure 1.1 shows the logarithm of the exposed to risk at each age in each calendar year for each data set. Regions of these graphs where the exposed to risk is high are coloured red

and those where the exposed to risk is low are coloured blue. We can make the following observations from Figure 1.1:

- Noting the different scales of the three graphs, the CMI data is a smaller data set than either of the England and Wales data sets. This is because it relates only to assured lives rather than to the general population.
- The exposed to risk in all three graphs decreases towards the top of the age range of the data. For England and Wales data, this can be explained by relatively few lives surviving to these high ages. For CMI data, another factor is that relatively few retired lives would be expected to hold life insurance policies, and this results in the significant decrease in exposed to risk starting at a rather younger age in CMI data than in the England and Wales data sets.
- In all three graphs, relatively low values of the exposed to risk can be seen at years of birth in the second half of the 1910s. This is likely to be a result of low birth rates during the First World War. For the male data sets, the fact that a substantial proportion of these lives were killed in the Second World War is also likely to be a significant factor.
- The England and Wales data sets show relatively high values of the exposed to risk for lives born in the years immediately following the Second World War and in the 1960s. This can be identified with high birth rates during those periods.
  - The CMI data set shows a significant decrease in the exposed to risk in the last few years of the data at ages below a typical retirement age. The decrease in exposed to risk begins rather earlier at the younger ages, where the policies tend to be relatively new. This can be explained by the tendency in recent years for individuals to take out repayment mortgages, under which a life insurance company provides a term assurance policy, rather than endowment mortgages, under which the insurer provides an endowment. The latter policies are included in the data set but the former are not. The decline in popularity of endowments can largely be explained by the ending of tax relief on premiums under this business for policies taken out from 1984.

Figure 1.2 shows the logarithm of the crude force of mortality,  $\log\left(\frac{D_{xt}}{E_{rt}}\right)$ , at each age in

each calendar year for each data set, with a red colour representing high mortality and a blue colour representing low mortality.

It is clear from all the graphs in Figure 1.2 that there is an increasing trend of mortality with age. In addition, as mentioned in the opening remarks, the bands of colour in the graphs slope from bottom left to top right, indicating a trend for mortality to improve over time. There is evidence that the rate of this improvement has been highest for CMI data and lowest for England and Wales female data.

Figure 1.4 shows graphs of the logarithm of the crude force of mortality for CMI data as a function of age *x* for *t* = 1960, 1975, 1990 and 2005 respectively. Figure 1.5 shows the same information for England and Wales male data and Figure 1.6 shows the same information for England and Wales female data, for *t* = 1970, 1980, 1990 and 2000 respectively. 95% confidence limits for the force of mortality are also shown, *i.e.* the lower limit is the 2.5<sup>th</sup> percentile and the upper limit is the 97.5<sup>th</sup> percentile. These confidence limits assume that the number of deaths at each age in each calendar year has a Poisson distribution. The graph for *t* = 2005 for CMI data (Figure 1.4(iv)) uses an age range of 35-90 instead of 30-90 because of the paucity of data at the youngest ages.

The following observations can be made from Figures 1.4-1.6:

- As for Figure 1.2, all the figures clearly demonstrate an increasing trend of mortality with age. Some evidence can be seen of the increasing trend being smaller in percentage terms at younger ages, *i.e.* the curves are less steep at younger ages.
- As in Figure 1.2, some tendency can be seen for mortality at each age to decrease over time.
- Figure 1.4 exhibits greater volatility and wider confidence intervals than Figures 1.5 and 1.6. This is because the CMI data is a smaller data set than the England and Wales data sets.
- In England and Wales data, for a given age and calendar year, the force of mortality for females is lower than that for males.
- In both CMI data and England and Wales data, the confidence intervals widen towards the bottom of the age range. This is because there are relatively few deaths at these ages.
- In CMI data, some widening of the confidence intervals is visible towards the top of the age range. Although mortality is high at these ages, the exposed to risk is much lower than around the middle of the age range.

### Structure of the thesis

Chapter 2 fits GM(*r*,*s*) models, for various values of *r* and *s*, to the three data sets, and arrives at a conclusion as to the most appropriate values of *r* and *s* on which to base future mortality projections. Chapter 3 considers a simple method of adjusting for cohort effects, introducing a further time series of parameters indexed by year of birth rather than by calendar year. Chapter 4 fits univariate time series models to the parameter estimates calculated in Chapters 2 and 3. Chapter 5 presents stochastic projections of the future values of the parameters and resulting sample immediate and deferred annuity functions, calculated at a deterministic interest rate. An application to risk-based capital calculations, under the Individual Capital Assessment (ICA) regime of the Financial Services Authority (FSA), is given. This chapter contains more applications and gives more discussion of the observations than can be found in most previously published papers on stochastic mortality models. Chapter 6 gives our conclusions.



Year

Figure 1.1 – Logarithms of the exposed to risk for the three data sets – (i) CMI data, (ii) England and Wales male data, (iii) England and Wales female data







**Figure 1.3 –** Fitted mortality curves under various GM models, with exaggeration to highlight certain features – (i) GM(0,2), (ii) GM(1,2), (iii) GM(1,3), (iv) GM(2,3)



**Figure 1.4** – CMI crude force of mortality as a function of age for various calendar years – solid curve = central estimate, dashed curves = 95% confidence limits (2.5<sup>th</sup> and 97.5<sup>th</sup> percentiles) – (i) 1960, (ii) 1975, (iii) 1990, (iv) 2005







**Figure 1.6** – England and Wales female crude force of mortality as a function of age for various calendar years – solid curve = central estimate, dashed curves = 95% confidence limits (2.5<sup>th</sup> and 97.5<sup>th</sup> percentiles) – (i) 1970, (ii) 1980, (iii) 1990, (iv) 2000

### 2: Fitting a Gompertz-Makeham model for each calendar year to CMI male assured lives data and to England and Wales population data

### 2.1 Model

As stated in Chapter 1, the GM(*r*,*s*) model is defined by:

$$\mu_{xt} = \sum_{i=0}^{r-1} \kappa_t^{(i)} x^i + \exp\left(\sum_{j=0}^{s-1} \kappa_t^{(r+j)} x^j\right),$$

where  $\mu_{xt}$  is the force of mortality at age *x* in calendar year *t* and  $\kappa_t^{(0)}, \ldots, \kappa_t^{(r+s-1)}$  are parameters to be estimated for each *t*. *x* and *t* are considered to be discrete variables, *i.e.* we assume that mortality rates only change when lives attain a new age label or when a new calendar year begins.

We follow Brouhns *et al.* (2002) in assuming that the number of deaths at each age *x* in each calendar year *t* has a Poisson distribution with parameter  $E_{xt}$  multiplied by this  $\mu_{xt}$ . The range of ages we consider is denoted  $x_1, \ldots, x_N$ . The largest values of *r* and *s* we shall consider are both 4. We also reparameterise the models so that as *x* varies for fixed *t*, the mean of the

quantities to which each of the kappa parameters (other than  $\kappa_t^{(0)}$  and  $\kappa_t^{(r)}$ ) is applied is zero. This is done to avoid excessively large numbers appearing in the estimation process for the kappa parameters. Under the GM(3,4) model, for example,  $D_{xt}$  then has a Poisson distribution with parameter  $\mu_{xt}E_{xt}$ , where:

$$\begin{split} \mu_{xt} &= \kappa_t^{(0)} + \kappa_t^{(1)} (x - \overline{x}) + \kappa_t^{(2)} ((x - \overline{x})^2 - \hat{\sigma}_x^2) \\ &+ \exp[\kappa_t^{(3)} + \kappa_t^{(4)} (x - \overline{x}) + \kappa_t^{(5)} ((x - \overline{x})^2 - \hat{\sigma}_x^2) + \kappa_t^{(6)} (x - \overline{x})^3], \\ &\quad \overline{x} = \frac{1}{N} \sum_{i=1}^N x_i, \\ &\quad \hat{\sigma}_x^2 = \frac{1}{N} \sum_{i=1}^N (x_i - \overline{x})^2. \end{split}$$

In what follows, we shall always label the kappa parameters as in the above formula, even when we are considering values of *r* less than 3 and/or values of *s* less than 4. In the case where r = 4, we can no longer label the kappa parameters in this way, but we shall have no need to refer explicitly to the kappa parameters in any model with r = 4.

For comparison, the CBD model referred to in Chapter 1 is equivalent to a GM(0,2) model except that, instead of a linear function of age for each calendar year being fitted to log  $\mu_{xt}$ , it is fitted to:

$$\log\left(\frac{q_{xt}}{1-q_{xt}}\right),$$

where  $q_{xt}$  is the initial mortality rate at age x in calendar year t.

### 2.2 Parameter estimation methodology

The Poisson assumption implies that the log-likelihood function of the full GM(3,4) model is:

$$\ell = c - \sum_{x} \sum_{t} E_{xt} \mu_{xt} (\kappa_{t}^{(0)}, \kappa_{t}^{(1)}, \kappa_{t}^{(2)}, \kappa_{t}^{(3)}, \kappa_{t}^{(4)}, \kappa_{t}^{(5)}, \kappa_{t}^{(6)}) + \sum_{x} \sum_{t} D_{xt} \log \mu_{xt} (\kappa_{t}^{(0)}, \kappa_{t}^{(1)}, \kappa_{t}^{(2)}, \kappa_{t}^{(3)}, \kappa_{t}^{(4)}, \kappa_{t}^{(5)}, \kappa_{t}^{(6)})$$

where:

$$\mu_{xt}(\kappa_t^{(0)}, \kappa_t^{(1)}, \kappa_t^{(2)}, \kappa_t^{(3)}, \kappa_t^{(4)}, \kappa_t^{(5)}, \kappa_t^{(6)}) = \kappa_t^{(0)} + \kappa_t^{(1)}(x - \bar{x}) + \kappa_t^{(2)}((x - \bar{x})^2 - \hat{\sigma}_x^2) + \exp[\kappa_t^{(3)} + \kappa_t^{(4)}(x - \bar{x}) + \kappa_t^{(5)}((x - \bar{x})^2 - \hat{\sigma}_x^2) + \kappa_t^{(6)}(x - \bar{x})^3]$$

and c is a constant.

The estimation of the kappa parameters was by maximum likelihood. An iterative scheme was used, where each step consisted of updating the values of one of the kappa parameters for all calendar years *t* using the Newton-Raphson method, leaving the other kappa parameters constant. Brouhns *et al.* (2002) use a similar scheme to fit the model they consider to Belgian population mortality data. Further details of the iterative scheme are given in Appendix A.

Table 2.1 shows the parameter values that were taken as the starting point for the estimation of each of the GM models fitted. The iterations were stopped when a complete loop of the iterations, from one step of updating the  $\kappa^{(0)}$  parameters to another, changed none of the parameter estimates by more than 10<sup>-6</sup>. In other words, the iterations were stopped when, in the notation of Appendix A:

$$\left|\boldsymbol{\varepsilon}_{t}^{(i)}-\boldsymbol{\alpha}_{t}^{(i)}\right|<10^{-6}$$

for all *t* and all *i*. An exception was that if this condition was satisfied at the very first loop of the iterations, then they were not stopped.

| Model   | Starting point for iterations   |
|---|---|
| GM(0,2)   | Mortality rate independent of age for   |
|   | each calendar year –  |
|   | $\kappa_t^{(3)} = \log\left(\frac{\sum_{x} D_{xt}}{\sum_{x} E_{xt}}\right), \ \kappa_t^{(4)} = 0$ |
| GM(1,2)   | Result of GM(0,2) model   |
| GM(1,3)   | Result of GM(1,2) model   |
| GM(2,3)   | Result of GM(1,3) model   |
| GM(3,3) (fitted to CMI data only)               | Result of GM(2,3) model   |
| GM(2,4)   | Result of GM(2,3) model   |
| GM(3,4) (fitted to England and Wales data only) | Result of GM(2,4) model   |
| GM(4,4) (fitted to England and Wales data only) | Result of GM(3,4) model   |

Table 2.1 – Starting points for the iterations to estimate the parameters of the GM models

In all the models fitted, the  $\kappa^{(3)}$  parameters were the ones of highest magnitude, so in practice, the iterations were stopped when the greatest absolute movement of any  $\kappa^{(3)}$  parameter over a loop was just less than 10<sup>-6</sup>.

It should be noted that there is no interaction between calendar years in this fitting procedure, other than in the criterion for stopping the iterations. We are therefore estimating the kappa parameters for each calendar year independently. We should like to be able to explain the kappa parameter graphs we obtain by reference to period effects alone. However, the next section shows that period effects alone do not always provide adequate explanations.

### 2.3 Parameter estimation results

### 2.3.1 CMI data

Graphs of the parameter estimates for each of the models fitted to CMI data are shown in Figures 2.1-2.6. These figures relate to the GM(0,2), GM(1,2), GM(1,3), GM(2,3), GM(3,3) and GM(2,4) models respectively.

The downward slope of the  $\kappa^{(3)}$  parameters in Figure 2.1(i) indicates an overall improvement of mortality rates over the period 1947-2005, as noted in the 'Data' section of Chapter 1. The upward slope of the  $\kappa^{(4)}$  parameters in Figure 2.1(ii) indicates that, in percentage terms, this improvement has been more rapid at younger ages than at older ages.

In the GM(1,2) model compared with the GM(0,2) model, a quantity,  $\kappa_t^{(0)}$ , independent of age *x* for each calendar year *t* has been added to the formula for  $\mu_{xt}$ . Intuitively, one might expect these quantities to be positive, representing mortality from unnatural causes such as

accidents and violence. However, Figure 2.2(i) shows that many of the fitted values of  $\kappa_t^{(0)}$ 

are in fact negative. In fact, the years *t* for which this happens are from 1948 to 1988 inclusive, excluding 1985. This potentially gives rise to an issue if the model is extrapolated to ages under 30 (the youngest age in the data set), in that the extrapolated forces of mortality might be negative. This possibility was investigated and it was found that the extrapolated forces of mortality at small positive ages were indeed negative in the years 1950-84 inclusive and 1987. In certain years, the negative forces of mortality extended up to and including age 21. This indicates that, for the CMI data set, the GM(1,2) model should not be extrapolated to ages under 30 without adjustment.

Figures 2.2(ii) and (iii), compared with Figures 2.1(i) and (ii), respectively show that the overall trend of the  $\kappa^{(3)}$  and  $\kappa^{(4)}$  parameters with time is similar under the GM(1,2) model to that under the GM(0,2) model. Figure 2.2(ii) is almost identical to Figure 2.1(i). The differences between Figures 2.2(iii) and 2.1(ii) are slightly greater but the upward slope of the  $\kappa^{(4)}$  parameters remains the key feature.

We see from Figure 2.3(i) that in the GM(1,3) model, unlike the GM(1,2) model, all the fitted  $\kappa^{(0)}$  parameters are positive. This is consistent with the intuitive interpretation of the  $\kappa^{(0)}$  parameters as representing mortality from unnatural causes. It means that, for this data set, the GM(1,3) model is more likely than the GM(1,2) model to be suitable for extrapolation to ages under 30, as all the extrapolated forces of mortality will be positive.

Figures 2.3(ii) and (iii) show, respectively, the fitted  $\kappa^{(3)}$  and  $\kappa^{(4)}$  parameters under the GM(1,3) model. As in the case of the GM(1,2) model compared with the GM(0,2) model, the overall trend has not changed in the GM(1,3) model compared with the GM(1,2) model for either the  $\kappa^{(3)}$  or  $\kappa^{(4)}$  parameters. Figure 2.3(ii) is almost identical to Figure 2.2(ii). The differences between Figures 2.3(iii) and 2.2(iii) are slightly greater but the upward slope of the  $\kappa^{(4)}$  parameters remains the key feature.

The negative fitted  $\kappa^{(5)}$  parameters under the GM(1,3) model shown in Figure 2.3(iv) indicate underlying mortality that increases more slowly than exponentially with age at the oldest ages.

According to the two statistical tests which we shall describe in Section 2.5, the GM(2,3) model achieves a statistically significant improvement in fit over the GM(1,3) model. However, inspection of the parameter graphs for the GM(2,3) model in Figure 2.4 reveals certain features which seem more likely to constitute evidence of overfitting rather than reflecting genuine features of the underlying mortality rates. Of particular note are the strong positive correlation between the fitted  $\kappa^{(0)}$  and  $\kappa^{(1)}$  parameters and the strong positive correlation between the grant the negatives of the  $\kappa^{(4)}$  parameters and the strong positive correlation between the grant transformed range of values taken by each of the parameter series individually. The volatility of the  $\kappa^{(3)}$  parameters in particular is also much increased compared with the GM(1,3) model. If such features are included in future projections, then the results are likely to be inappropriate. Note that the correlations might be eliminated by considering

linear combinations of the different kappa series that are uncorrelated, but this would give rise to the issue that there is no unique method of constructing such linear combinations.

Figures 2.4(i) and (ii), showing negative  $\kappa^{(0)}$  and  $\kappa^{(1)}$  parameters respectively, suggest that the GM(2,3) model, like the GM(1,2) model, has the potential to give negative forces of mortality when extrapolated to ages under 30. This was investigated and was found to be an issue for the calendar years 1998 and 2000-05 inclusive. The year for which this issue is most significant is 2004, with the extrapolated forces of mortality being negative up to and including age 25. This indicates that for this data set, the GM(2,3) model, like the GM(1,2) model, is not suitable for extrapolation to ages under 30 without adjustment.

Figure 2.5 shows that the fitted parameter values of the GM(3,3) model, as for the GM(2,3) model, have certain features which suggest that there has been overfitting. There are strong positive correlations between the  $\kappa^{(0)}$ ,  $\kappa^{(1)}$  and  $\kappa^{(2)}$  parameters and between the  $\kappa^{(3)}$  parameters, the negatives of the  $\kappa^{(4)}$  parameters and the  $\kappa^{(5)}$  parameters, and the ranges of values taken by the parameters are even wider than under the GM(2,3) model. The shape of the graph of  $\kappa^{(3)}$  parameters is difficult to interpret as one would expect a steady improvement in mortality with time, and hence a steady fall in the  $\kappa^{(3)}$  parameters.

Figure 2.6, showing graphs of the parameter estimates for the GM(2,4) model, has parameter values for 1955 and 1956 that are very different from those for adjacent years. It is not apparent what feature of the data is causing these differences, but in general, when a model is overfitted, issues can arise in that small changes in the input data can lead to significant changes in the parameter estimates. Figure 2.6 also shows strong positive correlation between the  $\kappa^{(3)}$  parameters, the negatives of the  $\kappa^{(4)}$  parameters, the  $\kappa^{(5)}$  parameters and the negatives of the  $\kappa^{(6)}$  parameters, again providing evidence of overfitting.

### 2.3.2 England and Wales male data

Figures 2.7, 2.8, 2.9 and 2.10 respectively show the parameter estimates for the GM(0,2) model, the GM(1,2) model, the GM(1,3) model and the GM(2,3) model fitted to England and Wales male data. For models more complex than the GM(2,3) model, the corresponding graphs have not been shown because overfitting is already apparent in the graphs for the GM(2,3) model.

As for Figure 2.1, Figure 2.7 shows that there is an overall trend of improving mortality with time and that the improvements have tended to be faster in percentage terms at younger ages than at older ages. Comparing Figure 2.7 with Figure 2.1, there is evidence that the rate of improvement has been lower for England and Wales male data than for CMI data, although the difference in scale on the time axis are should be borne in mind. As the CMI data is weighted towards the higher socio-economic groups, this is consistent with the observation that mortality has improved more quickly among higher than among lower socio-economic groups. See Willets *et al.* (2004). The values of the  $\kappa^{(4)}$  parameters are lower for the England and Wales male data than for the CMI data, indicating that these mortality differentials between socio-economic groups narrow with increasing age.

In Figure 2.8, we again have a downward-sloping graph of  $\kappa^{(3)}$  parameters and an upwardsloping graph of  $\kappa^{(4)}$  parameters. It is not clear how the graph of  $\kappa^{(0)}$  parameters should be interpreted – it is likely that it reflects effects that are more appropriately incorporated in the model by introducing  $\kappa^{(5)}$  parameters. As in Figure 2.2, the negative values of the earlier  $\kappa^{(0)}$ parameters will lead to negative forces of mortality if the model is extrapolated below age 30, but in more recent years the  $\kappa^{(0)}$  parameters have been positive.

Figure 2.9 shows that, as for CMI data, introducing  $\kappa^{(5)}$  parameters to the model for England and Wales male data makes all the  $\kappa^{(0)}$  parameters positive. However, the most striking features of Figure 2.9 are that the  $\kappa^{(0)}$  and  $\kappa^{(4)}$  parameters rise sharply from 1980 to the early 1990s and then fall again, the  $\kappa^{(3)}$  parameters fall more steeply than usual in the 1980s and early 1990s and the  $\kappa^{(5)}$  parameters rise sharply from the mid-1990s onwards. Similarly to the graphs of the fitted parameter values in Cairns *et al.* (2006), the shapes of the graphs in Figure 2.9 can be explained by the cohort effect described in Chapter 1. Lives in England and Wales born between 1925 and 1945, and particularly around 1931, have experienced significantly lower mortality than the preceding generation, to an extent not explained by the overall improvement of mortality with time. In the 1980s and early 1990s, these lives were mostly in their 50s, and so a significant improvement in their mortality but not in the mortality of older generations might be expected to lead to a sharper than usual fall in the general level of the mortality curve, as measured by the  $\kappa^{(3)}$  parameters, and to an increase in the slope of the mortality curve, as measured by the  $\kappa^{(4)}$  parameters. As the  $\kappa^{(4)}$  parameters have increased, the  $\kappa^{(0)}$  parameters will then tend to increase to maintain broadly the same level of mortality at the youngest ages. This is indeed what we observe. Between the early 1990s and 2005, these lives were mostly in their 60s and early 70s, and so we might expect a decrease in the slope of the mortality at the oldest ages relative to mortality in the 60s and early 70s of age, as measured by the  $\kappa^{(5)}$  parameters. Again we do in fact observe this. We also observe the increases in the  $\kappa^{(0)}$  parameters that occurred in the 1980s reversing out between the early 1990s and 2005.

As for Figure 2.4, Figure 2.10 suggests that the introduction of  $\kappa^{(1)}$  parameters constitutes overfitting. In the GM(2,3) model fitted to England and Wales male data, there is strong positive correlation between the  $\kappa^{(0)}$  parameters, the  $\kappa^{(1)}$  parameters, the negatives of the  $\kappa^{(3)}$  parameters, the  $\kappa^{(4)}$  parameters and the negatives of the  $\kappa^{(5)}$  parameters, and the range of variation of the  $\kappa^{(4)}$  parameters in particular has increased significantly compared with the GM(1,3) model. As for the GM(2,3) model fitted to CMI data, the GM(2,3) model fitted to England and Wales male data was also extrapolated below age 30. Some issues with the extrapolated forces of mortality being negative were found, although fewer than for CMI data.

### 2.3.3 England and Wales female data

Figures 2.11, 2.12, 2.13 and 2.14 respectively show the parameter estimates for the GM(0,2) model, the GM(1,2) model, the GM(1,3) model and the GM(2,3) model fitted to England and Wales female data. For models more complex than the GM(2,3) model, the corresponding graphs have again not been shown.

Figure 2.11(i), showing the  $\kappa^{(3)}$  parameters under the GM(0,2) model for female data, is slightly less steep than Figure 2.7(i), for male data, indicating that female mortality has improved slightly less rapidly than male mortality over the period 1962-2005, although the majority of the mortality differential between the genders that was present in 1962 remains in 2005. The higher values of the  $\kappa^{(4)}$  parameters for the GM(0,2) model fitted to England and Wales female data, as shown in Figure 2.11(ii), than fitted to England and Wales male data, as shown in Figure 2.7(ii), indicate that the mortality differential between the genders narrows with increasing age, although this effect is not significant enough to close the gap between the genders completely at the oldest ages in the data.

The key differences between Figure 2.12 and Figure 2.8 are that, between 1962 and the late 1970s, the  $\kappa^{(0)}$  parameters for females decreased whereas those for males increased, the  $\kappa^{(3)}$  parameters for females decreased rather more slowly than for males, and the  $\kappa^{(4)}$  parameters for females decreased whereas those for males increased. Similar (indeed, stronger) features can also be seen in Figure 2.13 and will be discussed below.

The discussion of the impact of cohort effects on Figure 2.9 is also relevant to Figure 2.13. However, in the case of Figure 2.13, the shapes of the graphs between 1962 and the late 1970s also require explanation. It seems likely that the graphs reflect a further cohort effect, but in the direction of low rather than high mortality improvements for a particular generation. Table 1b and Figure 1b of Willets (2004) indicate that females born around 1915 have experienced particularly low rates of mortality improvement over the previous generation. This is likely to be mainly a result of an increase in smoking prevalence – Figure 5 of the same paper shows that lifetime cigarette consumption for females as a function of year of birth was increasing most rapidly around 1915.

Two further features of Figure 2.13 which did not occur in the corresponding figures for either of the male data sets should be noted. Firstly, most of the fitted  $\kappa^{(0)}$  parameters are negative and this may lead to negative forces of mortality if the GM(1,3) model fitted to England and

Wales female data is extrapolated to ages below 30. This extrapolation was performed and it was found that in 26 of the 44 years from 1962 to 2005, the extrapolated force of mortality at age 0 was negative. In certain years the negative extrapolated forces of mortality extended up to and including age 21. This shows that the GM(1,3) model fitted to this data set should not be extrapolated below age 30 without adjustment.

The second feature of Figure 2.13 that should be noted is that, unlike either of the male data sets, the fitted  $\kappa^{(5)}$  parameters are all positive, indicating mortality that increases more rapidly than exponentially with age at the oldest ages. This increase is offset to some extent by the lower values of the fitted  $\kappa^{(4)}$  parameters than for the male data sets.

On the basis of Figure 2.14, the GM(2,3) model appears to be overfitted to England and Wales female data. There is strong positive correlation between the  $\kappa^{(0)}$  parameters, the  $\kappa^{(1)}$  parameters, the negatives of the  $\kappa^{(3)}$  parameters, the  $\kappa^{(4)}$  parameters and the negatives of the  $\kappa^{(5)}$  parameters, and the ranges over which the parameter values vary (except for the  $\kappa^{(0)}$  parameters) have increased substantially compared with the GM(1,3) model. This is the same conclusion that we reached when fitting the GM(2,3) model to the male data sets.

### 2.4 Results for the force of mortality

Under the GM(0,2) model, the graph of the logarithm of the fitted force of mortality against age for a given calendar year is a straight line. Figures 2.15-2.17 show the differences between the fitted forces of mortality under the more complex models and these straight lines for a typical calendar year, 2000.

The graphs labelled (ii) in Figures 2.15-2.17 show that, for all three data sets, by far the largest impact of moving from the GM(0,2) model to the GM(1,2) model is to increase the fitted force of mortality at younger ages. This is consistent with our observation in the discussion of Figures 1.4-1.6 that the percentage rate of increase of mortality with age slows down at the younger ages in the data. The graphs labelled (ii) also show that moving from GM(0,2) to GM(1,2) decreases fitted mortality over the approximate age range 50-75 and increases it at ages over approximately 75. It is not clear at this stage whether these changes improve the fit or whether they arise because of the restricted range of shapes of the mortality curve available under the GM(1,2) model.

The key changes in the graphs labelled (iii) in Figures 2.15-2.17, for the GM(1,3) model, compared with the graphs labelled (ii) are towards the top of the age range of the data, which is not surprising given that the  $\kappa^{(5)}$  parameter has more impact on fitted mortality towards the top of the age range than at other ages. For the male data sets, the negative  $\kappa^{(5)}$  parameter tends to reduce mortality at the oldest ages, but for England and Wales female data, the positive  $\kappa^{(5)}$  parameter tends to increase mortality at the oldest ages. The graphs give us no reason to doubt that the decrease in fitted mortality over the age range 50-75 under the GM(1,2) model compared with the GM(0,2) model did genuinely improve the fit, but they do suggest that for both the male data sets, the increase at ages over 75 was caused mainly by the restricted range of shapes under the GM(1,2) model. We shall consider the goodness of fit of the models at different ages further in Section 2.8.

The graphs in Figures 2.15-2.17 for models more complex than the GM(1,3) model show no significant qualitative changes in shape compared with the graphs for the GM(1,3) model, with the exception of the very youngest ages in England and Wales female data where there is a decrease in mortality compared with the GM(1,3) model. Thus Figures 2.15-2.17 do not provide any strong reason to support the use of a more complex model than GM(1,3).

### 2.5 Akaike Information Criterion (AIC) and Bayes Information Criterion (BIC)

We should like a model with enough parameters to give a good fit to the data. However, we should not like to complicate the model unnecessarily by introducing parameters which the data do not provide sufficient evidence to conclude are necessary. Two quantitative criteria that have been developed to inform the judgement as to this balance between goodness of fit

and simplicity are the Akaike Information Criterion (AIC) and the Bayes Information Criterion (BIC), also called the Schwarz-Bayes Criterion (SBC). See, for example, Cairns (2000).

The AIC of a model is defined as:

$$AIC = \ell - k$$
,

where  $\ell$  is the maximum log-likelihood of the model and *k* is the number of parameters in the model. Models with a high AIC are to be preferred to those with a low AIC. The AIC is derived from the classical problem of testing the null hypothesis that the data are adequately described by a simple model  $M_1$  against the alternative of a more complex model,  $M_2$ , which contains  $M_1$  as a special case. The AIC does not require us to formulate any subjective prior beliefs as to which of the proposed models are most likely to be appropriate, and this might be thought to be an advantage. However, a disadvantage of the AIC is that its derivation assumes that the models being compared are nested, and so does not address the question of whether it is appropriate to use the AIC to compare two models that are not nested.

The BIC of a model is defined as:

$$BIC = \ell - \frac{1}{2}k \log N,$$

where  $\ell$  and *k* are as above and *N* is the number of data cells, *i.e.* N = 3,599 for CMI data and 2,640 for England and Wales data. Models with a high BIC are to be preferred to those with a low BIC. The BIC is derived using Bayesian theory, starting from a prior belief that all the models under consideration are equally likely to be the correct one and comparing the posterior probabilities of the different models. Note that the BIC penalises complexity more severely than the AIC, particularly when, as in this case, *N* is large. The prior assumption that all models are equally likely could be modified subjectively if this were considered necessary, by assigning different prior probabilities to different models, but such subjectivity might still be considered undesirable.

Comparisons of the AIC and BIC between different models can be used to gain an indication of how significant certain features of the data are as well as to inform the choice of model. Therefore, in the tables below, we shall also show the AIC and BIC of the 'static GM(0,2)' model, in which the  $\kappa^{(3)}$  and  $\kappa^{(4)}$  parameters do not depend on calendar year, so that the force of mortality at age *x* for all calendar years *t* is:

$$\mu_{xt} = \exp[a + b(x - x)],$$

where *a* and *b* are parameters to be determined. The fitted values of *a* and *b* are shown in Table 2.2.

| Data set   | а       | b       |
|------------|---------|---------|
| CMI        | -4.5348 | 0.10410 |
| E&W Male   | -4.1818 | 0.09571 |
| E&W Female | -4.7492 | 0.10075 |

| Table 2.2 – Fitted values of the p | parameters a and b for the | static GM(0,2) model |
|------------------------------------|----------------------------|----------------------|
|------------------------------------|----------------------------|----------------------|

The improvement in the AIC and BIC from the static GM(0,2) model to the GM(0,2) model, compared with the improvements from the GM(0,2) model to the more complex GM models, will give an indication of the relative significance of the improvement of mortality over time and the nonlinearity of the shape of the graph of log mortality against age.

Table 2.3 shows the values of both the AIC and BIC for the GM models that were fitted to the CMI data.

|         | Maximum        |                      |            |            |
|---------|----------------|----------------------|------------|------------|
| Model   | log-likelihood | Number of parameters | AIC        | BIC        |
| Static  |                |                      |            |            |
| GM(0,2) | -65,320.81     | 2                    | -65,322.81 | -65,329.00 |
| GM(0,2) | -19,237.78     | 118                  | -19,355.78 | -19,720.90 |
| GM(1,2) | -18,536.30     | 177                  | -18,713.30 | -19,260.98 |
| GM(1,3) | -16,644.01     | 236                  | -16,880.01 | -17,610.25 |
| GM(2,3) | -16,257.82     | 295                  | -16,552.82 | -17,465.61 |
| GM(3,3) | -16,202.21     | 354                  | -16,556.21 | -17,651.56 |
| GM(2,4) | -16,181.63     | 354                  | -16,535.63 | -17,630.98 |

| Table 2.3 – AIC ar | nd BIC values | for a number | of possible | GM models | fitted to CMI dat | а |
|--------------------|---------------|--------------|-------------|-----------|-------------------|---|
|                    |               |              | 01 00001010 |           | necou to onni uut | ~ |

We conclude from Table 2.3 that each of the models up to and including GM(2,3) achieves a statistically significant improvement in fit over the previous one, under both the BIC and the AIC. However neither GM(3,3) nor GM(2,4) achieves a statistically significant improvement over GM(2,3) under the BIC, and only GM(2,4) achieves a marginal improvement over GM(2,3) under the AIC.

We also conclude from Table 2.3 that the improvement of mortality over time is a much more significant feature of the data than nonlinearity of the graph of log mortality against age, as moving from the static GM(0,2) model to the GM(0,2) model increases the AIC and BIC much more than moving from the GM(0,2) model to the more complex models. This suggests that there is unlikely to be justification in any application for failing to allow for the improvement of mortality over time. Failure to allow for this feature will result in unduly low reserves and prices being calculated for annuity business and, potentially, unduly high reserves and prices for assurance business. However, in applications where only an approximate result is required, it may be appropriate to assume that mortality is a linear function of age for each calendar year with parameters depending on the year, *i.e.* to use the GM(0,2) model.

The fact that the GM(2,3) model achieves a statistically significant improvement in fit over the GM(1,3) model, under both the BIC and the AIC, demonstrates that there is more structure in the data than is reflected in the GM(1,3) model. However, this does not necessarily imply that the GM(2,3) model is the most appropriate model to use for future projections. In general, if the fitted parameters under a model show features which we are unable to interpret, then we are unable to assess whether or not these features are likely to be repeated in the future, and hence whether or not it is appropriate to include them in future projections. In the case of the GM(2,3) model, interpretation of the changes in the kappa parameter values over time is significantly complicated by the strong correlations between them, and this would tend to support the use of the GM(1,3) model rather than the GM(2,3) model for projections.

Table 2.4 shows the values of the AIC and BIC for the GM models that were fitted to England and Wales male data, and Table 2.5 shows the values for England and Wales female data.

|         | Maximum        |                      |             |             |
|---------|----------------|----------------------|-------------|-------------|
| Model   | log-likelihood | Number of parameters | AIC         | BIC         |
| Static  |                |                      |             |             |
| GM(0,2) | -309,023.91    | 2                    | -309,025.91 | -309,031.79 |
| GM(0,2) | -47,095.64     | 88                   | -47,183.64  | -47,442.29  |
| GM(1,2) | -34,306.16     | 132                  | -34,438.16  | -34,826.14  |
| GM(1,3) | -18,478.49     | 176                  | -18,654.49  | -19,171.81  |
| GM(2,3) | -17,751.70     | 220                  | -17,971.70  | -18,618.34  |
| GM(2,4) | -17,181.03     | 264                  | -17,445.03  | -18,221.00  |
| GM(3,4) | -16,937.19     | 308                  | -17,245.19  | -18,150.48  |
| GM(4,4) | -16,785.46     | 352                  | -17,137.46  | -18,172.08  |

 Table 2.4 – AIC and BIC values for a number of possible GM models fitted to England and

 Wales male data

|         | Maximum        |                      |             |             |
|---------|----------------|----------------------|-------------|-------------|
| Model   | log-likelihood | Number of parameters | AIC         | BIC         |
| Static  |                |                      |             |             |
| GM(0,2) | -177,258.66    | 2                    | -177,260.66 | -177,266.54 |
| GM(0,2) | -26,688.23     | 88                   | -26,776.23  | -27,034.89  |
| GM(1,2) | -21,918.09     | 132                  | -22,050.09  | -22,438.07  |
| GM(1,3) | -18,911.86     | 176                  | -19,087.86  | -19,605.17  |
| GM(2,3) | -17,705.78     | 220                  | -17,925.78  | -18,572.42  |
| GM(2,4) | -16,411.35     | 264                  | -16,675.35  | -17,451.32  |
| GM(3,4) | -16,140.94     | 308                  | -16,448.94  | -17,354.23  |
| GM(4,4) | -15,872.44     | 352                  | -16,224.44  | -17,259.06  |

# Table 2.5 – AIC and BIC values for a number of possible GM models fitted to England and Wales female data

The GM(4,4) model was found to be the most complex model for which the iterative scheme converged, for both male and female England and Wales data.

Given the greater size of the England and Wales data sets than of the CMI data set, it was expected that more parameters would be statistically significant in Tables 2.4 and 2.5 than in Table 2.3. In fact, for both the male and the female England and Wales data sets, the most complex model we have fitted, the GM(4,4) model, has been determined to be optimal under both the AIC and the BIC, with the exception that the BIC of the GM(4,4) model fitted to male data is slightly less than that of the GM(3,4) model. However, the improvement of mortality over time is again a much more significant feature of the data than nonlinearity of the graph of log mortality against age.

As with the CMI data, the conclusion to be drawn from the fact that more complex models can improve the AIC and BIC compared with the GM(1,3) model is that there is additional structure in the data, rather than that it is appropriate to base projections on those more complex models. For similar reasons to those for the CMI data, the GM(1,3) model is more likely than the GM(2,3) model to be an appropriate basis for projections. However, as consideration of cohort effects was necessary to explain the shapes of the parameter graphs under the GM(1,3) model for both males and females, we should also investigate models that make explicit allowance for cohort effects. This will be the subject of Chapter 3.

### 2.6 Standardised residuals

For a given model fitted to a given data set, the standardised residual at age x in calendar year *t*,  $Z_{xt}$ , is defined as:

$$\frac{D_{xt} - \hat{\mu}_{xt} E_{xt}}{\sqrt{\hat{\mu}_{xt} E_{xt}}},$$

where  $\hat{\mu}_{xt}$  is the fitted force of mortality at age *x* in calendar year *t*. The  $\hat{\mu}_{xt}E_{xt}$  in the numerator represents the mean of the fitted Poisson distribution of the number of deaths at age *x* in calendar year *t* and the  $\hat{\mu}_{xt}E_{xt}$  in the denominator represents the variance of this Poisson distribution.

If the model under consideration were a perfect representation of reality, then the standardised residuals would be realisations of independent random variables with mean 0 and variance 1. In practice, the sample variance of the standardised residuals would be expected to be greater than 1, as there will always be some patterns of variation in the data that are not captured by the model. Table 2.6 shows the sample variances of the standardised residuals observed for each of the GM models fitted to the CMI data. These have been calculated as:

$$\frac{1}{N-k}\sum_{x}\sum_{t}(Z_{xt}-\bar{z})^2,$$

where *N* is the number of data cells, *k* is the number of parameters and  $\overline{z}$  is the sample mean of the standardised residuals, rather than as:

$$\frac{1}{N}\sum_{x}\sum_{t}(Z_{xt}-\bar{z})^2,$$

*i.e.* they are unbiased estimates rather than maximum likelihood estimates of the population variance of the standardised residuals.

| Table 2.6 – Sample variances of the standardised residuals for a number of possible ( | GΜ |
|---|----|
| models fitted to CMI data   |    |

| Model          | Sample variance of standardised residuals |
|----------------|---|
| Static GM(0,2) | 28.3338                                   |
| GM(0,2)        | 3.5309                                    |
| GM(1,2)        | 3.1662                                    |
| GM(1,3)        | 2.0829                                    |
| GM(2,3)        | 1.8843                                    |
| GM(3,3)        | 1.8847                                    |
| GM(2,4)        | 1.8723                                    |

It can be seen from Table 2.6 that the fit, as measured by the sample variance of the standardised residuals, improves significantly as we move from the GM(0,2) model to the GM(1,2) model and, particularly, from the GM(1,2) model to the GM(1,3) model. The improvement from the GM(1,3) model to the GM(2,3) model is rather smaller. There is very little improvement beyond the GM(2,3) model – in fact there is a deterioration for the GM(3,3) model compared with the GM(2,3) model as the decrease in the sum of squares is not sufficient to outweigh the increase in the number of parameters.

Table 2.7 shows the sample variances of the standardised residuals observed for each of the GM models fitted to the England and Wales male data, and Table 2.8 shows the same information for the England and Wales female data.

 Table 2.7 – Sample variances of the standardised residuals for a number of possible GM models fitted to England and Wales male data

| Model          | Sample variance of standardised residuals |
|----------------|---|
| Static GM(0,2) | 221.3328                                  |
| GM(0,2)        | 26.4706                                   |
| GM(1,2)        | 17.2309                                   |
| GM(1,3)        | 4.5875                                    |
| GM(2,3)        | 4.0648                                    |
| GM(2,4)        | 3.6561                                    |
| GM(3,4)        | 3.5154                                    |
| GM(4,4)        | 3.4497                                    |

| Model          | Sample variance of standardised residuals |
|----------------|---|
| Static GM(0,2) | 124.5906                                  |
| GM(0,2)        | 11.3289                                   |
| GM(1,2)        | 7.5568                                    |
| GM(1,3)        | 5.2179                                    |
| GM(2,3)        | 4.3146                                    |
| GM(2,4)        | 3.2973                                    |
| GM(3,4)        | 3.1266                                    |
| GM(4,4)        | 2.9505                                    |

 Table 2.8 – Sample variances of the standardised residuals for a number of possible GM models fitted to England and Wales female data

The figures in Tables 2.7 and 2.8 are higher than the corresponding figures for the same model in Table 2.6. This is to be expected because the greater size of the England and Wales data sets than of the CMI data set makes it likely that more features of the data not captured by the model are will be statistically significant in the former than in the latter. It is also likely that the higher figures in Tables 2.7 and 2.8 than in Table 2.6 are partly due to lower accuracy of the exposed to risk in the England and Wales data than in the CMI data. The CMI calculates the exposed to risk in its data sets every year, whereas a census of the England and Wales population is only carried out every ten years, with the exposed to risk being estimated in intermediate years.

Tables 2.7 and 2.8 show very significant improvements in fit for both males and females as we move from the GM(0,2) to the GM(1,2) model, and as we move from the GM(1,2) to the GM(1,3) model. From the GM(1,3) to the GM(2,3) model and from the GM(2,3) to the GM(2,4) model, the improvement in fit is fairly significant for females, but less so for males. Beyond the GM(2,4) model, the improvements are relatively small for both males and females.

### 2.7 Coefficient of determination, $R^2$

For each of the data sets, it is instructive to examine what proportion of the variation in the crude force of mortality between data cells is explained by each of the models. The measure generally used for this purpose is called the *coefficient of determination* and is denoted by  $R^2$ .

We define the *residual sum of squares* (*RSS*) to be the sum of the squares of the standardised residuals:

$$RSS = \sum_{x} \sum_{t} Z_{xt}^2 \; .$$

We also define:

$$RSS_{0} = \sum_{x} \sum_{t} \frac{\left(D_{xt} - \frac{D_{++}E_{xt}}{E_{++}}\right)^{2}}{\frac{D_{++}E_{xt}}{E_{++}}},$$

where:

$$D_{++} = \sum_{x} \sum_{t} D_{xt}$$

and:

$$E_{++} = \sum_{x} \sum_{t} E_{xt} \, .$$

Thus  $RSS_0$  is the RSS of the model in which  $\mu_{xt}$  is a constant,  $\mu$ , independent of both age x and calendar year t. The maximum likelihood estimate of  $\mu$  under this model is  $\frac{D_{++}}{E_{++}}$ .

 $R^2$  is then defined as:

$$R^2 = 1 - \frac{RSS}{RSS_0}.$$

Thus  $R^2 = 1$  in a model where the graduated force of mortality equals the observed force of mortality exactly in every data cell, and  $R^2 = 0$  if  $\hat{\mu}_{xt} = \frac{D_{++}}{E_{++}}$  for all *x* and all *t*.

The values of  $R^2$  for the different models fitted to CMI data, to England and Wales male data and to England and Wales female data are shown in Tables 2.9, 2.10 and 2.11 respectively. In Tables 2.9, 2.10 and 2.11, 'Flat' represents the model where  $\mu_{xt}$  is independent of both age *x* and calendar year *t*.

| Table 2.9 – Values of R | <sup>2</sup> for a number of | possible GM models | fitted to CMI data |
|-------------------------|------------------------------|--------------------|--------------------|
|-------------------------|------------------------------|--------------------|--------------------|

| Model          | $R^2$    |
|----------------|----------|
| Flat           | 0        |
| Static GM(0,2) | 0.981448 |
| GM(0,2)        | 0.997763 |
| GM(1,2)        | 0.998028 |
| GM(1,3)        | 0.998725 |
| GM(2,3)        | 0.998867 |
| GM(3,3)        | 0.998887 |
| GM(2,4)        | 0.998894 |

| <b>Table 2.10 –</b> Values of $R^2$ | for a number | of possible | GM models | fitted to | England a | and Wales |
|-------------------------------------|--------------|-------------|-----------|-----------|-----------|-----------|
|                                     |              | male data   |           |           |           |           |

| Model          | $R^2$    |
|----------------|----------|
| Flat           | 0        |
| Static GM(0,2) | 0.981410 |
| GM(0,2)        | 0.997849 |
| GM(1,2)        | 0.998624 |
| GM(1,3)        | 0.999640 |
| GM(2,3)        | 0.999687 |
| GM(2,4)        | 0.999723 |
| GM(3,4)        | 0.999739 |
| GM(4,4)        | 0.999749 |

| Model          | $R^2$    |
|----------------|----------|
| Flat           | 0        |
| Static GM(0,2) | 0.989749 |
| GM(0,2)        | 0.999098 |
| GM(1,2)        | 0.999409 |
| GM(1,3)        | 0.999599 |
| GM(2,3)        | 0.999674 |
| GM(2,4)        | 0.999756 |
| GM(3,4)        | 0.999773 |
| GM(4,4)        | 0.999789 |

# **Table 2.11 –** Values of $R^2$ for a number of possible GM models fitted to England and Walesfemale data

For all three data sets, by far the greatest increase in  $R^2$  occurs when we move from the flat model to the static GM(0,2) model. In each case, over 98% of the variation in the crude force of mortality between data cells is explained when we move from assuming that mortality is independent of age to assuming that it increases exponentially with age, without making any allowance for the improvement of mortality over time.

The GM(0,2) model, in which mortality increases exponentially with age for each calendar year with different parameters for each calendar year, accounts for over 99.75% of the variation in the crude force of mortality between data cells for all three data sets. However, we saw in Section 2.5 that there is still sufficient structure in the residuals of the GM(0,2) model to justify a more complex model, as more complex models improved the BIC and AIC compared with the GM(0,2) model.

Although Section 2.3.1 concluded that models more complex than the GM(1,3) model were overfitted to CMI data, the GM(1,3) model has a rather lower value of  $R^2$  than for England and Wales data. This can be attributed to the fact that, as the CMI data is a smaller data set than the England and Wales data sets, more of the variation in the crude force of mortality between data cells is a result of stochastic fluctuations.

### 2.8 Residual plots

For each of the GM models fitted to each of the data sets, it is instructive to examine which ages and calendar years have observed mortality rates above the modelled mortality rates (*i.e.* positive standardised residuals), and which have observed mortality rates below the modelled mortality rates (*i.e.* negative standardised residuals). For CMI data, these ages and years are shown respectively as red and blue in the graphs in Figure 2.18. Darker shades of red and blue are used to highlight the more extreme residuals.

The key observations from Figure 2.18(i) are that the GM(0,2) model systematically understates mortality at the very youngest ages of the data set, systematically overstates it around ages 35-50, systematically understates it around ages 55-65 and systematically overstates it above age 80. It is clear that modelling the force of mortality as an exponential of a linear function of age for each calendar year is inadequate to capture the structure of the data set. There is some evidence of diagonal patterns in the residuals, in that the age bands in which there is systematic understatement and systematic overstatement increase towards the end of the period 1947-2005. It seems likely that much of this reflects the favourable cohort effect centred on 1926. There is also a possible adverse cohort effect for lives born after 1945 – as discussed below, a similar effect is also visible for England and Wales male data.

Figure 2.18(ii) shows that the addition of  $\kappa^{(0)}$  parameters is not in itself sufficient to eliminate the above inadequacies in the GM(0,2) model. The GM(1,2) model shows systematic discrepancies between observed and fitted mortality at broadly the same ages as the GM(0,2) model. This is despite the statistically significant improvement the GM(1,2) model showed in goodness of fit, under both the BIC and AIC, compared with the GM(0,2) model. There is still
some evidence, though slightly weaker than for the GM(0,2) model, of diagonal patterns in the residuals.

The generally paler colour of Figures 2.18(iii) and (iv) compared with Figures 2.18(i) and (ii) indicates that the systematic overstatements of mortality at certain ages and the systematic understatements at others are much reduced in the GM(1,3) model and, particularly, the GM(2,3) model, compared with the GM(1,2) model. However, one feature that remains is a systematic understatement of mortality around ages 60-65 and a systematic overstatement around ages 65-70. As we shall see later in this section, this feature is specific to the CMI data and is not observed with England and Wales data. A plausible explanation for this feature is that the policies around ages 60-65 are mostly endowments maturing at or near age 65, whereas the policies around ages 65-70 are mostly whole life assurances. The life insurance companies that contributed to the CMI's investigation may have sold endowments and whole life assurances to different market segments, possibly with stricter underwriting for whole life assurances, or the understatement of mortality in the CMI data resulting from unclaimed policies may have been more significant for whole life than for endowment business. As shown in Figure 1.1(i), the exposed to risk in the CMI data decreases rapidly with increasing age around age 65, which is consistent with this explanation. The author contacted the CMI regarding this feature and received the following response:

'As far as we are aware this feature has not previously been investigated or brought to the attention of the CMI. Unfortunately, the scheduled nature of the data provided to the CMI means that we are unable to give firm answers on why this might be occurring, for example we cannot separately identify endowment business and whole of life business. However, most endowment business would be written to mature by age 65 and so it may be reasonable to treat the data after age 65 as arising mainly from whole life policies.'

This feature of the CMI data has, however, become less significant since the early 1990s.

The diagonal patterns that we observed in Figures 2.18(i) and (ii) have been largely eliminated in Figures 2.18(iii) and (iv). This suggests that cohort effects in the CMI data are largely captured indirectly via the parameter estimates of the GM(1,3) and GM(2,3) models. When we discussed the parameter graphs in Section 2.3, we did not identify cohort effects as a significant driver of the shapes of the graphs. However, further inspection of Figure 2.3 in particular reveals a particularly steep fall in the  $\kappa^{(3)}$  parameters, and a particularly steep rise in the  $\kappa^{(4)}$  parameters, from 1980 to the early 1990s. In Section 2.3 we explained a similar feature of the England and Wales data sets by reference to cohort effects.

Figures 2.18(v) and (vi) reveal very little change compared with Figure 2.18(iv). This is consistent with the fact that the GM(3,3) model failed to improve the BIC or AIC compared with the GM(2,3) model, and the GM(2,4) model failed to improve the BIC and improved the AIC only marginally.

Figures 2.19(i), (ii), (iii) and (iv) are similar residual plots for the England and Wales male data, for the GM(0,2), GM(1,2), GM(1,3) and GM(2,3) models respectively. The residual plots for more complex models than GM(2,3) have not been shown, as we concluded in Section 2.3 that even the GM(2,3) model was overfitted. Figures 2.20(i)-(iv) are the corresponding plots for the female data.

The standardised residuals for a given model tend to be larger in magnitude for England and Wales data than for CMI data. This is again a consequence of the larger population underlying the England and Wales data than underlying the CMI data.

Cohort effects, represented by diagonal bands, are much more apparent in the residual plots for England and Wales data than for CMI data. As mentioned in Section 1 and in Section 2.3, the cohort centred on year of birth 1931 has been identified as exhibiting particularly large mortality improvements compared with the previous generation, for both England and Wales males and England and Wales females. In fact, if we draw the diagonals on Figures 2.19 and 2.20 representing year of birth 1931, then we find that the area close to and above the diagonals tends to be red, representing observed mortality above the modelled mortality for

years of birth just before 1931, and the area close to and below the diagonals tends to be blue, representing observed mortality below the modelled mortality for years of birth just after 1931. This is consistent with what we would expect if the cohort centred on year of birth 1931 exhibits particularly large mortality improvements compared with the previous generation.

Section 2.3 also mentioned that females born around 1915 have exhibited particularly low mortality improvements compared with the previous generation. In fact, if we draw the diagonals representing year of birth 1915 on Figure 2.20, then we find that the area close to and above the diagonals is mostly blue, while the area close to and below the diagonals is red or lighter blue. This is consistent with females born around 1915 exhibiting particularly low mortality improvements compared with the previous generation.

One feature that can be seen in all the residual plots for England and Wales data, for both males and females, is that lives born in 1919 have exhibited unusually low mortality and lives born in 1920 have exhibited unusually high mortality. Renshaw and Haberman (2006) observe a similar feature and attribute it to the 1919 influenza epidemic. It seems plausible that this epidemic could have had either a favourable impact on the mortality of the cohort *born* during or shortly after the epidemic, by reducing the chances of unhealthy children being born alive and/or surviving past infancy, or an adverse impact, because the epidemic had a permanent impact on the health of this cohort. However, it is by no means clear that the epidemic would have had a favourable impact on the mortality of lives born in 1919 and an adverse impact on the mortality of lives born in 1920. This is an area that would benefit from further research.

Figure 2.19(i) also suggests a possible adverse cohort effect for England and Wales males born since 1945. This feature is no longer apparent in Figure 2.19(ii), having been captured indirectly via  $\kappa^{(0)}$  parameters that increase over time. Paragraph 2.15.6 of Willets *et al.* (2004) suggests the possible existence of such an adverse cohort effect for some generations born since 1945, but it cannot be said that there is sufficient evidence to demonstrate its existence definitively.

We shall now discuss each of the graphs in Figures 2.19 and 2.20 individually. From Figure 2.19(i), we see that the ages at which the GM(0,2) model systematically overstates and understates England and Wales male mortality are fairly similar to those for CMI data. However, with England and Wales male data there is no systematic understatement of mortality at the very youngest ages of the data set in the early years of the period 1962-2005, although such an understatement does appear in the later years. The systematic understatement at ages 55-65 that existed in CMI data has moved to the age range 60-75. There is stronger evidence than for CMI data of cohort effects giving rise to increases in the age bands of systematic overstatement and understatement in the later years of the data set.

Figure 2.19(ii) shows that the GM(1,2) model fitted to England and Wales male data systematically overstates and understates mortality at broadly the same ages as the GM(0,2) model. However, there is now a systematic understatement at the youngest ages in the earlier years as well as the later years of the data set, and elsewhere there is a limited amount of weakening of both the systematic overstatements and understatements.

Figure 2.19(iii) is significantly paler than Figures 2.19(i) and (ii), and the systematic overstatements and understatements of mortality at certain ages have largely been eliminated, indicating that the GM(1,3) model explains significantly more of the structure of the England and Wales male data than the GM(0,2) and GM(1,2) models. Some diagonal patterns remain, indicating cohort effects. We observed in Section 2.3 that the parameter estimates reflected the impact of cohort effects indirectly, but Figure 2.19(iii) shows that the parameter estimates captured only part of the cohort effects.

A comparison of Figure 2.19(iv) with Figure 2.19(iii) reveals some differences in specific regions, but it is unlikely that such a comparison would be used to justify the use of the GM(2,3) model in preference to the GM(1,3) model for England and Wales male data.

Comparing Figures 2.20(i) and (ii) with Figures 2.19(i) and (ii) shows that the ages for which the GM(0,2) and GM(1,2) models systematically overstate and understate England and Wales female mortality are broadly the opposite to those for England and Wales male mortality. In other words, mortality is systematically understated over the age range 35-50, systematically overstated over the age range 60-75 and systematically understated above age 80. As we move from the GM(0,2) to the GM(1,2) model, a systematic understatement of mortality at the youngest ages of the data set is replaced by a systematic overstatement.

The observations from Figures 2.20(iii) and (iv) are similar to those from Figures 2.19(iii) and (iv). There are signs that the GM(2,3) model achieves slightly more improvement in fit over the GM(1,3) model for females than for males. However, particularly as this greater improvement in fit is less significant in the later years of the data set, it is unlikely to be a sufficient reason to apply the GM(2,3) rather than the GM(1,3) model to England and Wales female data.

As an alternative visualisation of the structure of the standardised residuals, Figures 2.18-2.20 were projected onto the age axis and the year of birth diagonal to produce scatter diagrams. Figures 2.21 and 2.22 are the projections of the CMI standardised residuals onto the age axis and the year of birth diagonal respectively. Figures 2.23 and 2.24 are the projections of the England and Wales male standardised residuals onto the age axis and the year of birth diagonal respectively. Figures 2.26 are the projections of the England and Wales female standardised residuals onto the age axis and the year of birth diagonal respectively. Figures 2.26 are the projections of the England and Wales female standardised residuals onto the age axis and the year of birth diagonal respectively.

The following observations can be made from Figures 2.21-2.26:

- As we move from the simpler models to the more complex models, there are progressively fewer standardised residuals that are large in absolute value. This is to be expected because introducing each new time series of parameters captures more of the structure of the data.
- Figures 2.21(i) and (ii) show that, in the GM(0,2) and GM(1,2) models fitted to CMI data, the standardised residuals close to age 30 are mostly positive, those around age 40 are mostly negative, those around age 60 are mostly positive and those close to age 90 are mostly negative. This indicates that these models systematically understate mortality close to age 30, systematically overstate it around age 40, systematically understate it around age 60 and systematically overstate it close to age 90. This is consistent with what we observed from Figures 2.18(i) and (ii). As we move to the GM(1,3) model and to the more complex models, these systematic patterns in the sign of the residuals are largely eliminated.
- Even in the most complex models fitted to the CMI data, there is still evidence of the residuals being systematically positive in the age range 60-65 and negative in the age range 65-70. This feature was commented on in the discussion of Figures 2.18(iii) and (iv), where it was suggested that the explanation was that the exposed to risk at ages 60-65 consisted primarily of endowments, while that at ages 65-70 consisted primarily of whole life assurances. This feature is not observed in England and Wales data, which is what we would expect if this explanation is correct.
  - For the CMI data, the graphs for the GM(3,3) and GM(2,4) models are very similar to those for the GM(2,3) model. For the England and Wales data, the graphs for more complex models than GM(2,3) have not been shown. Some differences between the graphs for the GM(1,3) and GM(2,3) models are visible for both CMI data and England and Wales data, but none of these differences is likely to be regarded as significant grounds for preferring one model to the other.
  - In Figure 2.24, relating to England and Wales male data, and Figure 2.26, relating to England and Wales female data, the points relating to year of birth 1919 tend to have large negative standardised residuals and the points relating to year of birth 1920 tend to have large positive standardised residuals. As noted in the discussion of

Figures 2.19 and 2.20, this is likely to be due to the 1919 influenza epidemic. The same effect can be seen to some extent in CMI data in Figure 2.22, although it is less prominent.

- In England and Wales data compared with CMI data, there are more standardised residuals that are large in absolute value. As discussed in Section 2.6, this can be explained by the larger size of the England and Wales data sets than of the CMI data set, in that the England and Wales data are likely to contain more statistically significant features not captured by the models, and the more approximate procedure for calculating the exposed to risk for the England and Wales data than for the CMI data is also likely to have an effect.
- For the GM(0,2) model fitted to England and Wales male data, Figure 2.23(i) indicates that the residuals around age 70 are almost all positive, and that the residuals close to age 89 are almost all negative. This agrees with what we observed from Figure 2.19(i), namely a systematic understatement of mortality around age 70 and a systematic overstatement at ages close to 89. In both cases, it is necessary to introduce  $\kappa^{(5)}$  parameters as well as  $\kappa^{(0)}$  parameters to make the bias insignificant. Figure 2.23 also shows that the GM(0,2) and GM(1,2) models do not adequately capture the age structure of the England and Wales male data at younger ages, but that the GM(1,3) and GM(2,3) models give greatly improved results.
  - Cohort effects can be seen under the simpler models in all three of the figures 2.22, 2.24 and 2.26, over the ranges of years of birth that we would expect from our earlier remarks. For CMI data, the fall in standardised residuals which can be seen in Figure 2.22(i), representing the favourable cohort effect centred on 1926, has largely been eliminated in the more complex models, indicating that most of this cohort effect has been captured implicitly via the shape of the kappa parameter graphs. Again for CMI data, Figures 2.22(i) and (to a lesser extent) (ii) show an increase in standardised residuals for lives born since 1945, but this has been captured implicitly via the kappa parameter estimates in the more complex models. Figure 2.24(i), for England and Wales male data, also shows an increase in standardised residuals for lives born since 1945, but this time introducing  $\kappa^{(0)}$  parameters is sufficient to capture this effect implicitly. However, in the case of the favourable cohort effect centred on 1931 for both England and Wales males and England and Wales females, and the adverse cohort effect centred on 1915 for England and Wales females, many of the cohort effects remain in the residuals of the more complex models. For England and Wales males, the significant fall in the standardised residuals centred on 1931 visible in Figures 2.24(i) and (ii) has been noticeably reduced, but by no means eliminated, in Figures 2.24(iii) and (iv). For England and Wales females, the GM(1,3) and GM(2,3) models are still less successful at capturing the fall in the standardised residuals centred on 1931 visible under the GM(0,2) and GM(1,2) models. Also for England and Wales females, the GM(1,3) and GM(2,3) models have only limited success at capturing the rise in standardised residuals from 1905 to 1925, which is interrupted by the 1919-20 discontinuity.
  - From Figure 2.25(i), the residuals when the GM(0,2) model is fitted to England and Wales female data are mostly positive up to age 50 and above age 80 and negative around age 70. This indicates systematic understatement of mortality up to age 50 and above age 80 and systematic overstatement around age 70, consistently with Figure 2.20(i). From Figure 2.25(ii), when  $\kappa^{(0)}$  parameters are introduced, the residuals close to age 30 become mostly negative, those around ages 50 and 70 mostly remain positive and negative respectively and those close to age 89 become negative at certain years (actually in earlier years, although this cannot be seen from Figure 2.25(ii)). This agrees with Figure 2.20(ii).
  - In the GM(1,3) and GM(2,3) models fitted to England and Wales female data (Figures 2.25(iii) and (iv)) rather than male data (Figures 2.23(iii) and (iv)), there is rather more age structure remaining in the residuals. However, this should not be taken as support for the use of a more complex GM model than the GM(2,3) model for

England and Wales female data. We observed in Section 2.3 that even in the GM(2,3) model, there were significant problems of overfitting.

Some negative standardised residuals can be seen at the oldest ages in all the graphs in Figure 2.25, relating to England and Wales female data. It seems likely that these are due to issues with the quality of the data, with ages at death in the late 80s that were not known exactly being reported as 90 which is outside the range of the graphs. The graphs in Figure 2.20 suggest that these negative standardised residuals occur mainly in the earlier years of the data set, so that this issue has become less significant in more recent years.

## 2.9 Conclusion

A number of different GM models have been fitted to the three data sets. It was found that the parameter graphs for the GM(1,3) model had a clear interpretation but that the graphs for more complex GM models were much more difficult to interpret. In addition, no major features of the data have been identified that were not captured in the GM(1,3) model but were in more complex GM models. Therefore, despite the fact that some more complex GM models improved the AIC and BIC compared with the GM(1,3) model, we conclude that the GM(1,3) model is the most appropriate model on which to base future mortality projections. Cohort effects are reflected indirectly via the fitted parameters of this model, but there are still some cohort effects visible in the residuals, particularly for the England and Wales data sets. We shall next extend the model to incorporate an explicit allowance for these cohort effects.

**Figure 2.1 –** Maximum likelihood parameter estimates for the GM(0,2) model fitted to CMI data –  $\mu_{xt} = \exp[\kappa_t^{(3)} + \kappa_t^{(4)}(x - \bar{x})] - (i) \kappa_t^{(3)}$ , (ii)  $\kappa_t^{(4)}$ 

(i)



(ii)









**Figure 2.2** – Maximum likelihood parameter estimates for the GM(1,2) model fitted to CMI data –  $\mu_{xt} = \kappa_t^{(0)} + \exp[\kappa_t^{(3)} + \kappa_t^{(4)}(x - \overline{x})]$  – (i)  $\kappa_t^{(0)}$ , (ii)  $\kappa_t^{(3)}$ , (iii)  $\kappa_t^{(4)}$ 



**Figure 2.3 –** Maximum likelihood parameter estimates for the GM(1,3) model fitted to CMI data –  $\mu_{xt} = \kappa_t^{(0)} + \exp[\kappa_t^{(3)} + \kappa_t^{(4)}(x - \overline{x}) + \kappa_t^{(5)}((x - \overline{x})^2 - \hat{\sigma}_x^2)]$  – (i)  $\kappa_t^{(0)}$ , (ii)  $\kappa_t^{(3)}$ , (iii)  $\kappa_t^{(4)}$ , (iv)  $\kappa_t^{(5)}$ 

















-2e-05

-4e-05

Year, t Year, t

**Figure 2.7 –** Maximum likelihood parameter estimates for the GM(0,2) model fitted to England and Wales male data –  $\mu_{xt} = \exp[\kappa_t^{(3)} + \kappa_t^{(4)}(x - \bar{x})]$  – (i)  $\kappa_t^{(3)}$ , (ii)  $\kappa_t^{(4)}$ 

(i)



(ii)











**Figure 2.9** – Maximum likelihood parameter estimates for the GM(1,3) model fitted to England and Wales male data –  $\mu_{xt} = \kappa_t^{(0)} + \exp[\kappa_t^{(3)} + \kappa_t^{(4)}(x - \overline{x}) + \kappa_t^{(5)}((x - \overline{x})^2 - \hat{\sigma}_x^2)]$ – (i)  $\kappa_t^{(0)}$ , (ii)  $\kappa_t^{(3)}$ , (iii)  $\kappa_t^{(4)}$ , (iv)  $\kappa_t^{(5)}$ 





Year, t

**Figure 2.11 –** Maximum likelihood parameter estimates for the GM(0,2) model fitted to England and Wales female data –  $\mu_{xt} = \exp[\kappa_t^{(3)} + \kappa_t^{(4)}(x - \bar{x})]$  – (i)  $\kappa_t^{(3)}$ , (ii)  $\kappa_t^{(4)}$ 

(i)



(ii)







**Figure 2.12** – Maximum likelihood parameter estimates for the GM(1,2) model fitted to England and Wales female data –  $\mu_{xt} = \kappa_t^{(0)} + \exp[\kappa_t^{(3)} + \kappa_t^{(4)}(x - \overline{x})]$  – (i)  $\kappa_t^{(0)}$ , (ii)  $\kappa_t^{(3)}$ , (iii)  $\kappa_t^{(4)}$ 





Figure 2.13 - Maximum likelihood parameter estimates for the GM(1,3) model fitted to









Figure 2.16 – 2000 mortality curves fitted to England and Wales male data – differences from the GM(0,2) model – (i) GM(0,2), (ii) GM(1,2), (iii) GM(1,3), (iv) GM(2,3)



**Figure 2.17** – 2000 mortality curves fitted to England and Wales female data – differences from the GM(0,2) model –(i) GM(0,2), (ii) GM(1,2), (iii) GM(1,3), (iv) GM(2,3)







**Figure 2.19 –** Plots of standardised residuals for a number of possible GM models fitted to England and Wales male data – (i) GM(0,2), (ii) GM(1,2), (iii) GM(1,3), (iv) GM(2,3)



**Figure 2.20 –** Plots of standardised residuals for a number of possible GM models fitted to England and Wales female data – (i) GM(0,2), (ii) GM(1,2), (iii) GM(1,3), (iv) GM(2,3)



Figure 2.21 – Scatter diagrams of standardised residuals plotted against age for CMI data – (i) GM(0,2), (ii) GM(1,2), (iii) GM(1,3), (iv) GM(2,3), (v) GM(3,3), (vi) GM(2,4)







Age

**Figure 2.23** – Scatter diagrams of standardised residuals plotted against age for England and Wales male data – (i) GM(0,2), (ii) GM(1,2), (iii) GM(1,3), (iv) GM(2,3)

Age



**Figure 2.24 –** Scatter diagrams of standardised residuals plotted against year of birth for England and Wales male data – (i) GM(0,2), (ii) GM(1,2), (iii) GM(1,3), (iv) GM(2,3)



**Figure 2.25** – Scatter diagrams of standardised residuals plotted against age for England and Wales female data – (i) GM(0,2), (ii) GM(1,2), (iii) GM(1,3), (iv) GM(2,3)



**Figure 2.26 –** Scatter diagrams of standardised residuals plotted against year of birth for England and Wales female data – (i) GM(0,2), (ii) GM(1,2), (iii) GM(1,3), (iv) GM(2,3)

# 3: Adding a cohort effect to the model

### **3.1 Introduction**

We observed strong evidence of a cohort effect for England and Wales data in Chapter 2. Some of this cohort effect was implicit in the shapes of the parameter graphs under the more complex models, *e.g.* the graphs in Figure 2.9 for male data and Figure 2.13 for female data, and some of it remained in the residuals and is visible in Figure 2.24 for male data and Figure 2.26 for female data. For the CMI data, we also observed evidence of a cohort effect, but it was mainly implicit in the shapes of the parameter graphs, *e.g.* Figure 2.3, rather than remaining in the residuals in Figure 2.22.

It would be desirable to allow for cohort effects via a model that incorporates them directly and places them on an equal footing with period effects. An example of such a model would be:

$$\mu_{xt} = \kappa_t^{(0)} + \gamma_{t-x}^{(0)} + \exp[\kappa_t^{(3)} + \kappa_t^{(4)}(x - \overline{x}) + \kappa_t^{(5)}((x - \overline{x})^2 - \hat{\sigma}_x^2) + \gamma_{t-x}^{(3)} + \gamma_{t-x}^{(4)}(x - \overline{x}) + \gamma_{t-x}^{(5)}((x - \overline{x})^2 - \hat{\sigma}_x^2) ]$$

However, it is found that models of this type fail to produce sensible maximum likelihood estimates of the kappa and gamma parameters. The reasons for this are essentially the same as those in Appendix B, which is referred to below.

Although we shall not fit a model of this type, we should still allow for the cohort effects that have already been observed in certain generations when projecting future mortality rates for those generations. We should also allow in stochastic projections for the risk that future generations will experience either favourable or adverse cohort effects. Accordingly we shall apply a factor depending on year of birth to the force of mortality modelled in Chapter 2. Thus the revised force of mortality at age x in year t is:

$$\mu_{xt} = \gamma_{t-x} \mu_{xt}^0 ,$$

where  $\gamma_c$  is a parameter to be estimated for each year of birth *c* and  $\mu_{xt}^0$  is the force of mortality modelled in Chapter 2. In view of the findings of Chapter 2, we shall use the GM(1,3) model for the values of  $\mu_{xt}^0$ .

Ideally, we would use a single iterative procedure to estimate both the gamma and the kappa parameters. However, if this is done, then it is found that towards the extremities of the range of years of birth, the gamma parameters reflect effects that clearly cannot be genuine cohort effects. The gamma parameters instead serve to extend the range of shapes of the mortality curve as a function of age available in each calendar year. Appendix B gives further details of the investigations that were carried out in relation to this. Given this problem, in the estimation of the gamma parameters, we shall assume that the kappa parameters have already been fixed.

#### 3.2 Restricted data sets

Consistently with the approach of Cairns *et al.* (2007), it was considered that there were insufficient data to estimate the gamma parameters reliably for years of birth observed for less than five years in the data. Accordingly, the data relating to lives born before 1861 or after 1971 were removed from the CMI data set, and the data relating to lives born before 1877 or after 1971 were removed from the England and Wales data sets. In what follows, the resulting data sets will be referred to as the 'restricted data sets'.

Before estimating the gamma parameters, it was then necessary to re-estimate the parameters of the GM(1,3) model, *i.e.* the kappa parameters, based on the restricted data

sets. Graphs of the revised parameter values are shown in Figures 3.1, 3.2 and 3.3, which are revised versions of Figures 2.3, 2.9 and 2.13 respectively. Only the parameter values for calendar years before 1951 and after 2001 have changed for CMI data, and only the parameter values for calendar years before 1966 and after 2001 have changed for England and Wales data. This is what we would expect because these are the only years for which any data have been excluded.

Figures 3.4(i), (ii) and (iii) are the graphs corresponding to Figures 2.18(iii), 2.19(iii) and 2.20(iii) respectively for the GM(1,3) model fitted to the restricted data sets, before introducing gamma parameters. The graphs in Figure 3.4 show white triangles in the top left and bottom right corners, representing the years of birth excluded from the data. Otherwise the graphs are very similar to Figures 2.18(iii), 2.19(iii) and 2.20(iii) respectively, and identical for years of birth 1951-2001 inclusive for CMI data and 1966-2001 inclusive for England and Wales data.

Figures 3.5 and 3.6 are the graphs corresponding to Figures 2.21-2.26(iii) when the GM(1,3) model is fitted to the restricted data sets, before introducing gamma parameters. Again there are no significant changes compared with the unrestricted data sets.

## 3.3 y parameters (cohort parameters)

As discussed in Section 3.1 and in Appendix B, we have selected a simple method of adjusting for a cohort effect, estimating the gamma parameters, which are applied as multiplicative factors to the fitted forces of mortality, without changing our estimates of the kappa parameters. Specifically,  $\gamma_c$  was estimated, for each year of birth *c*, as:

$$\hat{\gamma}_c = \frac{\sum_x D_{x,c+x}}{\sum_x E_{x,c+x} \mu_{x,c+x}^0},$$

the maximum likelihood estimate of  $\gamma_c$  subject to the constraint that the kappa parameters are unchanged. This is the ratio of the actual number of deaths to the expected number of deaths under the GM(1,3) model for year of birth *c*, the 'A/E'.

Figures 3.7(i), (ii) and (iii) show the values of  $\hat{\gamma}_c$  for each year of birth *c*, for CMI data, England and Wales male data and England and Wales female data respectively. The following observations can be made from these figures:

- All three figures show a large increase in  $\hat{\gamma}_{1920}$  compared with  $\hat{\gamma}_{1919}$ . This feature was commented on in Section 2.8. It is slightly smaller than for CMI data than for the England and Wales data sets but still not insignificant.
- All three figures show a fall in the gamma parameters which can be identified with the 1925-45 favourable cohort effect. The fall is larger for England and Wales females than for England and Wales males, which ties in with the observation in Section 2.8 that more of the cohort structure was captured implicitly via the kappa parameters for England and Wales males than for England and Wales females. For CMI data, the fall is smaller than for England and Wales data, consistently with the observation in Section 2.8 that most of the cohort structure is captured implicitly via the kappa parameters.
- Figure 3.7(iii) shows a rise in the gamma parameters centred on 1915, which can be identified with the adverse cohort effect for females born around this time that was mentioned in Sections 2.3 and 2.8. There is evidence of this rise continuing into the early 1920s, on the far side of the 1919-20 discontinuity.
- The gamma parameters for CMI data in particular become volatile for the more recent years of birth and fall to exceptionally low values for the last two years of birth

in the data (1970 and 1971). These values are based on particularly small numbers of deaths and so should not be incorporated uncritically into future projections.

- Further similarities can be seen between the graphs for the two male data sets in addition to those relating to the 1919-20 discontinuity and to the 1925-45 favourable cohort effect. For example, both graphs show a rise in the gamma parameters for years of birth in the 1890s, followed by a fall from 1900 to around 1915.
- The most significant differences between the graphs for England and Wales males and females occur at the earliest years of birth, where the gamma parameters for males are relatively high and those for females are relatively low. This may be a consequence of data quality issues in this region of the data sets.

## 3.4 AIC and BIC

Table 3.1 shows the impact on both the AIC and BIC of introducing gamma parameters, for all three restricted data sets.

|            | With or without |              |            |            |            |
|------------|-----------------|--------------|------------|------------|------------|
|            | gamma           | Maximum log- | Number of  |            |            |
| Data set   | parameters?     | likelihood   | parameters | AIC        | BIC        |
| CMI        | Without         | -16,584.48   | 236        | -16,820.48 | -17,550.06 |
| CMI        | With            | -16,185.71   | 347        | -16,532.71 | -17,605.43 |
| E&W Male   | Without         | -18,277.96   | 176        | -18,453.96 | -18,970.60 |
| E&W Male   | With            | -15,016.88   | 271        | -15,287.88 | -16,083.39 |
| E&W Female | Without         | -18,669.46   | 176        | -18,845.46 | -19,362.10 |
| E&W Female | With            | -15,824.09   | 271        | -16,095.09 | -16,890.60 |

Table 3.1 – Impact on the AIC and BIC of introducing gamma parameters

Note that the number of data cells, *N*, used in the calculation of the BIC values in Table 3.1 has been reduced from 3,599 to 3,579 for CMI data and from 2,640 to 2,620 for the two England and Wales data sets, as a result of the exclusion of years of birth observed for less than five years in the data.

From Table 3.1, we see that the introduction of a cohort effect into the model has improved both the BIC and the AIC very significantly for England and Wales data for both males and females. For CMI data, however, the improvement in the AIC is much smaller, and there is actually a deterioration in the BIC. As discussed in Sections 2.8 and 3.1, it appears that most of the cohort effects are captured implicitly via the kappa parameters for CMI data, but that this is less true for England and Wales data. In addition, the smaller size of the CMI data set means that the remaining cohort effects will be less statistically significant. Nevertheless, the similarities between the graphs in Figure 3.7 for CMI data and for England and Wales male data suggest that we are justified in including the cohort effect in the model for CMI data.

It should be noted that it would be posssible to improve further on the AIC and BIC figures shown in Table 3.1 for the models with gamma parameters by estimating both the kappa and gamma parameters in a single iterative procedure. This was investigated for England and Wales male data under the two methods of carrying out the single iterative procedure described in Appendix B. It was found that under both methods, the AIC and BIC increased by approximately 600. The significant size of the improvements is consistent with the observations in Chapter 2 regarding cohort effects being reflected implicitly in the shapes of the kappa parameter graphs, as these observations suggest that we should be able to improve the fit significantly by making these cohort effects explicit. Unfortunately, however, the shapes of the parameter graphs given by the single iterative procedures, as shown in Figures B.1 and B.2, are unsuitable for forecasting.

# 3.5 Standardised residuals

Table 3.2 shows how the sample variances of the standardised residuals are affected by the introduction of gamma parameters.

 Table 3.2 – Impact on the sample variances of the standardised residuals of introducing gamma parameters

|            | Sample variance of       | Sample variance of          |  |
|------------|--------------------------|-----------------------------|--|
|            | standardised residuals   | standardised residuals with |  |
| Data set   | without gamma parameters | gamma parameters            |  |
| CMI        | 2.0878                   | 1.9106                      |  |
| E&W Male   | 4.5321                   | 1.9531                      |  |
| E&W Female | 5.1339                   | 2.9200                      |  |

The values in Table 3.2 for the model without gamma parameters are slightly different from the values in Tables 2.6-2.8 as a result of the exclusion of years of birth observed for less than five years in the data.

Table 3.2 shows that for the England and Wales data sets, a very significant part of the variation in the standardised residuals of the GM(1,3) model is explained by cohort effects. Consistently with the conclusions we drew from the AIC and BIC values in Table 3.1, the part of the variation explained by cohort effects for CMI data is smaller. The amount of variation in the standardised residuals that remains unexplained after introducing gamma parameters is greater for the England and Wales female data set than for the male data sets.

### 3.6 Residual plots

Figure 3.8 shows the graphs corresponding to those in Figure 3.4 after the introduction of gamma parameters. As expected from our previous observations, the changes are more significant for the England and Wales data sets than for the CMI data set. For England and Wales data, the significant diagonal patterns in Figures 3.4(ii) and (iii) have largely been eliminated in Figures 3.8(ii) and (iii), and for CMI data, the more limited diagonal patterns in Figure 3.4(i) have largely been eliminated in Figure 3.8(i). The introduction of gamma parameters does not help to eliminate the systematic understatement of mortality at ages 60-65 and the systematic overstatement at ages 65-70 in CMI data – these features can still be seen in Figure 3.8(i).

Figures 3.9 and 3.10 respectively show the standardised residuals plotted against age and year of birth for the GM(1,3) model after the introduction of gamma parameters. The corresponding figures before the introduction of gamma parameters are Figures 3.5 and 3.6 respectively. Points will have moved significantly in Figures 3.9 and 3.10 compared with Figures 3.5 and 3.6 for years of birth where the estimated gamma parameters are significantly different from 1, particularly if the expected number of deaths is sufficiently large to make the movement credible.

The graphs in Figure 3.9 have the same general shape as those in Figure 3.5. The main difference is that most of the extreme standardised residuals for England and Wales data that can be seen in Figure 3.5 have been eliminated in Figure 3.9. It would appear that cohort effects explain most of the extreme standardised residuals that existed before introducing gamma parameters. In fact, most of these extreme standardised residuals arose from the 1919-20 discontinuity, which is now reflected in the higher fitted value of  $\gamma_{1920}$  than of  $\gamma_{1919}$ .

It is clear that Figure 3.10 shows far fewer systematic patterns than Figure 3.6, particularly in England and Wales data where cohort effects not already captured implicitly via the kappa parameters are more significant than for CMI data. In particular, the jumps between years of birth 1919 and 1920 that existed in Figure 3.6 are no longer present in Figure 3.10, because this feature of the data is now reflected in the higher fitted value of  $\gamma_{1920}$  than of  $\gamma_{1919}$ .

# 3.7 Conclusion

An explicit cohort effect has been added to the GM(1,3) model fitted in Chapter 2. As a result of the need to ensure that the cohort parameters represented genuine cohort effects, a simple approach was selected, in which a multiplicative 'A/E' factor depending on year of birth was applied to the force of mortality under the GM(1,3) model without changing the parameter estimates of the latter. Introducing the cohort effect was found to improve the fit significantly for the England and Wales data sets but not for the CMI data set. However, the view was taken that it was appropriate to include the cohort effect in the model for CMI data as well as for England and Wales data, given the similarities between the fitted cohort parameter values for the different data sets.
**Figure 3.1** – Maximum likelihood parameter estimates for the GM(1,3) model fitted to the restricted set of CMI data, before the introduction of gamma parameters –

$$\mu_{xt} = \kappa_t^{(0)} + \exp[\kappa_t^{(3)} + \kappa_t^{(4)}(x - x) + \kappa_t^{(5)}((x - x)^2 - \hat{\sigma}_x^2)] - (i) \ \kappa_t^{(0)}, \ (ii) \ \kappa_t^{(3)}, \ (iii) \ \kappa_t^{(4)}$$
(iv)  $\kappa_t^{(5)}$ 



**Figure 3.2** – Maximum likelihood parameter estimates for the GM(1,3) model fitted to the restricted set of England and Wales male data, before the introduction of gamma parameters –  $\mu_{xt} = \kappa_t^{(0)} + \exp[\kappa_t^{(3)} + \kappa_t^{(4)}(x - \overline{x}) + \kappa_t^{(5)}((x - \overline{x})^2 - \hat{\sigma}_x^2)]$  – (i)  $\kappa_t^{(0)}$ , (ii)  $\kappa_t^{(3)}$ , (iii)  $\kappa_t^{(4)}$ , (iv)  $\kappa_t^{(5)}$ 



-6e-04

-8e-04

1970

1980

Year, t

2000

1990

0.105

1970

1980

Year, t

2000

1990

**Figure 3.3** – Maximum likelihood parameter estimates for the GM(1,3) model fitted to the restricted set of England and Wales female data, before the introduction of gamma parameters –  $\mu_{xt} = \kappa_t^{(0)} + \exp[\kappa_t^{(3)} + \kappa_t^{(4)}(x-\overline{x}) + \kappa_t^{(5)}((x-\overline{x})^2 - \hat{\sigma}_x^2)]$  – (i)  $\kappa_t^{(0)}$ , (ii)  $\kappa_t^{(3)}$ , (iii)  $\kappa_t^{(4)}$ , (iv)  $\kappa_t^{(5)}$ 



75





(iii)





Age

**Figure 3.5** – Scatter diagrams of standardised residuals plotted against age for the GM(1,3) model fitted to the restricted data sets, before the introduction of gamma parameters – (i) CMI data, (ii) England and Wales male data, (iii) England and Wales female data









**Figure 3.8 –** Plots of standardised residuals for the GM(1,3) model extended to incorporate a cohort effect – (i) CMI data, (ii) England and Wales male data, (iii) England and Wales female data



(iii)





**Figure 3.9 –** Scatter diagrams of standardised residuals plotted against age for the GM(1,3) model extended to incorporate a cohort effect – (i) CMI data, (ii) England and Wales male data, (iii) England and Wales female data

**Figure 3.10 –** Scatter diagrams of standardised residuals plotted against year of birth for the GM(1,3) model extended to incorporate a cohort effect – (i) CMI data, (ii) England and Wales male data, (iii) England and Wales female data



# 4: Fitting time series models to the parameter estimates

### 4.1 Introduction

In this chapter, time series models are fitted to the estimated parameters of the GM(1,3) model, after introducing gamma parameters, for all three data sets.

We needed to consider cohort effects to explain the shapes of some of the kappa parameter graphs in Chapter 2. None of the time series models we shall consider is capable of incorporating these cohort effects explicitly. Therefore, we shall need to give careful consideration to the impact of cohort effects on each of the kappa parameter graphs before fitting a time series model.

It is to be expected that the parameter estimates of the time series models we fit will be different for the three data sets. If there is clear justification from the data, then it may also be appropriate to fit different *models* to different data sets. However, this should not be done without careful consideration as it might lead to inconsistencies between the future mortality projections for the different data sets, *e.g.* mortality for assured lives might become higher than for the general population in the later years of the projection.

Ideally we would fit a multivariate time series model to all the kappa parameter graphs together. This was investigated but it was found that the cohort effects implicit in the kappa parameter graphs distorted the estimation of the parameters of the multivariate model. Therefore, we shall instead fit univariate time series models to each of the kappa parameter graphs individually, calculate the correlations between the resulting residuals and (in Chapter 5) use these as the correlations between the innovations in future projections. We shall also fit a separate time series model to the gamma parameters.

Appendix C contains the results from the theory of time series analysis that we shall use. However, we note here that a *first-order autoregressive process*, or AR(1) process,  $(X_t)$  is defined by:

$$X_t = \mu + \alpha (X_{t-1} - \mu) + e_t,$$

where the  $e_t$  are independent N(0, $\sigma^2$ ) random variables and  $\alpha$ ,  $\mu$  and  $\sigma^2$  are parameters to be estimated, with  $|\alpha| < 1$ . Under such a process, the values of X tend to revert to the mean level  $\mu$ , with the reversion being more rapid for small absolute values of  $\alpha$ . If  $\alpha > 0$ , then a value of X above  $\mu$  tends to be followed by another value of X above  $\mu$  and vice versa. If  $\alpha < 0$ , then a value of X above  $\mu$  tends to be followed by a value of X below  $\mu$  and vice versa.

A first-order moving average process, or MA(1) process,  $(X_t)$  is defined by:

$$X_t = \mu + e_t + \beta e_{t-1},$$

where the  $e_t$  are independent N(0, $\sigma^2$ ) random variables and  $\beta$ ,  $\mu$  and  $\sigma^2$  are parameters to be estimated. If  $\beta > 0$ , then a value of  $X_t$  above  $\mu$  suggests that  $X_{t+1}$  will also be above  $\mu$ , but subsequent values of X are equally likely to be above or below  $\mu$ . If  $\beta < 0$ , then a value of  $X_t$  above  $\mu$  suggests that  $X_{t+1}$  will be below  $\mu$ , but subsequent values of X are equally likely to be above or below  $\mu$ .

# 4.2 κ<sup>(0)</sup> parameters

Figures 3.1(i), 3.2(i) and 3.3(i) show no clear signs of a trend over time in the  $\kappa^{(0)}$  parameters, with the shapes of Figures 3.2(i) and 3.3(i) being driven mostly by cohort effects. In addition, the  $\kappa^{(0)}$  parameters may be thought of as representing mortality from unnatural causes, and it may be reasonable to assume that rates of unnatural death remain broadly constant over time. Thus it is unlikely to be appropriate to fit a model incorporating a trend over time to the  $\kappa^{(0)}$  parameters.

Figure 4.1 shows the sample autocorrelation functions, as defined in Section C.1, of the  $\kappa^{(0)}$  parameters for the three data sets. All three graphs in Figure 4.1 start with a significant positive autocorrelation at lag 1, which then decays but remains positive for at least the next six lags. As explained in Section C.1, this is a characteristic feature of an AR(1) process. It appears that the  $\kappa^{(0)}$  parameters are reverting to some mean level,  $\mu$ .

Table 4.1 shows the estimates of the parameters  $\alpha$ ,  $\mu$  and  $\sigma^2$  when AR(1) processes are fitted to each of the  $\kappa^{(0)}$  series. Figure 4.2 plots the resulting residuals, and Figure 4.3 plots their sample autocorrelation functions.

| Data set   | α      | μ                         | σ <sup>2</sup>           |
|------------|--------|---------------------------|--------------------------|
| CMI        | 0.7388 | 3.624 x 10 <sup>-4</sup>  | 5.734 x 10 <sup>-9</sup> |
| E&W Male   | 0.9374 | 4.693 x 10 <sup>-4</sup>  | 2.945 x 10 <sup>-9</sup> |
| E&W Female | 0.9439 | -1.738 x 10 <sup>-₄</sup> | 6.798 x 10 <sup>-9</sup> |

| <b>Table 4.1 –</b> Parameter estimates of the AR(1) pro | processes fitted to the $\kappa^{(0)}$ se | eries |
|---|---|-------|
|---|---|-------|

On inspection of Figure 4.2(i), for CMI data, it seems reasonable to model the residuals as independent N(0, $\sigma^2$ ) random variables, where  $\sigma^2$  has been estimated as 5.734 x 10<sup>-9</sup>, and thus the AR(1) model seems reasonable. Figures 4.2(ii) and (iii), for England and Wales data, do show certain patterns in the residuals but they are in line with what we would expect given the known cohort effects.

As per Section C.2, we should regard the autocorrelations shown in Figure 4.3 as statistically significant if their absolute value exceeds  $2 / N^{1/2}$ , where *N* is the number of years of data, so that N = 44 for the England and Wales data sets and 59 for the CMI data set. Thus autocorrelations exceeding 0.260 in absolute value for CMI data, and exceeding 0.302 in absolute value for England and Wales data, are statistically significant. These critical values are shown as blue dashed lines in Figure 4.3, as they are in all the autocorrelation graphs we shall show. Figure 4.3 shows a few autocorrelations outside the blue dashed lines for each data set but the lags where this occurs are different for each data set and Figure 4.3 does not suggest any key feature of the data which we have failed to model. Accordingly, we shall use the AR(1) model rather than introducing more complex models.

# 4.3 κ<sup>(3)</sup> parameters

Figures 3.1(ii), 3.2(ii) and 3.3(ii) all show a clear downward trend over time in the  $\kappa^{(3)}$  parameters. We should therefore fit a time series model to the differences between the  $\kappa^{(3)}$  parameters for successive years, rather than to the  $\kappa^{(3)}$  parameters themselves. This will represent the general improvement in mortality over time. Figure 4.4 shows, for each data set, the difference between the  $\kappa^{(3)}$  parameter for each year and the parameter for the previous year. Figure 4.5 shows the sample autocorrelation functions of these differences.

Figure 4.5(i), for CMI data, and Figure 4.5(ii), for England and Wales male data, show significant negative autocorrelation at lag 1. The negative autocorrelation does not persist at higher lags. As explained in Section C.1, this is a characteristic feature of an MA(1) process. In Figure 4.5(iii), for England and Wales female data, the negative autocorrelation at lag 1 is not quite statistically significant, but it is reasonable on grounds of consistency with the other data sets to introduce a moving average parameter for England and Wales female data also.

Table 4.2 shows the estimates of the parameters  $\beta$ ,  $\mu$  and  $\sigma^2$  when MA(1) processes are fitted to each of the series of  $\kappa^{(3)}$  differences. Figure 4.6 plots the resulting residuals, and Figure 4.7 plots their sample autocorrelation functions.

| Fable 4.2 – Paramete | r estimates | of the MA | (1) proces | ses fitted t | o the $\kappa^{^{(3)}}$ | differences |
|----------------------|-------------|-----------|------------|--------------|-------------------------|-------------|
|----------------------|-------------|-----------|------------|--------------|-------------------------|-------------|

| Data set   | β       | μ        | σ <sup>2</sup> |
|------------|---------|----------|----------------|
| CMI        | -0.4041 | -0.02066 | 0.001697       |
| E&W Male   | -0.2257 | -0.01609 | 0.0007734      |
| E&W Female | -0.1067 | -0.01050 | 0.0008256      |

The model fitted here, *i.e.* an MA(1) model fitted to the differences, is generally referred to as an ARIMA(0,1,1) model in the literature on time series analysis. ARIMA stands for 'autoregressive integrated moving average'. The 0 in (0,1,1) indicates that no autoregressive parameters have been introduced, the first 1 indicates that the time series has been differenced once and the second 1 indicates that one moving average parameter has been introduced.

Figure 4.6 shows some evidence of the residuals being systematically positive until 1980, negative from 1980 to the early 1990s and positive thereafter, particularly for England and Wales data. This is a consequence of the 1925-45 cohort effect, as lives born in 1925-45 have experienced particularly high mortality improvements compared with the previous generation and the mortality of these lives will be driven mainly by the  $\kappa^{(3)}$  parameters from 1980 to the early 1990s. However, after considering Figures 3.2(ii) and 3.3(ii), it was concluded that is still reasonable to fit a time series model to the  $\kappa^{(3)}$  differences over the full range of years in the data. This is because these figures show a flattening of the  $\kappa^{(3)}$  parameters after the early 1990s, as the 1925-45 cohort ages and their mortality is driven more by the  $\kappa^{(4)}$  and  $\kappa^{(5)}$  parameters. This effectively cancels out the steepening of the fall in the  $\kappa^{(3)}$  parameters from 1980 to the early 1990s.

Figure 4.7(i) suggests that the ARIMA(0,1,1) model fits the CMI data well, with no statistically significant autocorrelations of the residuals. For England and Wales data, Figures 4.7(ii) and (particularly) (iii) reveal one potential source of concern, namely the significant positive autocorrelations at lag 2. However, cohort effects appear to have caused the autocorrelations at the shorter lags in these graphs to be systematically positive. If we fitted a model that placed cohort effects on an equal footing with period effects, it seems likely that the autocorrelations at lag 2 would no longer be statistically significant. On balance, therefore, introducing additional complexity into the model for the England and Wales data sets to capture this feature would not appear to be warranted. Thus we shall use the ARIMA(0,1,1) model.

It is of interest to consider what property of the underlying mortality rates the moving average parameter is reflecting. The negative moving average parameter is implying that high mortality in one year has a tendency to be followed by low mortality in the following year, and vice versa. In fact, if mortality is high in a particular year, then it is likely that some external factor driving mortality was adverse in that year, *e.g.* the year may have had a particularly harsh winter. In the following year, not only is it unlikely that this external factor will be equally adverse, but even if it is, mortality is unlikely to be as high because many of the lives who would otherwise have died in the second year died in the first year instead.

# 4.4 κ<sup>(4)</sup> parameters

We shall consider the most appropriate time series model for the  $\kappa^{(4)}$  parameters for England and Wales male data first, then for England and Wales female data and finally for CMI data.

Before 1980, the graph of  $\kappa^{(4)}$  parameters for England and Wales male data, Figure 3.2(iii), remains relatively level around 0.105. After 1980, the graph is dominated by the peak in the early 1990s which is a consequence of the 1925-45 cohort effect. By the final years of the data set, the graph has returned to levels around 0.105. Due to the singular nature of the cohort effect, this suggests that it is reasonable to assume a long-term mean of 0.105 for the  $\kappa^{(4)}$  parameters.

If the  $\kappa^{(4)}$  parameters move away from 0.105 as a result of future cohort effects, then it is likely that they will follow a similar pattern to that of Figure 3.2(iii), moving away over a period of 10-15 years before returning to a level around 0.105 over the following 10-15 years. If we fit an AR(1) process with  $\mu = 0.105$  to the  $\kappa^{(4)}$  parameters, then we will not be able to replicate this behaviour exactly as the time taken to return to a level around 0.105 will be random. However, for the purpose of quantifying the extent of uncertainty in future mortality rates, such a process is likely to be appropriate. If we fit an AR(1) process to the  $\kappa^{(4)}$  parameters, forcing  $\mu$  to equal 0.105, then the estimated parameter  $\alpha$  is 0.9678. It is to be expected that the estimated value of  $\alpha$  is close to 1, as the cohort effect caused the  $\kappa^{(4)}$  parameters to be away from 0.105 over an extended period, rather than to move away from 0.105 briefly and then return.

For England and Wales female data, it seems likely that the long-term mean of the  $\kappa^{(4)}$  parameters is approximately 0.09. This conclusion has been based mainly on the years since the early 1990s, once the lives born around 1915 who experience an adverse cohort effect have reached an age where they no longer have a significant impact on the  $\kappa^{(4)}$  parameters. We see that, since the early 1990s, Figure 3.3(iii) has closely resembled a shifted-down version of Figure 3.2(iii). This suggests that, consistently with the treatment of England and Wales male data, we should use the lower end of the downward trend from the early 1990s to 2005 as our estimate of the long-term mean of the  $\kappa^{(4)}$  parameters.

If we fit an AR(1) process to the estimated  $\kappa^{(4)}$  parameters, forcing  $\mu$  to equal 0.09, then the estimated parameter  $\alpha$  is 0.9712. Again it is to be expected that the estimated value of  $\alpha$  is close to 1.

For CMI data, a first glance at Figure 3.1(iii) suggests that we should be fitting a time series model incorporating an upward trend to the  $\kappa^{(4)}$  parameters. However, it does not seem likely that the  $\kappa^{(4)}$  parameters will continue to increase indefinitely for assured lives but not for the general population, as this would eventually cause assured lives mortality to exceed population mortality at older ages. Therefore, instead of fitting a time series model incorporating an upward trend, we shall seek to estimate a long-term mean  $\mu$  of the  $\kappa^{(4)}$  parameters and fit an AR(1) process with this value of  $\mu$ , in a similar way to that for England and Wales data.

Although cohort effects are not as apparent in Figure 3.1(iii) as in Figures 3.2(iii) and 3.3(iii), it is still likely that the apparent trend in Figure 3.1(iii) is in fact mainly a consequence of cohort effects. We indicated in Section 2.8 that the 1925-45 cohort effect is likely to be the explanation for the rise in  $\kappa^{(4)}$  parameters from 1980 to the early 1990s, and the overall trend of a rise in the  $\kappa^{(4)}$  parameters is in fact heavily concentrated in this period. The more interesting question is why, in CMI data, the  $\kappa^{(4)}$  parameters have shown no sign of falling since the early 1990s as in England and Wales data, but the  $\kappa^{(3)}$  parameters have instead continued to fall at broadly the same rate as during the period when they were the main driver of the mortality of the 1925-45 cohort.

As a difference has arisen between the behaviour of the parameter graphs for CMI data and for England and Wales data, there must have been a change in the characteristics of the assured lives population relative to the general population. The key change which suggests itself in this case is that UK life insurance companies began to write large volumes of mortgage-related endowment business when tax relief on premiums under qualifying life insurance contracts was introduced in 1979. Although this tax relief was abolished for new business from 1984, insurers continued to write mortgage-related endowment business in significant quantities until the early 1990s. The CMI does not publish assured lives mortality data separately for mortgage-related and non-mortgage-related business, but what we observe regarding the  $\kappa^{(3)}$  and  $\kappa^{(4)}$  parameters is consistent with mortgage-related business exhibiting lower mortality than non-mortgage-related business. This is because, as the mortgage endowment policyholders reach ages around 60, the  $\kappa^{(3)}$  parameters can then be expected to fall. The  $\kappa^{(4)}$  parameters will then tend to rise to maintain broadly the same level of mortality at older ages, by which the mortgage endowments have mostly matured. We also observe that the  $\kappa^{(0)}$  parameters did not increase as much in the 1980s for CMI data as for England and Wales data, and this is consistent with mortgage endowment policyholders of working age experiencing low mortality.

If the above is the correct explanation of why the  $\kappa^{(3)}$  parameters have continued to fall steeply, and the  $\kappa^{(4)}$  parameters have not fallen, since the early 1990s, then the phenomenon is likely to be temporary as most of the mortgage endowment business is due to mature within the next 5-10 years. It is therefore likely that the falls in the  $\kappa^{(3)}$  parameters will slow down to a rate more in line with historical rates. The  $\kappa^{(4)}$  parameters will then tend to decrease, so that

mortality in the early years of retirement continues to improve at a similar rate to that in the past. Thus we conclude that the  $\kappa^{(4)}$  parameters are likely to fall back to the levels of the 1960s and 1970s. We shall fit an AR(1) process to the estimated  $\kappa^{(4)}$  parameters forcing  $\mu$  to equal 0.1125, which is approximately equal to the mean value of the  $\kappa^{(4)}$  parameters over the 1960s and 1970s.

We cannot estimate the parameter  $\alpha$  of this AR(1) process by reference to the whole period 1947-2005, as the estimation procedure will fail to take account of the fact that the  $\kappa^{(4)}$  parameters can be expected to fall back to a level around 0.1125 after 2005. Neither is it likely to be appropriate to estimate  $\alpha$  by reference only to the period 1947-1980, as this would not consider the possibility of the  $\kappa^{(4)}$  parameters being away from 0.1125 for an extended period of time because of cohort effects. Instead we shall use the same value of  $\alpha$  as for England and Wales males, 0.9678.

Table 4.3 summarises the parameters of the AR(1) processes we have selected. It also shows the estimates of the error variances  $\sigma^2$ .

| Data set   | α      | μ      | σ <sup>2</sup>           |
|------------|--------|--------|--------------------------|
| CMI        | 0.9678 | 0.1125 | 7.476 x 10⁻ <sup>6</sup> |
| E&W Male   | 0.9678 | 0.105  | 1.446 x 10⁻ <sup>6</sup> |
| E&W Female | 0.9712 | 0.09   | 3.102 x 10 <sup>-6</sup> |

**Table 4.3** – Parameter estimates of the AR(1) processes fitted to the  $\kappa^{(4)}$  series

Figure 4.8 shows the residuals resulting from fitting these AR(1) processes. The residuals for CMI data show no clear pattern. The shapes of the graphs for England and Wales data are generally in line with what we would expect given the known cohort effects, with the residuals tending to be positive over periods where cohort effects cause the  $\kappa^{(4)}$  parameters to rise, and negative over periods where cohort effects cause the  $\kappa^{(4)}$  parameters to fall. It seems unlikely that we would be able to eliminate the patterns remaining in these graphs without using a model placing cohort effects on an equal footing with period effects.

Figure 4.9 shows the sample autocorrelation functions of the residuals in Figure 4.8.

The CMI data show a significant negative autocorrelation at lag 1, which does not persist at subsequent lags, possibly suggesting a need to introduce a moving average parameter. The England and Wales data sets do not show similar effects. Introducing a (negative) moving average parameter for the CMI data only would improve the fit somewhat, but it would make the model for CMI data inconsistent with those for England and Wales data. It is also unlikely to have a significant impact on future projections as, rather than implying a different overall trend in mortality over time, it simply means that mortality increasing rapidly with age in one year tends to be followed by mortality increasing more slowly with age in the following year and vice versa. We shall not introduce such a moving average parameter here.

The England and Wales male data show significant positive autocorrelations for the first few lags, but this is more likely to be related to the systematic patterns in the residuals arising from the cohort effect rather than suggesting a need to introduce a further autoregressive parameter.

# 4.5 $\kappa^{(5)}$ parameters

The graphs of  $\kappa^{(5)}$  parameters in Figures 3.1(iv), 3.2(iv) and 3.3(iv) show a rapid increase in the  $\kappa^{(5)}$  parameters since the early 1990s, particularly for England and Wales data, as a consequence of the 1925-45 cohort effect. There is strong reason to believe that this increase will reverse out over the next 10-15 years, as lives in the 1925-45 cohort approach the upper age limit of the data sets. It is therefore necessary to fit a model under which the  $\kappa^{(5)}$  parameters revert to a long-term mean, whose estimation should not be biased by cohort effects. This mean reversion should occur over a period of 10-15 years rather than a single year, suggesting the use of an AR(1) rather than an MA(1) model.

Inspection of Figures 3.1(iv) and 3.2(iv) suggests  $-6 \times 10^{-4}$  as an appropriate long-term mean for the  $\kappa^{(5)}$  parameters, based on the period up to the early 1990s. We shall therefore fit AR(1) models to the  $\kappa^{(5)}$  parameters for both CMI data and England and Wales male data, forcing the long-term mean  $\mu$  to equal  $-6 \times 10^{-4}$ .

We cannot estimate the autoregressive parameter  $\alpha$  directly by reference to the values in Figures 3.1(iv) and 3.2(iv), as this will not allow for the fact that the  $\kappa^{(5)}$  parameters can be expected to fall back to a level around  $-6 \times 10^{-4}$  after 2005. However, a cohort effect would be expected to cause the  $\kappa^{(5)}$  parameters to deviate from their long-term mean over a similar number of years to that for the  $\kappa^{(4)}$  parameters. We shall therefore use the same value of  $\alpha$  as for the  $\kappa^{(4)}$  parameters,  $\alpha = 0.9678$ .

For England and Wales female data, it seems likely that the long-term mean of the  $\kappa^{(5)}$  parameters is approximately  $1 \times 10^{-4}$ , which is approximately equal to the minimum value the  $\kappa^{(5)}$  parameters reach in the early 1990s. It is not immediately obvious that the combined impact on the  $\kappa^{(5)}$  parameters of the adverse cohort effect centred on 1915 and the favourable cohort effect centred on 1931 will be broadly neutral over this period. However, the fall in  $\kappa^{(4)}$  parameters since the early 1990s, which should not be significantly affected by the adverse cohort effect centred on 1915, is approximately 90% of that for the male data, and this suggests that, in the absence of the adverse cohort effect, the rise in  $\kappa^{(5)}$  parameters since the early 1990s due to the cohort effect centred on 1931 would also be approximately 90% of that for male data, which is in fact close to the rise we observe.

Again the parameter  $\alpha$  cannot be estimated directly from the values in Figure 3.3(iv), so we shall use the same value as for the  $\kappa^{(4)}$  parameters,  $\alpha = 0.9712$ .

Table 4.4 summarises the parameters of the AR(1) processes we have selected. It also shows the estimates of the error variances  $\sigma^2$ .

| Data set   | α      | μ                     | σ²                       |
|------------|--------|-----------------------|--------------------------|
| CMI        | 0.9678 | -6 x 10 <sup>-4</sup> | 1.291 x 10 <sup>-8</sup> |
| E&W Male   | 0.9678 | -6 x 10⁻⁴             | 3.742 x 10 <sup>-9</sup> |
| E&W Female | 0.9712 | 1 x 10 <sup>-4</sup>  | 2.852 x 10 <sup>-9</sup> |

Figure 4.10 shows the residuals arising from fitting these AR(1) processes. The residuals for CMI data show no clear pattern. The patterns that can be seen for England and Wales data can largely be explained by the known cohort effects, so it seems unlikely that we would be able to eliminate these patterns without using a model placing cohort effects on an equal footing with period effects.

Figure 4.11 contains graphs of the sample autocorrelation functions of the residuals in Figure 4.10. For CMI data, there is a similar issue to that for the  $\kappa^{(4)}$  parameters in that there is a significant negative autocorrelation at lag 1, but the additional complexity of introducing a moving average parameter to accommodate this was again not considered to be justified. For the England and Wales male data, Figure 4.11(ii) does not suggest any justification for introducing additional parameters. Figure 4.11(iii), for England and Wales female data, shows a significant positive autocorrelation at lag 2, but similarly to Figure 4.7(iii), it is doubtful whether this autocorrelation would still be significant if a model were fitted that placed cohort effects on an equal footing with period effects, and so we shall not introduce additional complexity by attempting to model this feature.

#### 4.6 y parameters (cohort parameters)

The procedure we used to fit the model involved setting all the gamma parameters to 1 initially, estimating the kappa parameters, and then estimating the gamma parameters as A/Es without changing the kappa parameter estimates. This procedure places a limit on the period for which the gamma parameters can be away from 1. For example, if there were any calendar year such that the gamma parameters were less than 1 for all years of birth that

were within the age range of the data in that year, then the kappa parameters for that year could not maximise the log-likelihood of the model without gamma parameters – it must be possible to increase the log-likelihood by increasing both the  $\kappa^{(0)}$  and  $\kappa^{(3)}$  parameters for that year in such a way that all the fitted forces of mortality for that year increase by a multiplicative constant. Therefore, we should represent the gamma parameters by a time series model with the property of mean reversion to 1.

We commented in Section 3.3 that the estimates of the gamma parameters for the last few years of birth in the restricted set of CMI data are volatile, and are likely to be unreliable because they are based on small numbers of deaths. For example, if the mean number of deaths in a particular year of birth is 1,000, then the variance is also 1,000 because the variance of a Poisson distribution is equal to its mean. Thus the standard deviation is  $(1,000)^{1/2} = 31.6$ , or 3.16% of the mean. This corresponds to a difference of 0.0316 in the estimated gamma parameter value, so random fluctuations potentially distort Figure 3.7 significantly in years of birth where there are fewer than 1,000 deaths. We shall therefore regard estimates of the gamma parameters as unreliable if they are based on fewer than 1,000 deaths.

In the CMI data, there are fewer than 1,000 deaths for each year of birth from 1956 onwards. (There are also fewer than 1,000 deaths for the year of birth 1861 but this should not have material implications for the reliability of future projections.) In the England and Wales data for both males and females, there are over 1,000 deaths for each year of birth, with the one exception of females born in 1971, where the number of deaths is 958 which is only marginally less than 1,000. Accordingly, we shall fit a time series model to the gamma parameters for CMI data based on years of birth 1861-1955 only. We shall take the gamma parameters for subsequent years of birth from projections, ignoring the values calculated from the data. We shall fit time series models to the gamma parameters for England and Wales data based on the full range of years of birth in the restricted data sets, 1877-1971.

We should like to ensure that there is no possibility, however remote, of the gamma parameters becoming negative in future projections, so that the forces of mortality do not become negative in future projections. We shall therefore fit a time series model to the logarithms of the gamma parameters rather than to the gamma parameters themselves. These logarithms are illustrated in Figure 4.12. We then need a time series model with the property of mean reversion to 0 rather than to 1. As we have observed cohort effects that move the gamma parameters away from 1 over a period of several years rather than for just a single year, we shall use an AR(1) model rather than an MA(1) model.

We should consider the treatment of the 1919-20 discontinuity relating to the influenza epidemic. We have chosen to retain it in the data on the grounds that similar epidemics may occur in the future. An alternative approach, which we do not pursue here but which constitutes a possible area for future research, would be to exclude the discontinuity from the data used to fit the main part of the model but to extend the model to incorporate shocks either in a particular year or for lives born in a particular year.

Table 4.5 shows the estimates of the parameters  $\alpha$  and  $\sigma^2$  when AR(1) processes with  $\mu = 0$  are fitted to each of the logged  $\gamma$  series. Figure 4.13 plots the resulting residuals, and Figure 4.14 plots their sample autocorrelation functions.

| Data set   | α      | μ | σ <sup>2</sup>           |
|------------|--------|---|--------------------------|
| CMI        | 0.4182 | 0 | 8.008 x 10⁻⁴             |
| E&W Male   | 0.4378 | 0 | 5.115 x 10 <sup>-4</sup> |
| E&W Female | 0.4923 | 0 | 5.118 x 10 <sup>-4</sup> |

**Table 4.5** – Parameter estimates of the AR(1) processes fitted to the logged  $\gamma$  series

We are now fitting a time series model to 95 rather than 59 data items for CMI data, and to 95 rather than 44 data items for England and Wales data. The threshold absolute value above which we should regard autocorrelations in Figure 4.14 as statistically significant has therefore decreased to 2 /  $(95)^{1/2}$  = 0.205, for both CMI data and England and Wales data.

Figure 4.14 shows that this threshold is exceeded at lag 3 only for CMI data and for England and Wales female data, and is not exceeded at any lag for England and Wales male data. Overall, Figures 4.13 and 4.14 do not suggest any need to add further parameters to the AR(1) model.

## 4.7 Correlations

We expect there to be significant correlations between the residuals of the time series models we have fitted to the different kappa parameter graphs. Cohort effects are an important source of such correlations – for example, the 1925-45 cohort effect caused a decrease in the  $\kappa^{(3)}$  parameters together with an increase in the  $\kappa^{(4)}$  parameters from 1980 to the early 1990s. It is necessary to allow for these correlations in projecting future parameter values and thus projecting future mortality rates. Tables 4.6, 4.7 and 4.8 show the sample correlation coefficients between the residuals of the different kappa time series models, for CMI data, England and Wales male data and England and Wales female data respectively. As mentioned in Section 4.1, these correlations will be assumed for the innovations in the future projections in Chapter 5.

 Table 4.6 – Sample correlation coefficients between the residuals of the time series models fitted to the kappa parameter graphs for CMI data

|                         | $\kappa^{(0)}$ | κ <sup>(3)</sup> | κ <sup>(4)</sup> | <i>К</i> <sup>(5)</sup> |
|-------------------------|----------------|------------------|------------------|-------------------------|
| <i>κ</i> <sup>(0)</sup> | 1              |                  |                  |                         |
| K <sup>(3)</sup>        | -0.5187        | 1                |                  |                         |
| K <sup>(4)</sup>        | 0.6306         | -0.5852          | 1                |                         |
| K <sup>(5)</sup>        | -0.4404        | 0.5804           | -0.6771          | 1                       |

**Table 4.7** – Sample correlation coefficients between the residuals of the time series models fitted to the kappa parameter graphs for England and Wales male data

|                  | κ <sup>(0)</sup> | <i>к</i> <sup>(3)</sup> | К <sup>(4)</sup> | κ <sup>(5)</sup> |
|------------------|------------------|-------------------------|------------------|------------------|
| $\kappa^{(0)}$   | 1                |                         |                  |                  |
| κ <sup>(3)</sup> | -0.6744          | 1                       |                  |                  |
| κ <sup>(4)</sup> | 0.6798           | -0.3946                 | 1                |                  |
| κ <sup>(5)</sup> | -0.7421          | 0.5879                  | -0.6349          | 1                |

**Table 4.8** – Sample correlation coefficients between the residuals of the time series models fitted to the kappa parameter graphs for England and Wales female data

|                  | $\kappa^{(0)}$ | $\kappa^{(3)}$ | $\kappa^{(4)}$ | $\kappa^{(5)}$ |
|------------------|----------------|----------------|----------------|----------------|
| κ <sup>(0)</sup> | 1              |                |                |                |
| K <sup>(3)</sup> | -0.7427        | 1              |                |                |
| K <sup>(4)</sup> | 0.8835         | -0.5362        | 1              |                |
| κ <sup>(5)</sup> | -0.7999        | 0.7033         | -0.7204        | 1              |

We shall not consider the possibility of cross-correlation between the kappa residuals and the gamma residuals here. Ideally, all period effects should be captured in the kappa parameters and all cohort effects should be captured in the gamma parameters. In this situation, it becomes clear that it is appropriate to assume no cross-correlation.

### 4.8 Conclusion

Univariate time series models have been fitted to the kappa and gamma parameter graphs obtained from fitting the GM(1,3) model, after extension to incorporate a cohort effect, to all three data sets. The models were AR(1) models for the  $\kappa^{(0)}$ ,  $\kappa^{(4)}$  and  $\kappa^{(5)}$  parameters, an ARIMA(0,1,1) model, incorporating the downward trend, for the  $\kappa^{(3)}$  parameters and an AR(1) model fitted to the logarithms of the  $\gamma$  parameters. For the  $\kappa^{(4)}$ ,  $\kappa^{(5)}$ , and  $\gamma$  parameters, the form and parameters of the time series models selected depended fundamentally on a subjective consideration of how cohort effects influenced these parameter graphs, and it is likely that

very different models and/or parameter estimates would have been obtained by naïve application of standard fitting techniques. The possibility of fitting a multivariate time series model to all the kappa parameter graphs together was considered, but was not pursued because, in the multivariate case, it was not feasible to give subjective consideration to the impact of cohort effects on the parameter values and naïve application of standard fitting techniques was found not to produce reasonable results. Instead the correlations between the residuals of the time series models for the different kappa parameters were calculated, and these will be used as the correlations between the future innovations in Chapter 5.



**Figure 4.1** – Sample autocorrelation functions of the  $\kappa^{(0)}$  parameters – (i) CMI data, (ii) England and Wales male data, (iii) England and Wales female data



**Figure 4.2** – Residuals of the AR(1) processes fitted to the  $\kappa^{(0)}$  series – (i) CMI data, (ii) England and Wales male data, (iii) England and Wales female data









Year, t



Lag

**Figure 4.5** – Sample autocorrelation functions of the differences in Figure 4.4 – (i) CMI data, (ii) England and Wales male data, (iii) England and Wales female data











**Figure 4.8** – Residuals of the AR(1) processes fitted to the  $\kappa^{(4)}$  series – (i) CMI data, (ii) England and Wales male data, (iii) England and Wales female data







**Figure 4.10** – Residuals of the AR(1) processes fitted to the  $\kappa^{(5)}$  series – (i) CMI data, (ii) England and Wales male data, (iii) England and Wales female data



**Figure 4.11** – Sample autocorrelation functions of the residuals in Figure 4.10 – (i) CMI data, (ii) England and Wales male data, (iii) England and Wales female data



Year of birth, c

**Figure 4.12** – Logarithms of the *y* parameters – (i) CMI data, (ii) England and Wales male data, (iii) England and Wales female data







**Figure 4.14** – Sample autocorrelation functions of the residuals in Figure 4.13 – (i) CMI data, (ii) England and Wales male data, (iii) England and Wales female data

# 5: Forecasting and life insurance applications

## 5.1 Introduction

For each of the data sets, under the time series models fitted in Chapter 4, 500 sets of sample paths of the future parameter values were generated. The same 500 sets of future normal variates were used for all three data sets. The correlations assumed between the different kappa innovations were those in Tables 4.6-4.8.

Section 5.2 shows prediction intervals to illustrate the extent of uncertainty in the future parameter values. We do not base these prediction intervals on the simulation output, as closed formulae exist for them.

The relationship between the uncertainty in the future parameter values and the uncertainty in projected forces of mortality is relatively complex because of the significant correlations between the different kappa parameter series. We therefore show separate prediction intervals for projected forces of mortality in Section 5.3. These prediction intervals are based on the simulation output. Section 5.3 also compares these projected forces of mortality with a number of previously published projections.

In order to use these forces of mortality to calculate annuity values, it is necessary to extrapolate the mortality curves to ages above the highest age in the data, *i.e.* above 90 for CMI data and 89 for England and Wales data. We identify two methods that might be considered appropriate for extrapolating the mortality curve for 2005 (the latest year in the data). These methods are, however, unlikely to be appropriate for extrapolating projected mortality curves in future years, and so a separate method is required for determining future improvements at ages above the highest age in the data. Two possible methods are considered for this as well.

Section 5.5 uses the projected forces of mortality to calculate immediate and deferred annuity values for lives at different ages in 2005. These annuity values are compared with values from previously published projections.

Section 5.6 applies the projections to calculate risk-based capital requirements for the same immediate and deferred annuities as in Section 5.5, under the Individual Capital Assessment (ICA) regime of the UK insurance regulator, the Financial Services Authority (FSA). The capital requirements calculated are found to be significantly lower than those given by a 'rule of thumb' widely used within the UK life insurance industry, and the implications of this are discussed. Section 5.7 concludes.

### 5.2 Prediction intervals for the future parameter values

Figures 5.1-5.5 show 95% prediction intervals for the projected parameter values, based on the 2.5<sup>th</sup> and 97.5<sup>th</sup> percentiles of the distributions of those values. Both the kappa (period) and gamma (cohort) parameters are shown. This gives the range within which a particular parameter for a particular future year lies under 95% of scenarios. As stated in Section 5.1, these prediction intervals were calculated analytically rather than being based on the simulation output. Section C.3 details the formulae used.

The following observations can be made from Figures 5.1-5.5:

- In Figures 5.1, 5.4 and 5.5, the prediction intervals broadly correspond to the range of values that the  $\kappa^{(0)}$ ,  $\kappa^{(5)}$  and  $\gamma$  parameters (respectively) have exhibited in the past. This is because we have fitted a time series model not incorporating a trend to these parameters.
- In Figure 5.3(i), for CMI data, the prediction interval broadly corresponds to the range of values the  $\kappa^{(4)}$  parameter has taken in the past. In Figures 5.3(ii), for England and Wales male data, and 5.3(iii), for England and Wales female data, the prediction intervals are biased towards the lower end of the range of past values. For both the

England and Wales data sets, we argued in Section 4.4 that future values of this parameter were likely to be around the lower end of the range of past values, and fitted an AR(1) model with a value of  $\mu$  reflecting this.

- In Figure 5.2, relating to the  $\kappa^{(3)}$  parameters, for which we have fitted a time series model incorporating a downward trend, the prediction intervals show a continuation of this trend. Although the long-term average annual rate of this trend is fixed, being equal to the parameter  $\mu$  of the MA(1) process fitted to the  $\kappa^{(3)}$  parameters, there is significant uncertainty in the actual rate of the trend realised over the projection period.
- The prediction intervals are, in most cases, widest for CMI data and narrowest for England and Wales male data, consistently with the relative values of the estimated innovation variances  $\sigma^2$ . It is likely that the relatively small volume of data in the CMI data set has given rise to volatility in the parameter estimates for successive calendar years and years of birth, resulting in an increase in the estimated innovation variances. For England and Wales female data, the existence of two significant cohort effects, one favourable and one adverse, which are not being modelled explicitly leads to more variation in the parameter values being reflected in the innovation terms than for England and Wales male data.
- For the parameters other than the  $\kappa^{(3)}$  parameters, the future width of the prediction intervals converges relatively quickly over time to a limiting value. For the  $\kappa^{(3)}$ parameters, however, the prediction intervals continue to widen over time, and their width would in fact increase without limit if the projections were extended indefinitely. In other words, under the model we have fitted, the uncertainty about the overall level of mortality increases the further into the future projections are made, but the uncertainty about the shape of the mortality curve far into the future is not very much greater than in the near future.

It is important to note that the boundaries of the prediction intervals shown in Figures 5.1-5.5 are not sample paths. A typical sample path will have high parameter values in some future years and low parameter values in others, and in addition, as a result of the correlations between the different kappa innovations, a typical sample path will have high values of some kappa parameters and low values of others. Consequently, much less than 5% of sample paths will be as extreme, in terms of financial impact, as the boundaries of the prediction intervals.

### 5.3 Prediction intervals for projected forces of mortality

Figure 5.6 extends the contour plots of the logarithm of the force of mortality in Figure 1.2 to future years, using a deterministic projection with all the future innovation terms set to zero. The fact that the bands of colour in the graphs continue along a broadly linear path indicates that we are forecasting a broadly constant future percentage improvement rate in mortality.

Figures 5.7-5.9 show 95% prediction intervals for future forces of mortality at specific ages, for all three data sets, based on the 2.5<sup>th</sup> and 97.5<sup>th</sup> percentiles. Unlike the prediction intervals for the future parameter values calculated in Section 5.2, these prediction intervals are based on the simulation output.

Superimposed on Figures 5.7-5.9 are the future forces of mortality given by a number of previously published projections. These projections were all obtained from version 1.0 of the 'library' of mortality projections issued by the CMI in 2007. They were applied to the 2005 mortality curves fitted in this thesis, before the introduction of gamma parameters, and the gamma parameter for the appropriate year of birth was then applied to the resulting force of mortality, taking the gamma parameter from a deterministic projection of the time series model fitted in this thesis with all the future innovation terms set to zero if the year of birth was after 1955 for CMI data or 1971 for England and Wales data. Further details of the previously published projections are given below.

For CMI data, two of the three previously published projections we consider are the 'Medium Cohort' and 'Long Cohort' projections first proposed in CMI (2002). These projections were issued by the CMI after the favourable cohort effect centred on 1926 had first been observed in CMI data. This cohort effect was not allowed for in the projection originally published with the '92' series of mortality tables in CMI (1999). The Medium Cohort and Long Cohort projections assume higher mortality improvement rates compared with the previous generation than the projection in CMI (1999) for lives born between 1910 and 1942 inclusive. Between 1992 (the base year of the '92' series mortality tables) and 2000, the higher improvement rates apply to the whole of the range of years of birth from 1910 to 1942, but after 2000, the range of years of birth where they apply reduces linearly, reaching zero in 2020 for the Medium Cohort projection and in 2040 for the Long Cohort projection. The range continues to be centred on 1926.

Baxter (2007) argues that the future increases in period life expectancy implied by the CMI projections are too slow compared with what has been observed in the England and Wales population, a separate issue from the cohort effect. Baxter proposes applying an underpin to the improvement rate across all ages and all future years. Thus the third previously published projection we consider for CMI data is the Medium Cohort projection with an underpin of 1% *p.a.* applied to the improvement rate.

The previously published projections we consider for England and Wales data, for both males and females, are those underlying the population projections produced by the Government Actuary's Department (GAD) on the basis of data to 2004 (GAD, 2006). Responsibility for producing the official UK population projections has since been transferred to the Office for National Statistics (ONS). A 'Principal' projection, a 'High Life Expectancy' (HLE) variant and a 'Low Life Expectancy' (LLE) variant were produced. The male projections relate to England, Wales and Northern Ireland, while the female projections relate to the whole of the UK.

The key points to be noted from Figures 5.7-5.9 are as follows:

- For both the male data sets, the model of this thesis predicts higher future mortality improvements, in percentage terms, at older ages than at younger ages. For England and Wales female data, the long-term mortality improvements predicted are broadly independent of age, but faster at the very youngest ages as a result of the negative  $\kappa^{(0)}$  parameters.
- For CMI data, the improvements given by the previously published projections are the same at age 30 as at age 50, as the previously published projections as implemented in the 'library' are the same below age 60 as at age 60. Nevertheless, it is clear that the previously published projections give higher mortality improvements than the model of this thesis in the early years of the projection at younger ages, but that this is no longer the case in the later years of the projection, except in the case of age 30 where a 1% underpin is applied to the improvement rate from the previously published projection.
- For lives aged 70 in CMI data, the three previously published projections are broadly in line with the central projection from the model of this thesis in the early years of the projections. This also applies to the Long Cohort projection, although not to the Medium Cohort or Medium Cohort subject to a 1% underpin, for lives aged 90. However, the Medium Cohort and Long Cohort projections only exhibit higher improvements than the projection in CMI (1999) for one particular generation, so it is to be expected that for subsequent generations, these projections will show lower improvements than the model of this thesis. This can be seen in Figure 5.7(iii) and (iv). Imposing a 1% underpin on the improvement rate is not sufficient to close this gap, because the model of this thesis predicts an improvement rate greater than 1% *p.a.*
- The curves representing both the model of this thesis and the previously published projections become smoother when the year of birth being represented passes 1955
for CMI data or 1971 for England and Wales data. This is because the applicable gamma parameters then begin to come from a projection rather than from the data.

- For England and Wales data, all the curves representing the GAD projections eventually become straight lines, with the lines being horizontal for the LLE projections, downward-sloping for the HLE projections and downward-sloping with half the HLE slope for the Principal projections. This is because the GAD projections have the property that after 2029, mortality improves at a constant rate across all ages and all calendar years, the constant rate being 0% *p.a.* for the LLE projections, 1% *p.a.* for the Principal projections from the model of this thesis, except those at age 30, eventually become close to straight lines, which are reasonably close to being parallel to the green lines, signifying a long-term improvement rate of approximately 1% *p.a.* For males aged 70 and 90, the long-term improvement rate is slightly faster than this.
- The 2.5<sup>th</sup> and 97.5<sup>th</sup> percentile curves from the model of this thesis are less smooth than the median curves. This can be attributed to sampling error, which typically has more effect on the estimation of the more extreme percentiles of a distribution for a given sample size.
- The 2.5<sup>th</sup> percentile curve for England and Wales females aged 30 falls without limit in the late 2030s. This is because, by this point, the force of mortality has become negative in more than 2.5% of scenarios. The structure of the model of this thesis does not impose any theoretical reason why the force of mortality must be positive, and it is the negative values of the  $\kappa^{(0)}$  parameters for females which make it a practical possibility that the force of mortality may become negative at the youngest ages far into the future. It is unlikely that these negative forces of mortality are a major barrier to the practical applicability of the model. For assurance business, the scenario where mortality falls to zero is unlikely to be of much interest as it is favourable for the insurer. For deferred annuity business, as mortality rates are so low around age 30, the value of a deferred annuity is unlikely to be significantly understated as a result of underestimating mortality at these ages. The business affected will, in any case, not come into force for some years.

#### 5.4 Ages above the highest age in the data

In order to use these projections to calculate annuity values, it was necessary to calculate extrapolated mortality rates at ages above the highest age in the data. For the 2005 mortality curve, before the introduction of gamma parameters, the simplest approach is to perform this extrapolation log-linearly, with the slope and intercept of the line set so that the logarithm of the force of mortality is a differentiable function of age at the highest age of the data set. Specifically, let *a* be the highest age of the data set, so that *a* = 90 for CMI data and *a* = 89 for England and Wales data. It is assumed that the force of mortality in 2005 at ages x > a, before the introduction of gamma parameters, is:

$$\mu_{x,2005}^{0} = \mu_{a,2005}^{0} \exp(\lambda[x-a]),$$

where  $\lambda$  is calculated so as to make  $\log \mu_{x,2005}^0$  a differentiable function of *x* at *x* = *a*. We find that:

$$\lambda = \frac{\left[\kappa_{2005}^{(4)} + 2\kappa_{2005}^{(5)}(a-\bar{x})\right]\exp[\kappa_{2005}^{(3)} + \kappa_{2005}^{(4)}(a-\bar{x}) + \kappa_{2005}^{(5)}((a-\bar{x})^2 - \hat{\sigma}_x^2)]}{\mu_{a,2005}^0}.$$

This extrapolation method, which we shall refer to as the 'log-linear extrapolation method', will be used in most of the applications in this chapter. Section A.3.6 of LifeMetrics (2007) suggests an alternative extrapolation method which we shall also investigate for comparison.

In this method, the 2005 force of mortality at ages above *a*, before the introduction of gamma parameters, is assumed to be a cubic polynomial function of age, with coefficients determined by the following four constraints. The log-linear extrapolation method satisfies the first two of these constraints but not (in general) the last two.

- The force of mortality is a continuous function of age *x* at *x* = *a*.
- The derivative of the force of mortality is a continuous function of age x at x = a.
- The force of mortality at age 119 is that corresponding to an initial mortality rate of 0.6, *i.e.* –log (1 – 0.6).
- The derivative of the force of mortality at age 119 is zero.

In this extrapolation method, all remaining lives are assumed to die just before their 120<sup>th</sup> birthday.

Figures 5.10 and 5.11 respectively show the extrapolations that would have been obtained if the log-linear and the LifeMetrics extrapolation method (respectively) had been applied in the past. Unlike Figure 5.6, the mortality shown in the region of the data in these graphs is the fitted mortality, before introducing gamma parameters, rather than the crude mortality.

The conclusion to be drawn from Figures 5.10 and 5.11 is that both extrapolation methods are potentially appropriate for extrapolating the 2005 mortality curve to ages above a but, for different reasons in each case, neither method is likely to be appropriate for extrapolating projected future mortality curves. In the case of the log-linear extrapolation method, there is some evidence of the historical extrapolated mortality curves being unstable from year to year, which is not a desirable feature to incorporate into a projection. This instability arises from the fact that small differences in mortality close to age a tend to be magnified by the extrapolation. In the case of the LifeMetrics extrapolation method, the method assumes that no improvements occur at age 119, which is an extreme assumption. Furthermore, even if this assumption is considered appropriate, then it is still an unnecessarily complex procedure to carry out an extrapolation using the LifeMetrics method for each future year in each scenario of a stochastic projection - it would be simpler to use the 'low improvement assumption' described below. Accordingly we shall not use either the log-linear or the LifeMetrics extrapolation method to extrapolate projected future mortality curves. We shall instead consider two possible assumptions as to how this extrapolation should be carried out. to demonstrate a range of potential outcomes:

Mortality improvements after 2005 at ages above *a*, before the introduction of gamma parameters, occur at the same percentage rate as at age *a* in the same calendar year, *i.e.* for *x* > *a* and *t* > 2005:

$$\mu_{xt}^0 = \frac{\mu_{at}^0}{\mu_{a,2005}^0} \mu_{x,2005}^0 \,.$$

This assumption will be referred to as the 'high improvement assumption'.

Mortality improvements after 2005 at ages close to and above *a*, before the introduction of gamma parameters, occur at a similar rate to those at age *a*. There are no mortality improvements after 2005 at age 120, before the introduction of gamma parameters. Improvement rates at ages between *a* and 120 are derived by linear interpolation. In other words, for  $a < x \le 120$  and t > 2005:

$$\mu_{xt}^{0} = \mu_{x,2005}^{0} \left[ \frac{120 - x}{120 - a} \left( \frac{\mu_{at}^{0}}{\mu_{a,2005}^{0}} \right) + \frac{x - a}{120 - a} \right].$$

This assumption will be referred to as the 'low improvement assumption'.

The high improvement assumption produces lower projected forces of mortality at ages above *a*, and hence higher annuity values, than the low improvement assumption. The high improvement assumption might be considered appropriate if prudent annuity values are required and the low improvement assumption might be considered appropriate if best estimate annuity values are required.

Under the log-linear extrapolation method, no fixed limiting age was used at which all remaining lives were assumed to die. In practice the calculations of annuity values in Section 5.5 could only be performed up to a finite age, but it was verified in each case that allowing for the possibility of survival beyond that age would affect the calculated annuity value by less than  $10^{-3}$ . In the case of the low improvement assumption, it was found in all cases that it was sufficient at this level of accuracy to consider only ages up to 120, so that there was no need to define a formula for the force of mortality at ages over 120 under the low improvement assumption.

#### 5.5 Annuity values

Annuity values under each of the 500 scenarios were calculated for lives who had just attained each of the age labels 35, 45, 55, 65 and 75 at the end of 2005, based on all three data sets, under both the low and the high improvement assumptions. As a result of the age definitions used, *i.e.* age nearest birthday for CMI data and age last birthday for England and Wales data, this means that the exact ages of the lives at the end of 2005 were 34.5,44.5,...,74.5 for CMI data and 35,45,...,75 for England and Wales data, but for simplicity we shall refer to the ages only as 35,45,...,75 in 2005 in what follows. The annuities for lives aged 35, 45 and 55 in 2005 were deferred annuities payable from age label 65, *i.e.* from an exact age of 64.5 for CMI data and 65 for England and Wales data, and the annuities for lives aged 65 and 75 in 2005 were immediate annuities. The log-linear extrapolation method was used. A deterministic interest rate of 4.5% *p.a.* was used, and the annuities were assumed to be level single life annuities of 1 *p.a.* payable annually in advance. Thus the annuity value under the  $k^{th}$  scenario was calculated as:

$$\sum_{i=x_0-x}^{\infty} v^i S^k(x,i),$$

where:

$$S^{k}(x,i) = \exp\left(-\sum_{j=0}^{i-1} \mu_{x+j,2006+j}^{k}\right)$$

is the proportion of lives having just attained the age label *x* at the end of 2005 who survive for a further *i* years, v = 1 / 1.045,  $x_0 = \max\{x, 65\}$  and the superscript *k* represents the  $k^{\text{th}}$  scenario.

For CMI data, for lives aged 35 and 45 in 2005, the year of birth is outside the range where we identified our estimates of the gamma parameters as being reliable in Section 4.6. Accordingly, in these cases, instead of using the gamma parameter estimated from the data, it was taken from the stochastic projection. In all other cases, the gamma parameter estimated from the data was used.

Tables 5.2-5.14 show the mean, standard deviation and some key percentiles of the empirical distributions of the annuity values. Although not all the figures in these tables will be referred to in what follows, the same information is shown in all the tables for consistency. Table 5.1 shows which data set, which improvement assumption (high or low) and which extrapolation method were used in each of the tables 5.2-5.14. Table 5.1 also shows that, for one of the 12 possible combinations of data set, improvement assumption and extrapolation method, a

table was calculated using 5,000 scenarios in addition to the 500-scenario table. This was to investigate the extent to which it would be desirable to add more scenarios.

The following are the key observations from Tables 5.2-5.14:

- For all five values of the age in 2005, under both the low and the high improvement assumptions and under both the log-linear and the LifeMetrics extrapolation methods, the mean annuity value for England and Wales male data is less than that for England and Wales female data, which is less than that for CMI data. This ordering of the mean annuity values indicates that the mortality differential between assured lives and the general population is more significant than that between males and females.
- As the age in 2005 increases from x to x + 10, where x = 35, 45 or 55, *i.e.* before the annuities come into payment, we might expect the mean annuity value to increase by a factor close to (1.045)<sup>10</sup>, as we are discounting for 10 fewer years. The fact that lives aged x in 2005 benefit from 10 years of mortality improvements compared with lives aged x + 10 in 2005 would tend to make the factor less than this, whereas the possibility that a life aged x in 2005 will not survive to age x + 10 would tend to make the factor greater than this. In fact, we observe that the factor is less than (1.045)<sup>10</sup> in all cases for an increase in age in 2005 from 35 to 45 or from 45 to 55. From 55 to 65, the factor is greater than  $(1.045)^{10}$  except for CMI data under the high improvement assumption. This seems reasonable as there is a relatively significant probability of a life aged 55 dying before age 65. Only in the CMI data set under the high improvement assumption, with relatively high improvements over the period of the data, a relatively small probability of death between ages 55 and 65 and an equally high percentage rate of improvement above age 90 as at age 90, do the improvements outweigh the possibility of death between ages 55 and 65 to give a value of the factor less than  $(1.045)^{10}$ .
  - As we move from the low to the high improvement assumption, it is clear that the mean annuity value should increase. The standard deviation of the annuity values, as a percentage of the mean, also increases. The justification for this is that, under the low improvement assumption, the extent of uncertainty about future mortality falls after age 90 (89 for England and Wales data), reducing to zero at age 120, whereas under the high improvement assumption, the uncertainty about future mortality, in percentage terms, is the same above age 90 as at age 90.
  - The standard deviation of the annuity values, as a percentage of the mean, tends to be greater for younger than for older generations. This is not surprising in view of the remark made in Section 5.2 that under this model, uncertainty about the overall level of mortality increases significantly the further into the future projections are made (though uncertainty about the shape of the mortality curve does not increase significantly). The movement between ages 65 and 75 in 2005 is an exception to this. The increase in the standard deviation as a percentage of the mean from age 65 to 75 in 2005 is a consequence of the removal of the first ten years' annuity payments, which have a relatively high probability of being made under all scenarios.
  - In terms of impact on the annuity values, the difference between the high and the low improvement assumptions is most significant (by far) for CMI data and least significant for England and Wales female data. As the CMI data set has generally lower base mortality and higher improvements than the England and Wales data sets, a higher proportion of lives survive to an age where the difference between the high and low improvement assumptions becomes relevant. The relatively high improvements also make the differences between projected mortality rates under the high and low improvement assumptions relatively large for CMI data. As the England and Wales female data set has generally lower improvements than the male data sets, the difference between projected mortality rates under the high and low improvement assumptions relatively large for CMI data. As the England and Wales female data set has generally lower improvements than the male data sets, the difference between projected mortality rates under the high and low improvement assumptions is less significant.

- Compared with the log-linear extrapolation method, the LifeMetrics extrapolation method produces lower mean annuity values for CMI data and higher mean annuity values for both England and Wales data sets. The differences are smaller than the differences between the low and high improvement assumptions. To gain an understanding of why the change of extrapolation method has the opposite effect for CMI data to that for England and Wales data, the graphs in Figure 5.12 were plotted. In these graphs, the solid curves, representing the extrapolated 2005 mortality curves under the log-linear extrapolation method, before the introduction of gamma parameters, are straight lines above age 90 (89 for England and Wales data). The dashed curves represent the LifeMetrics extrapolation method. We see that for the England and Wales data sets, mortality is lower under the LifeMetrics extrapolation method than under the log-linear extrapolation method at all ages above 89, which explains why the LifeMetrics annuity values are higher. For CMI data, the LifeMetrics extrapolation method produces slightly higher forces of mortality than the log-linear method from age 90 up to and including age 103, but significantly lower forces of mortality above age 103. The proportion of lives surviving to age 103 is sufficiently small that the higher forces of mortality from age 90 to age 103 outweigh the lower forces of mortality above age 103. Thus the LifeMetrics annuity values are lower than the log-linear annuity values.
- A comparison of the mean and median annuity values in Table 5.8 with those in Table 5.5 reveals that if we are only interested in the mean or median annuity values, then there is little benefit to be gained from considering more than 500 scenarios as the sampling errors incurred as a result will generally be smaller than the impact of the choice of extrapolation method, and much smaller than the impact of the choice of extrapolation. If we are interested in relatively extreme percentiles of the distributions of annuity values, then it becomes more important to use a large number of scenarios. In this thesis, although we shall make some reference to percentiles of the distributions, including some relatively extreme percentiles, we are primarily interested in mean annuity values, with Section 5.6 focusing on the distribution of the conditional mean annuity value after one year.

The annuity values we have calculated will now be compared with values from the same previously published projections as referred to in Section 5.3. The values from the previously published projections are shown in Tables 5.15-5.17, and were calculated consistently with the values in Tables 5.2-5.8. In particular, the log-linear extrapolation method was used. We shall discuss the results in Tables 5.15-5.17 for ages 35-65 in 2005 first, and then discuss the results for age 75 in 2005.

For lives aged 65 in 2005 in CMI data, the model of this thesis generally predicts higher annuity values, *i.e.* greater mortality improvements, than the previously published projections. The mean annuity value of 14.305 under the high improvement assumption is higher than all the values from the previously published projections, and the mean value of 14.160 under the low improvement assumption is only slightly lower than the value of 14.182 from the Long Cohort projection. Under the low improvement assumption, over 90% of the annuity values exceed that under the Medium Cohort projection, just under 90% exceed that under the Medium Cohort projection. Under the high improvement assumption, just under 50% exceed that under the Long Cohort projection. Under the high improvement assumption, just under 95% of the annuity values exceed that under the Medium Cohort projection, between 90% and 95% exceed that under the Medium Cohort projection subject to a 1% underpin, and just underpin and between 60% and 70% exceed that under the Long Cohort projection.

For younger generations in CMI data, the annuity values from the previously published projections move further towards the lower end of the ranges of annuity values from the model of this thesis. In the extreme, for lives aged 35 in 2005, all but two of the 500 annuity values under the low improvement assumption exceed even that from the Long Cohort projection and all 500 annuity values under the high improvement assumption exceed even that from the Long Cohort projection.

For lives aged 65 in 2005 in England and Wales male data, the model of this thesis predicts generally lower annuity values, *i.e.* lower mortality improvements, than the GAD projections. The mean annuity values of 12.546 under the low improvement assumption and 12.591 under the high improvement assumption are lower than all the values from the GAD projections. The annuity values from the GAD projections do, however, lie within the ranges produced by the model of this thesis. Between 30% and 40% of the annuity values under the low improvement assumption, exceed that from the LLE projection. Between 5% and 10% of the annuity values under the low improvement assumption, and between 10% and 20% of the annuity values under the high improvement assumption, exceed that from the Principal projection. Between 0.5% and 1% of the annuity values under the low improvement assumption, exceed that from the Principal projection. Between 0.5% of the annuity values under the high improvement assumption, exceed that from the Principal projection. Between 0.5% of the annuity values under the high improvement assumption, exceed that from the Principal projection. Between 0.5% of the annuity values under the high improvement assumption, and just under 2.5% of the annuity values under the high improvement assumption.

For younger generations in England and Wales male data, the model of this thesis gives annuity values more in line with those from the GAD projections. For lives aged 55 in 2005. the mean annuity value under both the low and high improvement assumptions lies between the values from the LLE and Principal projections. For lives aged 45 in 2005, the mean annuity value under the low improvement assumption lies between the values from the LLE and Principal projections, while the mean annuity value under the high improvement assumption lies between the values from the Principal and HLE projections. For lives aged 35 in 2005, the mean annuity value under both the low and high improvement assumptions lies between the values from the Principal and HLE projections. However, one notable feature of Table 5.16 compared with Tables 5.4 and 5.5 is that the difference between the LLE, Principal and HLE projections is wider compared with the range of annuity values from the model of this thesis for younger generations than for older generations. The explanation for this is that the range of annuity values from the model of this thesis represents the potential impact of stochastic fluctuations, which have some tendency to average out over long periods, whereas the difference between the LLE, Principal and HLE projections represents the potential impact of parameter risk, i.e. the risk that parameters have been estimated incorrectly. As mentioned in Section 5.3, the LLE, Principal and HLE projections all assume that improvement rates tend towards a common value across all ages between 2004 (the base year of the projections) and 2029, the difference between the projections being that this common value is set to zero for the LLE projection, 1% for the Principal projection and 2% for the HLE projection. It would be a valuable topic for further research to produce stochastic projections that allow for both stochastic fluctuations and parameter risk together.

For England and Wales female data, the model of this thesis again generally predicts lower annuity values than the GAD projections. For females aged 65 in 2005, the mean annuity values under both the low and high improvement assumptions are less than the values from the GAD projections. This continues to be the case for females aged 55 in 2005, but for females aged 35 and 45 in 2005, the mean annuity value is between the LLE projection and the Principal projection. For females aged 65 in 2005, the annuity value from the Principal projection is between the 95<sup>th</sup> and 97.5<sup>th</sup> percentiles of the distribution of annuity values from the model of this thesis under the low improvement assumption and between the 90<sup>th</sup> and 95<sup>th</sup> percentiles under the high improvement assumption. As the age in 2005 decreases, the Principal annuity value moves rather closer to the centre of the distribution, being close to the 80<sup>th</sup> percentile under the high improvement assumption, and between the 80<sup>th</sup> and 90<sup>th</sup> percentiles under the low improvement assumption, for age 35 in 2005. As for England and Wales male data, the difference between the LLE, Principal and HLE projections is wider compared with the range of annuity values from the model of this thesis for younger than for older generations.

As the age in 2005 increases from 65 to 75, the trend for CMI data for the annuity values from the model of this thesis to decrease relative to those from the previously published projections with increasing age continues, but under the Medium Cohort projection and the Medium Cohort projection subject to a 1% underpin, it remains the case that the mean annuity value from the previously published projection is less than from the model of this thesis, under both the low and high improvement assumptions. For England and Wales data, however, the trend reverses and the annuity values from the previously published projections move closer to the centre of the distribution from the model of this thesis. Despite this, the values from the

previously published projections remain above the mean values from the model of this thesis, with the exception of the high improvement assumption compared with the LLE projection for male data.

To gain a greater understanding of the reasons for the differences between Tables 5.2-5.7 and Tables 5.15-5.17, Figures 5.13-5.24 were plotted. Figures 5.13-5.18 compare the distribution of the projected forces of mortality at each future age with the previously published projections, and are similar to Figures 5.7-5.9 except that we are following a particular generation as it ages rather than considering what mortality rate each generation experiences as it passes a particular age. Figures 5.19-5.24 compare the distribution of the projected number of survivors to each future age with the previously published projections.

The following are the key observations from Figures 5.13-5.18:

- For CMI data, most of the stochastic scenarios produce higher forces of mortality in the first year of the projection, for the ages considered, than the previously published projections. Thus the improvement rates from over 2006 predicted by the model of this thesis are generally lower than those shown in the previously published projections.
- For England and Wales data, for both males and females, most of the stochastic scenarios again produce higher forces of mortality in the first year of the projection than the previously published projections, although there are exceptions. This again indicates generally lower 2006 improvement rates predicted by the model of this thesis than by the previously published projections.
- All the graphs for CMI data show that the Medium Cohort projection, the Medium Cohort projection subject to a 1% underpin and the Long Cohort projection are identical below age 65. The reason for this is that by 2005, the range of years of birth for which the Medium Cohort projection is different from the projection in CMI (1999) has narrowed to 1914-1938, and the range of years of birth for which the Long Cohort projection is different has narrowed to 1912-1940. Lives born in 1940 are aged 65 in 2005. In addition, for lives in the age range considered in 2005, the 1% underpin does not 'bite' until after the lives have reached age 65.
- For England and Wales data, for both males and females, the differences between the three GAD projections in the early years are smaller than the range of the stochastic scenarios. This is because the differences between the GAD projections relate to the long-term value improvement rates tend towards, whereas the range of the stochastic scenarios relates to random fluctuations.
- For CMI data, for lives aged 35, 45 and 55 in 2005, most of the stochastic scenarios continue to produce higher forces of mortality than the previously published projections until the life passes age 60. In the model of this thesis, mortality at these younger ages is driven mainly by the  $\kappa^{(0)}$  parameters, which are assumed to revert to a mean level, and so improvement rates are lower below age 60 than at age 60. However, in the previously published projections, as defined in the 'library', future mortality improvement rates below age 60 are assumed to be the same as at age 60.
- For England and Wales data, for both males and females, most of the scenarios produce higher forces of mortality over the age range 60-80 than the Principal GAD projection. This effect is generally stronger for females than for males. The differences tend to be larger for higher values of the age in 2005, although the reverse is the case as the age in 2005 increases from 65 to 75.
- For CMI data, as the age approaches 90, mortality from the model of this thesis decreases relative to mortality from the previously published projections. The previously published projections exhibit mortality improvements which slow down as the age approaches 90. The model of this thesis, on the other hand, forecasts the same long-term percentage rate of improvement at all ages, because the

improvements are driven entirely by the downward trend in the  $\kappa^{(3)}$  parameters. (An exception to this is at the youngest ages, where the positive  $\kappa^{(0)}$  parameters cause mortality to improve more slowly than this.)

- At the oldest ages, under the low improvement assumption, the model of this thesis produces higher forces of mortality than any previously published projection under which the improvement rates do not approach zero as the age approaches 120. For CMI data, this applies to the Medium Cohort projection subject to a 1% underpin. For both the England and Wales data sets, this applies to the Principal projection and to the HLE projection.
  - At the oldest ages under the high improvement assumption for CMI data, all the stochastic scenarios produce lower forces of mortality than the Medium Cohort and Long Cohort projections, as mortality improvements under the model of this thesis do not approach zero at the oldest ages but those under the Medium Cohort and Long Cohort projections do. There are only a few scenarios producing higher forces of mortality than the Medium Cohort projection subject to a 1% underpin, *i.e.* there are only a few scenarios for which the model of this thesis predicts improvements of less than 1% *p.a.* at the oldest ages.
- At the oldest ages under the high improvement assumption for England and Wales data, most or all of the stochastic scenarios produce higher forces of mortality than both the Principal and HLE projections, the former representing a long-term improvement rate of 1% *p.a.* and the latter representing a long-term improvement rate of 2% *p.a.* Most or all of the scenarios produce lower forces of mortality than the LLE projection, as the improvement rates under the scenarios do not approach zero as the age approaches 120 but those under the LLE projection do.
- In most of the graphs, the interval between the 2.5<sup>th</sup> and 97.5<sup>th</sup> percentiles widens with increasing age up to age 90 (89 for England and Wales data), as a result of the expanding funnels of doubt in Figures 5.1-5.5.
- In the graphs for age 35 in 2005 for CMI data (Figures 5.13(i) and 5.14(i)), the interval between the 2.5<sup>th</sup> and 97.5<sup>th</sup> percentiles stands out as being particularly wide in the early years of the projection. This is due to two main factors. Firstly, the gamma parameter comes from the projection rather than from the data and is therefore subject to stochastic variation the only other graphs where the gamma parameter comes from the projection are Figures 5.13(ii) and 5.14(ii). Secondly, the uncertainty in the  $\kappa^{(0)}$  parameters gives rise to significant uncertainty in log mortality at younger ages.
- As the age increases above 90 (89 for England and Wales data), the width of the interval between the 2.5<sup>th</sup> and 97.5<sup>th</sup> percentiles under the high improvement assumption remains approximately constant. There is a slight increase arising from the continued expansion of the funnel of doubt in the  $\kappa^{(3)}$  parameters, but the funnels of doubt in the other parameters have almost reached their ultimate width by this point. Under the low improvement assumption, the width of the interval decreases, reaching zero at age 120 where the projected force of mortality under all scenarios is the same as that extrapolated from the 2005 mortality curve.

The vertical position of a point in Figures 5.19-5.24 is determined by the vertical positions of all the points to the left of it in Figures 5.13-5.18, *i.e.* the proportion of lives surviving to a given age is determined by mortality rates at all younger ages. Thus Figures 5.19-5.24 contain the same information as Figures 5.13-5.18, with the exception that the range of the stochastic scenarios is rather narrower in Figures 5.19-5.24 because of scenarios that have high mortality in some future years but low mortality in others. The main reason why the presentation of Figures 5.19-5.24 is useful is to give an understanding of how the different projected forces of mortality in Figures 5.13-5.18 affect annuity values. The areas under the graphs in Figures 5.19-5.24 are expectations of life, *i.e.* annuity values calculated at a 0% interest rate.

#### 5.6 Individual Capital Assessments (ICAs)

We shall now apply the model to risk-based capital calculations under the Individual Capital Assessment (ICA) regime of the UK insurance regulator, the Financial Services Authority (FSA), for deferred and immediate annuities. We shall continue to use a deterministic interest rate of 4.5%. Mortality at ages above 90 (89 for England and Wales data) will be assumed to be in accordance with the log-linear extrapolation method and the high improvement assumption. For lives aged 35 and 45 in 2005 in CMI data, we shall not use the stochastic projection of the gamma parameters, but we shall instead use a deterministic projection with all the future innovation terms set to zero, as used for the previously published projections in Section 5.3.

Section 7.1.42R of the FSA's Prudential Sourcebook for Insurers (INSPRU) states:

'Where the FSA requests a firm to submit to it a written record of the firm's assessments of the adequacy of its capital resources...those assessments must include an assessment comparable to a 99.5% confidence level over a one year timeframe that the value of assets exceeds the value of liabilities, whether or not this is the confidence level otherwise used in the firm's own assessments.'

If the calculation described is performed by a simulation approach, then it requires a number of scenarios to be generated for the kappa parameter values, and the resulting mortality experience, in 2006 (the first year beyond the period of the data). For lives that survive 2006, the mean annuity value must then be calculated separately under each scenario. These mean annuity values need to be multiplied by the probability of surviving 2006 in the same scenario, and the 99.5<sup>th</sup> percentile of the resulting distribution needs to be taken. As the mean annuity values themselves need to be calculated by a simulation approach, we are faced with having to run simulations within simulations, which is highly computationally intensive. Accordingly we shall develop a less computationally intensive approximation to the ICA, and carry out some testing to confirm that the approximation is sufficiently accurate.

Consider a group of *N* deferred or immediate annuitants who have just attained the age label *x* at the start of calendar year *t*, where calendar year t - 1 is the last year of the data. (The assumption that the annuitants have just attained a particular age label is not critical to the analysis, but has been made to simplify the presentation.) The annuities are assumed to be payable annually in advance.  $x_1$  denotes the age label from which the annuities are payable for deferred annuities, and is set equal to x + 1 for immediate annuities. The probability of a particular one of the annuitants surviving to age label x + 1 is:

$$\exp(-\mu_{xt}) = \exp(-\gamma_{t-x}[\kappa_t^{(0)} + \exp\{\kappa_t^{(3)} + \kappa_t^{(4)}(x-\bar{x}) + \kappa_t^{(5)}((x-\bar{x})^2 - \hat{\sigma}_x^2)\}]).$$

The expected number of annuitants surviving to age label x + 1 is therefore:

$$N \exp(-\gamma_{t-x} [\kappa_t^{(0)} + \exp\{\kappa_t^{(3)} + \kappa_t^{(4)} (x - \bar{x}) + \kappa_t^{(5)} ((x - \bar{x})^2 - \hat{\sigma}_x^2)\}])$$

In what follows, we shall assume that N is sufficiently large that the actual number of annuitants surviving to age label x + 1 may be assumed to be equal to the above value.

We now use the time series models fitted in Chapter 4 to express the kappa parameters for year *t* in terms of the innovations for year *t* and other quantities that are assumed to be known:

$$\kappa_t^{(i)} = \mu^{(i)} + \alpha^{(i)} (\kappa_{t-1}^{(i)} - \mu^{(i)}) + \sigma^{(i)} Z_t^{(i)}, i = 0, 4, 5,$$
  

$$\kappa_t^{(3)} = \kappa_{t-1}^{(3)} + \mu^{(3)} + \sigma^{(3)} Z_t^{(3)} + \sigma^{(3)} \beta^{(3)} Z_{t-1}^{(3)},$$

where  $Z_t^{(0)}$ ,  $Z_t^{(3)}$ ,  $Z_t^{(4)}$  and  $Z_t^{(5)}$  are jointly normally distributed with zero mean, unit variance and the correlations shown in Section 4.7. A small letter *z* has been used for the innovation term  $z_{t-1}^{(3)}$  from year *t* – 1 to signify that the value is assumed known.

The number of annuitants surviving to age label x + 1 is then:

$$N \exp \left(-\gamma_{t-x} \left[ \mu^{(0)} + \alpha^{(0)} (\kappa_{t-1}^{(0)} - \mu^{(0)}) + \sigma^{(0)} Z_{t}^{(0)} + \alpha^{(0)} Z_{t-1}^{(0)} + \alpha^{(0)} Z_{t-1}^{(0)}$$

We now linearise this expression in the  $Z_t^{(l)}$ . We shall refer to this linearisation as Approximation 1 in what follows. We shall later carry out some testing to confirm that the error introduced by the linearisation is acceptably small. We find that the number of annuitants surviving to age label x + 1 is:

$$N\overline{p}_{xt}[1+a_0Z_t^{(0)}+a_3Z_t^{(3)}+a_4Z_t^{(4)}+a_5Z_t^{(5)}],$$

where:

$$\begin{aligned} a_{0} &= -\gamma_{t-x} \sigma^{(0)}, \\ a_{3} &= -\gamma_{t-x} \sigma^{(3)} \overline{\mu}_{\exp}(x,t), \\ a_{4} &= -\gamma_{t-x} \sigma^{(4)} (x - \overline{x}) \overline{\mu}_{\exp}(x,t), \\ a_{5} &= -\gamma_{t-x} \sigma^{(5)} ((x - \overline{x})^{2} - \hat{\sigma}_{x}^{2}) \overline{\mu}_{\exp}(x,t), \\ \overline{\kappa}_{t}^{(i)} &= \mu^{(i)} + \alpha^{(i)} (\kappa_{t-1}^{(i)} - \mu^{(i)}), i = 0, 4, 5, \\ \overline{\kappa}_{t}^{(3)} &= \kappa_{t-1}^{(3)} + \mu^{(3)} + \sigma^{(3)} \beta^{(3)} z_{t-1}^{(3)}, \\ \overline{\mu}_{\exp}(x,t) &= \exp\{\overline{\kappa}_{t}^{(3)} + \overline{\kappa}_{t}^{(4)} (x - \overline{x}) + \overline{\kappa}_{t}^{(5)} ((x - \overline{x})^{2} - \hat{\sigma}_{x}^{2})\}, \\ \overline{\mu}_{xt} &= \gamma_{t-x} [\overline{\kappa}_{t}^{(0)} + \overline{\mu}_{\exp}(x,t)], \\ \overline{p}_{xt} &= \exp(-\overline{\mu}_{xt}). \end{aligned}$$

Consider an annuitant who has survived to age label x + 1. We express the (random) value at the start of year t + 1 of the liabilities in respect of this annuitant, L(t + 1), in terms of the  $Z_t^{(l)}$ , some 'future annuity values' independent of the  $Z_t^{(l)}$ , some of which have payments varying over time and/or are calculated at revised interest rates, and other quantities that are assumed to be known. In deriving this expression, we again linearise in the  $Z_t^{(l)}$ , and we shall refer to this linearisation as Approximation 2 in what follows. Appendix D shows that the resulting expression is:

$$L(t+1) = FutAnnVal(x,t) + b_0 Z_t^{(0)} + b_3 Z_t^{(3)} + b_4 Z_t^{(4)} + b_5 Z_t^{(5)},$$

where:

$$\begin{split} b_{0} &= \frac{\gamma_{t-s} \sigma^{(0)} (\alpha^{(0)})^{s-s+1} (e^{\lambda} - 1)}{(1 - \alpha^{(0)})^{t} - \alpha^{(0)} e^{\lambda}} FutAnnVal_Maxa(x,t) \\ &+ \frac{\gamma_{t-s} \sigma^{(0)} \alpha^{(0)}}{1 - \alpha^{(0)}} FutAnnVal_Maxa_a alpha0(x,t) \\ &- \gamma_{t-s} \sigma^{(0)} \left( \frac{\alpha^{(0)} (1 - (\alpha^{(0)})^{s-x-1})}{1 - \alpha^{(0)} e^{\lambda}} \right) FutAnnVal(x,t) \\ &+ \frac{\gamma_{t-s} \sigma^{(0)} \alpha^{(0)} e^{-(s-s-1)\lambda}}{1 - \alpha^{(0)} e^{\lambda}} FutAnnVal_Mina_alpha0_almbda(x,t), \\ &+ \frac{\gamma_{t-s} \sigma^{(0)} (\alpha^{(0)} e^{-(s-s-1)\lambda})}{1 - \alpha^{(0)} e^{\lambda}} FutAnnVal_Mina_alpha0_almbda(x,t), \\ &= \gamma_{t-s} \sigma^{(3)} (1 + \beta^{(3)}) FutAnnVal_Mina_alpha0_almbda(x,t), \\ &= \gamma_{t-s} \sigma^{(4)} FutAnnVal_Varying 4(x,t), \\ &= \gamma_{t-s} \sigma^{(5)} FutAnnVal_Varying 5(x,t), \\ FutAnnVal_(x,t) &= \sum_{i=x_{t-s-1}}^{\infty} v^{i} \exp\left(-\sum_{j=1}^{i} \overline{\mu}_{x+j,t+j}\right), \\ FutAnnVal_Maxa_a(x,t) &= \sum_{i=x_{t-s-1}}^{\infty} v^{i} \exp\left(-\sum_{j=1}^{i} \overline{\mu}_{x+j,t+j}\right), \\ FutAnnVal_Mina_a alpha0_almbda(x,t) &= \sum_{i=x_{t-s-1}}^{\infty} (v\alpha^{(0)})^{i} \exp\left(-\sum_{j=1}^{i} \overline{\mu}_{x+j,t+j}\right), \\ FutAnnVal_Varying 3(x,t) &= \sum_{i=x_{t-s-1}}^{s-s-1} v^{i} \left(\sum_{j=1}^{i} \overline{\mu}_{eeg}(x+j,t+j)\right) \exp\left(-\sum_{j=1}^{i} \overline{\mu}_{eeg}(x+j,t+j)\right) \right) \\ FutAnnVal_Varying 4(x,t) &= \sum_{i=x_{t-s-1}}^{s-s-1} v^{i} \left(\sum_{j=1}^{i} (\alpha^{(4)})^{j} (x+j-\bar{x}))\overline{\mu}_{eeg}(x+j,t+j)\right) \\ + \sum_{i=\sigma-x}^{s} \exp\left(-\sum_{j=1}^{i} \overline{\mu}_{x+j,t+j}\right) \\ FutAnnVal_Varying 5(x,t) &= \sum_{i=x_{t-s-1}}^{s-s-1} v^{i} \left(\sum_{j=1}^{i} (\alpha^{(5)})^{j} ((x+j-\bar{x}))^{2} - \hat{\sigma}_{x}^{2}) \overline{\mu}_{eeg}(x+j,t+j)\right) + \\ FutAnnVal_Varying 5(x,t) &= \sum_{i=x_{t-s-1}}^{s-s-1} v^{i} \left(\sum_{j=1}^{i} (\alpha^{(5)})^{j} ((x+j-\bar{x}))^{2} - \hat{\sigma}_{x}^{2}) \overline{\mu}_{eeg}(x+j,t+j)\right) + \\ FutAnnVal_Varying 5(x,t) &= \sum_{i=x_{t-s-1}}^{s-s-1} v^{i} \left(\sum_{j=1}^{i} (\alpha^{(5)})^{j} ((x+j-\bar{x}))^{2} - \hat{\sigma}_{x}^{2}) \overline{\mu}_{eeg}(x+j,t+j)\right) + \\ FutAnnVal_Varying 5(x,t) &= \sum_{i=x_{t-s-1}}^{s-s-1} exp\left(-\sum_{j=1}^{i} \overline{\mu}_{x+j,t+j}\right) \\ FutAnnVal_Varying 5(x,t) &= \sum_{i=x_{t-s-1}}^{s-s-1} exp\left(-\sum_{j=1}^{i} \overline{\mu}_{x+j,t+j}\right) + \\ FutAnnVal_Varying 5(x,t) &= \sum_{i=x_{t-s-1}}^{s-s-1} exp\left(-\sum_{j=1}^{i} \overline{\mu}_{x+j,t+j}\right) \\ FutAnnVal_Varying 5(x,t) &= \sum_{i=x_{t-s-1}}^{s-s-1} exp\left(-\sum_{j=1}^{i} \overline{\mu}_{x+j,t+j}\right) \\ FutAnnVal_V$$

$$+\sum_{i=a-x}^{\infty} v^{i} \begin{pmatrix} \sum_{j=1}^{a-x-1} (\alpha^{(5)})^{j} ((x+j-\overline{x})^{2} - \hat{\sigma}_{x}^{2}) \overline{\mu}_{\exp}(x+j,t+j) \\ + ((a-\overline{x})^{2} - \hat{\sigma}_{x}^{2}) \sum_{j=a-x}^{i} (\alpha^{(5)})^{j} \overline{\mu}_{\exp}(x+j,t+j) \end{pmatrix} \\ \exp \left(-\sum_{j=1}^{i} \overline{\mu}_{x+j,t+j}\right) \end{pmatrix}$$

and  $\lambda$  is as in Section 5.4.

Multiplying this by the number of survivors to age label x + 1 gives the (random) value at the start of year t + 1 of all the liabilities together as:

$$\begin{split} N\overline{p}_{xt} [1 - \gamma_{t-x} \sigma^{(0)} Z_{t}^{(0)} - \gamma_{t-x} \sigma^{(3)} \overline{\mu}_{exp}(x,t) Z_{t}^{(3)} - \gamma_{t-x} \sigma^{(4)}(x-\overline{x}) \overline{\mu}_{exp}(x,t) Z_{t}^{(4)} \\ &- \gamma_{t-x} \sigma^{(5)} ((x-\overline{x})^{2} - \hat{\sigma}_{x}^{2}) \overline{\mu}_{exp}(x,t) Z_{t}^{(5)}] \\ \\ \begin{bmatrix} FutAnnVal(x,t) + \frac{\gamma_{t-x} \sigma^{(0)} (\alpha^{(0)})^{a-x+1} (e^{\lambda} - 1) Z_{t}^{(0)}}{(1 - \alpha^{(0)})(1 - \alpha^{(0)} e^{\lambda})} FutAnnVal_Maxa(x,t) \\ &+ \frac{\gamma_{t-x} \sigma^{(0)} \alpha^{(0)} Z_{t}^{(0)}}{1 - \alpha^{(0)}} FutAnnVal_Maxa_alpha0(x,t) \\ &- \gamma_{t-x} \sigma^{(0)} Z_{t}^{(0)} \left( \frac{\alpha^{(0)} (1 - (\alpha^{(0)})^{a-x-1})}{1 - \alpha^{(0)}} + \frac{(\alpha^{(0)})^{a-x}}{1 - \alpha^{(0)} e^{\lambda}} \right) FutAnnVal(x,t) \\ &+ \frac{\gamma_{t-x} \sigma^{(0)} \alpha^{(0)} e^{-(a-x-1)\lambda} Z_{t}^{(0)}}{1 - \alpha^{(0)}} FutAnnVal_Mina_alpha0_alpha0_alpha0(x,t) \\ &- \gamma_{t-x} \sigma^{(3)} (1 + \beta^{(3)}) Z_{t}^{(3)} FutAnnVal_Val_Varying3(x,t) \\ &- \gamma_{t-x} \sigma^{(4)} Z_{t}^{(4)} FutAnnVal_Val_Varying4(x,t) - \gamma_{t-x} \sigma^{(5)} Z_{t}^{(5)} FutAnnVal_Varying5(x,t) \\ &= N\overline{p}_{xt} [c_0 Z_{t}^{(0)} + c_3 Z_{t}^{(3)} + c_4 Z_{t}^{(4)} + c_5 Z_{t}^{(5)}] \end{split}$$

after linearisation in the  $Z_t^{\left(l
ight)}$  , where:

$$\begin{split} c_{0} &= \frac{\gamma_{t-x} \sigma^{(0)} (\alpha^{(0)})^{a-x+1} (e^{\lambda} - 1)}{(1 - \alpha^{(0)})(1 - \alpha^{(0)} e^{\lambda})} FutAnnVal\_Maxa(x,t) \\ &+ \frac{\gamma_{t-x} \sigma^{(0)} \alpha^{(0)}}{1 - \alpha^{(0)}} FutAnnVal\_Maxa\_alpha0(x,t) \\ &- \gamma_{t-x} \sigma^{(0)} \left( 1 + \frac{\alpha^{(0)} (1 - (\alpha^{(0)})^{a-x-1})}{1 - \alpha^{(0)}} + \frac{(\alpha^{(0)})^{a-x}}{1 - \alpha^{(0)} e^{\lambda}} \right) FutAnnVal(x,t) \\ &+ \frac{\gamma_{t-x} \sigma^{(0)} \alpha^{(0)} e^{-(a-x-1)\lambda}}{1 - \alpha^{(0)} e^{\lambda}} FutAnnVal\_Mina\_alpha0\_lambda(x,t), \\ &c_{3} &= -\gamma_{t-x} \sigma^{(3)} \overline{\mu}_{exp}(x,t) FutAnnVal(x,t) \\ &- \gamma_{t-x} \sigma^{(3)} (1 + \beta^{(3)}) FutAnnVal\_Varying3(x,t), \end{split}$$

$$c_{4} = -\gamma_{t-x}\sigma^{(4)}(x-\bar{x})\overline{\mu}_{exp}(x,t)FutAnnVal(x,t)$$
  
-  $\gamma_{t-x}\sigma^{(4)}FutAnnVal\_Varying4(x,t),$   
 $c_{5} = -\gamma_{t-x}\sigma^{(5)}((x-\bar{x})^{2} - \hat{\sigma}_{x}^{2})\overline{\mu}_{exp}(x,t)FutAnnVal(x,t)$   
-  $\gamma_{t-x}\sigma^{(5)}FutAnnVal Varying5(x,t).$ 

This linearisation will be referred to as Approximation 3 in what follows.

As the 'future annuity values', *e.g.* FutAnnVal(x,t), do not depend on the  $Z_t^{(l)}$ , the conditional expected value at the start of year t + 1 of all the liabilities, given the  $Z_t^{(l)}$ , is therefore NC(x), where:

$$\begin{split} & E[FutAnnVal(x,t)] \\ &+ \frac{\gamma_{t-x}\sigma^{(0)}(\alpha^{(0)})^{a-x+1}(e^{\lambda}-1)Z_{t}^{(0)}}{(1-\alpha^{(0)})^{(1-\alpha^{(0)}e^{\lambda})}} E[FutAnnVal\_Maxa(x,t)] \\ &+ \frac{\gamma_{t-x}\sigma^{(0)}Z_{t}^{(0)}}{1-\alpha^{(0)}} E[FutAnnVal\_Maxa\_alpha0(x,t)] \\ &- \gamma_{t-x}\sigma^{(0)}Z_{t}^{(0)} E[FutAnnVal(x,t)] \\ &- \gamma_{t-x}\sigma^{(0)}Z_{t}^{(0)} \left(\frac{\alpha^{(0)}(1-(\alpha^{(0)})^{a-x-1})}{1-\alpha^{(0)}} + \frac{(\alpha^{(0)})^{a-x}}{1-\alpha^{(0)}e^{\lambda}}\right) E[FutAnnVal(x,t)] \\ &+ \frac{\gamma_{t-x}\sigma^{(0)}\alpha^{(0)}e^{-(a-x-1)\lambda}Z_{t}^{(0)}}{1-\alpha^{(0)}e^{\lambda}} E[FutAnnVal\_Mina\_alpha0\_lambda(x,t)] \\ &- \gamma_{t-x}\sigma^{(3)}\overline{\mu}_{exp}(x,t)Z_{t}^{(3)}E[FutAnnVal\_Varying3(x,t)] \\ &- \gamma_{t-x}\sigma^{(3)}(1+\beta^{(3)})Z_{t}^{(3)}E[FutAnnVal\_Varying4(x,t)] \\ &- \gamma_{t-x}\sigma^{(5)}((x-\overline{x})^{2}-\hat{\sigma}_{x}^{2})\overline{\mu}_{exp}(x,t)Z_{t}^{(5)}E[FutAnnVal(x,t)] \\ &- \gamma_{t-x}\sigma^{(5)}Z_{t}^{(5)}E[FutAnnVal\_Varying5(x,t)] \end{split}$$

and all the expectations can be calculated in the same way as the mean annuity values in Section 5.5, except that some of the annuities have payments varying over time or are valued at revised interest rates, The 'annuitant' has just attained the age label x + 1 rather than x, and the starting kappa parameters have been rolled forward for one year on best estimate assumptions. Note that the random variable C(x) is a constant plus a linear combination of the  $Z_t^{(l)}$ .

It is instructive to consider how the different terms in the above expression for C(x) arise.  $\overline{p}_{xt}E[FutAnnVal(x,t)]$  is the expected value at the start of year t + 1 of all the liabilities conditional on all the  $Z_t^{(l)}$  being equal to zero. All the other terms containing E[FutAnnVal(x,t)], with the exception of

$$-\gamma_{t-x}\sigma^{(0)}Z_{t}^{(0)}\left(\frac{\alpha^{(0)}(1-(\alpha^{(0)})^{a-x-1})}{1-\alpha^{(0)}}+\frac{(\alpha^{(0)})^{a-x}}{1-\alpha^{(0)}e^{\lambda}}\right)E[FutAnnVal(x,t)], \text{ arise from the}$$

variation of the number of lives surviving year *t* with the  $Z_t^{(l)}$ . The remaining terms arise from the variation of the conditional mean annuity value at the end of year *t* with the  $Z_t^{(l)}$ . Thus for each kappa parameter, there is one term inside the square brackets arising from the variation of the number of lives surviving year *t* and one term arising from the variation of the conditional mean annuity value, with the exception that there are four different terms containing  $Z_t^{(0)}$  arising from the variation of the conditional mean annuity value. The sum of these four terms can alternatively be written as:

$$-\frac{\gamma_{t-x}\sigma^{(0)}\alpha^{(0)}Z_{t}^{(0)}}{1-\alpha^{(0)}}E[FutAnnVal\_Maxa(x,t)] \\ +\frac{\gamma_{t-x}\sigma^{(0)}\alpha^{(0)}Z_{t}^{(0)}}{1-\alpha^{(0)}}E[FutAnnVal\_Maxa\_alpha0(x,t)] \\ -\gamma_{t-x}\sigma^{(0)}Z_{t}^{(0)}\left(\frac{\alpha^{(0)}(1-(\alpha^{(0)})^{a-x-1})}{1-\alpha^{(0)}}+\frac{(\alpha^{(0)})^{a-x}}{1-\alpha^{(0)}e^{\lambda}}\right) \\ (E[FutAnnVal(x,t)]-E[FutAnnVal\_Maxa(x,t)]) \\ +\frac{\gamma_{t-x}\sigma^{(0)}\alpha^{(0)}e^{-(a-x-1)\lambda}Z_{t}^{(0)}}{1-\alpha^{(0)}e^{\lambda}}E[FutAnnVal\_Mina\_alpha0\_lambda(x,t)].$$

The first and third of these terms can be thought of as the impact of  $Z_t^{(0)}$  on the conditional mean value of the annuity payments up to and including age *a* and after age *a* respectively, ignoring the fact that not all of the impact of  $Z_t^{(0)}$  will have occurred by the time these annuity payments are made as the conditional mean of the AR(1) process for the  $\kappa^{(0)}$  parameters following the disturbance  $Z_t^{(0)}$  will not have reverted completely to  $\mu^{(0)}$  by the time the payments are made. The second and fourth terms, respectively, then make an adjustment for this to the impact on the conditional mean value of the annuity payments up to and including age *a* and after age *a* respectively.

For a portfolio of annuities on lives of different ages, the conditional expected value at the start of year t + 1 of all the liabilities is:

$$\sum_{x} N(x)C(x),$$

where N(x) is the number of annuitants at each age x. This conditional expected value is again a constant plus a linear combination of the  $Z_t^{(l)}$ . We should expect the portfolio to produce a lower ICA capital requirement than the sum of the capital requirements for the individual ages, as a result of diversification benefits arising from the fact that mortality changes at different ages are not perfectly correlated.

As we are assuming that interest rates are constant, and as the amount of annuity payments made in the first year is deterministic, being equal to the number of immediate annuitants, the value of the assets at the end of the first year is deterministic. Thus the ICA capital is simply the difference between the 99.5<sup>th</sup> percentile and the mean of  $\sum_{x} N(x)C(x)$ , as the  $Z_t^{(l)}$ 

vary, discounted for one year. To calculate this, we use the fact that  $\sum_{x} N(x)C(x)$  is

normally distributed, with the variance being derived from the parameter estimates of the time series models in Chapter 4, from the 'future annuity values' calculated in the same way as the

mean annuity values in Section 5.5, and from the correlations between the different  $Z_t^{(l)}$  in Section 4.7. The ICA capital is then given by:

$$v\Phi^{-1}(0.995)\sqrt{\mathrm{var}\left(\sum_{x}N(x)C(x)\right)},$$

where  $\Phi$  is the N(0,1) distribution function.

Only 50 scenarios, rather than 500, were used to calculate the 'future annuity values'. The possibility of adding further scenarios was investigated but it was found that the results for ICA capital changed only in the third decimal place, *i.e.* the last decimal place quoted in Table 5.18, as a result.

Table 5.18 shows the results for ICA capital considering only one age x in 2005 at a time, *i.e.* N(y) is set to zero for  $y \neq x$ . N(x) is set to 1 – this is for presentational purposes only and does not imply that it is sensible to use this method to calculate the ICA capital for a single annuity.

It is not apparent what relationship these ICA capital amounts should bear to the distributions of annuity values derived in Section 5.5. The former are based on distributions of mean annuity values after one year as the new information arising during the year varies, whereas the latter are ranges of outcomes over the full outstanding lifetime of the annuitants. A 'rule of thumb' for calculating ICA capital for longevity risk that is widely used within the UK life insurance industry is to take the 95<sup>th</sup> percentile of the distribution of annuity values derived in Section 5.5. If the discounted mean term of a portfolio of annuity business is approximately 10 years, then this might be considered reasonable on the grounds that a 99.5% confidence level over a one-year timeframe is approximately equivalent to a  $(100 - 0.5 \times 10) = 95\%$ confidence level over a 10-year timeframe. However, Table 5.18 shows that the ICA capital amounts we have calculated correspond to much lower percentiles than the 95<sup>th</sup> percentile of the distributions of annuity values, particularly for younger generations and for CMI data. An insurance company might construe this as supporting the use of a lower percentile than the 95<sup>th</sup> percentile in its ICA calculation. This may in some circumstances be reasonable, but the company should bear in mind the following before deciding to use a lower percentile than the 95<sup>th</sup> percentile:

- The most appropriate percentile to use will depend on the age distribution of the company's annuity portfolio, as the percentages in Table 5.18 vary significantly by age in 2005. The company should also consider how its annuity portfolio compares with the populations underlying the three data sets.
- The percentages in Table 5.18 make no allowance for parameter risk, *i.e.* the risk that the parameters of the model have been estimated incorrectly.
- The percentages in Table 5.18 make no allowance for model risk, *i.e.* the risk that the model itself is incorrect.

We might expect the percentages in Table 5.18 to be lower for groups of annuitants with a long future life expectancy, as this will tend to make the risk over one year relatively small compared with the risk over the whole future lifetime of the annuitants. This is indeed what we observe – the percentages increase as the age in 2005 increases, they increase as we move from CMI data to England and Wales female data, and in most cases they increase as we move from England and Wales female data to England and Wales male data.

Table 5.19 shows the coefficients of the different  $Z_{2006}^{(l)}$  in the expressions for C(x) for each age *x* and for each data set. Large absolute values in this table indicate that the ICA capital is, in absolute terms, highly sensitive to that 2006 innovation term. The following observations can be made from Table 5.19:

- All the coefficients of  $Z_{2006}^{(0)}$  and  $Z_{2006}^{(3)}$  are negative. This is as expected because an increase in these innovation terms is always associated with an increase in mortality and hence with a decrease in annuity values.
- All the coefficients of  $Z_{2006}^{(4)}$  are again negative. An increase in this innovation term is associated with a decrease in mortality at ages below the mean age of the data set (60 for CMI data and 59.5 for England and Wales data) and an increase at ages above the mean age of the data set. The negative coefficients show that the effect of ages above the mean age of the data set always outweighs the effect of ages below the mean age of the data set. This can be explained by the low mortality rates below the mean ages of all the data sets and by the fact that, as the  $\alpha^{(4)}$  parameters are relatively close to 1,  $Z_{2006}^{(4)}$  has a not insignificant impact on mortality at older ages for lives that are still below the mean age.
- The coefficients of  $Z_{2006}^{(5)}$  are positive for lower values of the age in 2005 but have become negative by the time the age in 2005 increases to 65 (75 for England and Wales male data). For CMI data, an increase in  $Z_{2006}^{(5)}$  leads to a decrease in mortality between ages 43 and 77 inclusive and to an increase in mortality outside this age range. For England and Wales data, an increase in  $Z_{2006}^{(5)}$  leads to a decrease in mortality between ages 43 and 76 inclusive and to an increase in mortality outside this age range. The positive coefficients of  $Z_{2006}^{(5)}$  at ages in 2005 up to and including 55 show that the age range 43-77 or 43-76 has the greatest impact. Once the age in 2005 has increased to 65, mortality above age 77 (76 for England and Wales data) has more of an impact. We find that this impact outweighs the impact of the age range 43-77 or 43-76, to give a negative coefficient of  $Z_{2006}^{(5)}$ , for CMI data and for England and Wales female data but not for England and Wales male data. This seems reasonable as the England and Wales male data set has the highest probability of death before age 77 for a life now aged 65. Once the age in 2005 has increased to 75, the coefficient of  $Z_{2006}^{(5)}$  is driven almost entirely by mortality above age 77 (76 for England and Wales data), so it is negative for all data sets.
- The coefficient of  $Z_{2006}^{(3)}$  has the largest magnitude except at ages 65 and 75 for CMI data and at age 75 for England and Wales female data. For lower values of the age *x* in 2005, the key influence on C(x) will be from ages relatively close to the mean age of the data set, and the  $\kappa^{(3)}$  parameters are the main driver of mortality in this age range. However, as the  $\kappa^{(3)}$  parameters are, in the long term, the only driver of mortality improvements over time in the model, it is not surprising that the coefficient of  $Z_{2006}^{(3)}$  continues to be significant for the higher values of the age in 2005.
- At ages 65 and 75 for CMI data and 75 for England and Wales female data, the coefficient of  $Z_{2006}^{(4)}$  has the largest magnitude. The  $\kappa^{(4)}$  parameters provide a first-order approximation to how quickly the logarithm of the force of mortality increases with age as the age rises above the mean age of the data set, so it would be expected that the coefficient of  $Z_{2006}^{(4)}$  would become increasingly significant in magnitude as the age in 2005 increases. Although the  $\kappa^{(4)}$  parameters also have a significant impact in percentage terms on the force of mortality at the younger ages of the data sets, there are relatively few deaths at those ages. The higher estimated value of  $\sigma^{(4)}$  for CMI data than for England and Wales female data, and for England and Wales female data to why the change in the coefficient with the highest magnitude from

 $Z_{2006}^{(3)}$  to  $Z_{2006}^{(4)}$  occurs at a younger age for CMI data than for England and Wales female data, and at a younger age for England and Wales female data than for England and Wales male data.

- The  $\kappa^{(0)}$  parameters have a relatively minor impact in most cases, with the coefficients of  $Z_{2006}^{(0)}$  being much smaller in magnitude than those of  $Z_{2006}^{(3)}$ , even for the lower values of the age in 2005.
- The  $\kappa^{(5)}$  parameters also have a relatively minor impact for ages in 2005 up to and including 65. However, for lives aged 75 in 2005, the coefficient of  $Z_{2006}^{(5)}$  is significant in magnitude compared with those of  $Z_{2006}^{(3)}$  and  $Z_{2006}^{(4)}$ .

Figures 5.25-5.27 provide a check on the ICA capital amounts in Table 5.18 and are also of interest in their own right. The figures show the projected future sizes of a fund set up in 2005 equal to the mean annuity value plus the ICA capital, for different values of the age x in 2005 and for each of the three data sets. As before, N(x) has been set to 1 for presentational purposes only and N(y) has been set to zero for  $y \neq x$ . The red curves show the  $(100 - p)^{\text{th}}$  percentiles of the distributions of the future fund size, where p% is the percentile specified in Table 5.18. It is the fact that the red curves reach a fund size close to zero at age 120 that provides the check on the ICA capital amounts. The following remarks can also be made on Figures 5.25-5.27:

- At ages under 65, the graphs show no stochastic variation as no annuity payments are being made. The fund simply earns interest at the deterministic rate of 4.5%.
- At the highest ages shown in Figures 5.25-5.27, the graphs are close to exponential curves representing interest being earned at the deterministic rate of 4.5%, as there are few annuity payments still being made at these ages. Thus positive fund sizes become more positive exponentially and negative fund sizes, representing insolvency, become more negative exponentially.
- The black solid curve in each graph, representing the median, has a turning point when the rate at which annuity payments are being made equals the rate at which interest is being earned on the fund. As a result of the improvement of mortality over time, the turning point occurs at higher ages for lower values of the age in 2005.

Table 5.20 calculates the diversification benefit for a portfolio with N(x) = 1 for x = 35, 45, 55, 65 and 75, and N(y) = 0 for all other ages y, by comparing the ICA capital for this portfolio with the sum of the ICA capital results for the individual ages in Table 5.18. We find that the diversification benefit is greatest for CMI data and least for England and Wales male data. Some understanding of this result can be gained by observing that for CMI data, Table 5.19 shows that the main risk factor at the younger ages,  $Z_{2006}^{(3)}$ , is different from the main risk

factor at the older ages,  $Z_{
m 2006}^{
m (4)}$  , whereas for England and Wales male data, Table 5.19 shows

that  $Z_{2006}^{(3)}$  is the main risk factor at all the ages. In addition, Tables 4.6-4.8 show that the correlations between the different risk factors are lowest for CMI data.

A greater understanding of the sources of the diversification benefits shown in Table 5.20 can be gained from Figures 5.28-5.30 and Tables 5.21-5.23. Figures 5.28, 5.29 and 5.30 are scatter diagrams showing the values of C(x) and C(y) for each pair of ages (x,y) in 2005,

under the same 500 sets of values of the  $Z_{
m 2006}^{(l)}$  that were used in calculating the annuity

values in Section 5.5, for CMI data, England and Wales male data and England and Wales female data respectively. Tables 5.21, 5.22 and 5.23 show the corresponding correlations between C(x) and C(y), for CMI data, England and Wales male data and England and Wales female data respectively. The correlations in Tables 5.21-5.23 were calculated analytically rather than being based on the data points in Figures 5.28-5.30.

The correlations in Tables 5.21-5.23 tend to be higher, and the points in Figures 5.28-5.30 tend to lie closer to a straight line of positive slope, when *x* and *y* are close together than when they are far apart. The correlations also tend to be higher and the points closer to a straight line of positive slope for England and Wales male data compared with England and Wales female data, and for England and Wales female data compared with CMI data. This suggests that pairs of ages *x* and *y* that are close together give rise to relatively little diversification benefit, and is consistent with the observation from Table 5.20 that the overall diversification benefit is greatest for CMI data and least for England and Wales male data. However, the relative magnitudes of C(x) and C(y) for different pairs of ages (x,y) also influence where the diversification benefits come from. As the annuity values increase with age in 2005, with the exception of a decrease from age 65 to 75, a large value of *x* and a large value of *y*, even though the correlation is lower in the latter case.

Tables 5.24-5.29 and Figures 5.31-5.36 present some investigations into the accuracy of the linear approximations we have used, considering only one age in 2005 at a time. Tables 5.24-5.26 and Figures 5.31-5.33 relate to Approximation 1, while Tables 5.27-5.29 and Figures 5.34-5.36 relate to Approximation 2. A comparable investigation was carried out in relation to Approximation 3, but the results are not shown here as the errors were found to be considerably smaller than in Approximations 1 and 2.

Each row of Tables 5.24-5.29 represents a particular set of values of the  $Z_{2006}^{(l)}$ . The first ten rows represent randomly generated sets of values. Rows 11, 12, 13 and 14 represent, respectively, the effect of setting  $Z_{2006}^{(0)}$ ,  $Z_{2006}^{(3)}$ ,  $Z_{2006}^{(4)}$  and  $Z_{2006}^{(5)}$  to 0.1 and the other  $Z_{2006}^{(l)}$  to zero. Thus the main purpose of Rows 11-14 is to check that the coefficients in Table 5.19 have been calculated correctly, as with these relatively small values of the  $Z_{2006}^{(l)}$ , it is unlikely that terms which are second or higher order in the  $Z_{2006}^{(l)}$  will be significant. Rows 1-10 then indicate the errors introduced by ignoring terms that are second or higher order in the  $Z_{2006}^{(l)}$ , and rather smaller margin of error as acceptable in Rows 11-14 than in Rows 1-10.

Column (1) of Tables 5.24-5.29 shows the value of the quantity we are approximating to on the basis that the  $Z_{2006}^{(l)}$  are all equal to zero. This is the 'deterministic' probability of surviving 2006,  $\overline{p}_{x,2006}$ , in Tables 5.24-5.26 and the 50-scenario estimate of the 'deterministic' mean annuity value at the end of 2006, E[FutAnnVal(x,2006)], in Tables 5.27-5.29. In Tables 5.24-5.26, Column (2) shows the linear approximations to the conditional probability of surviving 2006 given the values of the  $Z_{2006}^{(l)}$  that are used in the relevant row, while in Tables 5.27-5.29, this column shows the linear approximations to the conditional mean annuity value at the end of 2006 given the values of the  $Z_{2006}^{(l)}$  that are used in the relevant row. Column (3) shows the actual probability of surviving 2006 or the actual 50-scenario estimate of the mean annuity value at the end of 2006, as appropriate. Column (4) shows the differences between Columns (2) and (1), while Column (5) shows the differences between Columns (3) and (1). We should like the differences between Columns (2) and (3), or equivalently the differences between Columns (4) and (5), to be small. The size of these differences can be measured either as the absolute error in the approximation of Column (3) by Column (2), which is shown in Column (6), or as the percentage error, which is shown in Column (7).

Figures 5.31-5.36 provide a visual representation of the comparisons between Columns (4) and (5) of Tables 5.24-5.29, for Rows 1-10 only. We should like all the points in these figures to lie close to a straight line with zero intercept and unit slope, which is also shown in the figures.

Rows 11-14 in Tables 5.24-5.29 do not give any reason to doubt that the coefficients of the  $Z_{2006}^{(l)}$  have been calculated correctly – there is only one case where the percentage difference between Columns (4) and (5) is greater than 0.5%.

The points in Figures 5.34-5.36 all lie very close to the line with zero intercept and unit slope, providing evidence that the use of Approximation 2 is appropriate. In addition, all but two of the differences between Columns (4) and (5) of Tables 5.27-5.29, as reported in Column (6), are less than 0.001 in absolute value, with most of the differences being significantly less than this. These differences of less than 0.001 should have a probability of surviving 2006 applied to them and should then be discounted for one year to calculate the impact on ICA capital, so the impact on ICA capital is also less than 0.001 in absolute value.

Some of the graphs in Figures 5.31-5.33 reveal outliers. However, the outliers occur at low values of the age in 2005, for which the probability of death in 2006 is small, so it is not clear that there will be a significant impact on ICA capital. In fact, there are only two cases where the difference between Columns (4) and (5) of Tables 5.24-5.26 is greater than  $4.5 \times 10^{-5}$ , and these two cases both occur at age 75 rather than at the younger ages. To calculate the impact on ICA capital, the difference between Columns (4) and (5) needs to be multiplied by an annuity factor, which cannot exceed the value at 4.5% interest of a perpetuity of 1 *p.a.* payable annually in advance, and then discounted for one year. Thus, except in the two cases where the difference is greater than  $4.5 \times 10^{-5}$ , the absolute impact on ICA capital cannot exceed  $4.5 \times 10^{-5}$  multiplied by the value at 4.5% interest of a perpetuity of 1 *p.a.* payable annually in advance, and then discounted for one year. Thus, except in the two cases where the difference is greater than  $4.5 \times 10^{-5}$ , the absolute impact on ICA capital cannot exceed  $4.5 \times 10^{-5}$  multiplied by the value at 4.5% interest of a perpetuity of 1 *p.a.* payable annually in arrears, or  $4.5 \times 10^{-5} / 0.045 = 0.001$ .

#### 5.7 Conclusion

Prediction intervals have been constructed for the future values of the different time series of parameters for all three data sets. However, the uncertainty in the future mortality projected by our model is not in fact as great as the width of these prediction intervals might suggest, as a result of the correlations between the different kappa parameter series and because a typical scenario will have high parameter values in some future years but low parameter values in others.

To use the mortality projections given by our model to calculate annuity values, it was necessary to consider both extrapolation of the 2005 mortality curve to ages above the highest age in the data and future mortality improvements at ages above the highest age in the data. Two possible assumptions were considered for each of these to provide an indication of the range of possible outcomes.

Values of both immediate and deferred annuities on lives of different ages were presented at different percentiles, at a deterministic interest rate of 4.5%. The means and standard deviations of the distributions of annuity values were also presented. These annuity values were then compared with previously published projections, starting from the 2005 mortality curves fitted in this thesis. It was found that our model generally produced higher annuity values for England and Wales data for both males and females. For ages in 2005 below 65, the annuity values from our model increased relative to those from the previously published projections. Plots of projected forces of mortality and probabilities of survival were produced to give an understanding of the sources of these differences.

Risk-based capital calculations under the ICA regime were presented. It was found that our model produced significantly lower capital requirements than a particular 'rule of thumb' that is widely used within the UK life insurance industry. This might suggest that a company currently using the 'rule of thumb' can reasonably justify reducing the amount of ICA capital it holds in respect of longevity risk, but some issues for the company to consider before reducing this capital were highlighted. ICA capital calculations were carried out for portfolios of immediate and deferred annuities at different ages as well as for single ages, and it was found that the diversification benefit to ICA capital from holding a portfolio of annuities at different ages was greatest for CMI data and least for England and Wales male data. Some linear approximations were used in the calculation of the ICA capital amounts, and some test results

were presented to confirm that these approximations were sufficiently accurate for the purpose in hand.

| Table | Data set   | Improvement | Extrapolation | Number of |
|-------|------------|-------------|---------------|-----------|
|       |            | assumption  | method        | scenarios |
| 5.2   | CMI        | Low         | Log-linear    | 500       |
| 5.3   | CMI        | High        | Log-linear    | 500       |
| 5.4   | E&W Male   | Low         | Log-linear    | 500       |
| 5.5   | E&W Male   | High        | Log-linear    | 500       |
| 5.6   | E&W Female | Low         | Log-linear    | 500       |
| 5.7   | E&W Female | High        | Log-linear    | 500       |
| 5.8   | E&W Male   | High        | Log-linear    | 5,000     |
| 5.9   | CMI        | Low         | LifeMetrics   | 500       |
| 5.10  | CMI        | High        | LifeMetrics   | 500       |
| 5.11  | E&W Male   | Low         | LifeMetrics   | 500       |
| 5.12  | E&W Male   | High        | LifeMetrics   | 500       |
| 5.13  | E&W Female | Low         | LifeMetrics   | 500       |
| 5.14  | E&W Female | High        | LifeMetrics   | 500       |

### Table 5.1 – Key to Tables 5.2-5.14

**Table 5.2** – Means, standard deviations and key percentiles of the empirical distributions of deferred/immediate annuity values for *CMI data* under the *low improvement assumption* and the *log-linear extrapolation method* – level single life annuity of 1 *p.a.* payable annually in advance from age 65 and valued using 4.5% *p.a.* interest – *500 scenarios* 

| Quantity                      | Age 35 in  | Age 45 in  | Age 55 in  | Age 65 in   | Age 75 in   |
|-------------------------------|------------|------------|------------|-------------|-------------|
|                               | 2005       | 2005       | 2005       | 2005        | 2005        |
|                               | (Deferred) | (Deferred) | (Deferred) | (Immediate) | (Immediate) |
| Mean                          | 4.065      | 6.092      | 9.073      | 14.160      | 10.334      |
| St. dev.                      | 0.090      | 0.142      | 0.211      | 0.290       | 0.335       |
| St. dev. / Mean               | 2.2%       | 2.3%       | 2.3%       | 2.0%        | 3.2%        |
| 0.5 <sup>th</sup> percentile  | 3.847      | 5.750      | 8.487      | 13.401      | 9.453       |
| 1 <sup>st</sup> percentile    | 3.871      | 5.764      | 8.558      | 13.470      | 9.498       |
| 2.5 <sup>th</sup> percentile  | 3.895      | 5.811      | 8.625      | 13.509      | 9.659       |
| 5 <sup>th</sup> percentile    | 3.917      | 5.844      | 8.710      | 13.626      | 9.778       |
| 10 <sup>th</sup> percentile   | 3.943      | 5.893      | 8.814      | 13.798      | 9.887       |
| 20 <sup>th</sup> percentile   | 3.987      | 5.977      | 8.905      | 13.918      | 10.071      |
| 30 <sup>th</sup> percentile   | 4.019      | 6.027      | 8.961      | 14.014      | 10.159      |
| 40 <sup>th</sup> percentile   | 4.045      | 6.061      | 9.016      | 14.107      | 10.262      |
| 50 <sup>th</sup> percentile   | 4.067      | 6.092      | 9.072      | 14.181      | 10.347      |
| 60 <sup>th</sup> percentile   | 4.087      | 6.132      | 9.132      | 14.248      | 10.428      |
| 70 <sup>th</sup> percentile   | 4.112      | 6.169      | 9.196      | 14.323      | 10.508      |
| 80 <sup>th</sup> percentile   | 4.145      | 6.220      | 9.253      | 14.401      | 10.624      |
| 90 <sup>th</sup> percentile   | 4.178      | 6.276      | 9.346      | 14.520      | 10.763      |
| 95 <sup>th</sup> percentile   | 4.208      | 6.317      | 9.407      | 14.611      | 10.871      |
| 97.5 <sup>th</sup> percentile | 4.236      | 6.352      | 9.457      | 14.678      | 10.962      |
| 99 <sup>th</sup> percentile   | 4.265      | 6.391      | 9.529      | 14.778      | 11.068      |
| 99.5 <sup>th</sup> percentile | 4.278      | 6.421      | 9.563      | 14.830      | 11.131      |

**Table 5.3** – Means, standard deviations and key percentiles of the empirical distributions of deferred/immediate annuity values for *CMI data* under the *high improvement assumption* and the *log-linear extrapolation method* – level single life annuity of 1 *p.a.* payable annually in advance from age 65 and valued using 4.5% *p.a.* interest – *500 scenarios* 

| Quantity                      | Age 35 in          | Age 45 in          | Age 55 in          | Age 65 in   | Age 75 in           |
|-------------------------------|--------------------|--------------------|--------------------|-------------|---------------------|
| Quantity                      | 2005               | 2005               | 2005               | 2005        | 2005                |
|                               | 2005<br>(Deferred) | 2005<br>(Deferred) | 2005<br>(Deferred) | 2005        | 2005<br>(Immediate) |
|                               | (Deletted)         | (Deletted)         | (Delened)          | (immediate) | (immediate)         |
| Mean                          | 4.209              | 6.249              | 9.224              | 14.305      | 10.458              |
| St. dev.                      | 0.126              | 0.183              | 0.255              | 0.341       | 0.389               |
| St. dev. / Mean               | 3.0%               | 2.9%               | 2.8%               | 2.4%        | 3.7%                |
| 0.5 <sup>th</sup> percentile  | 3.901              | 5.801              | 8.566              | 13.457      | 9.499               |
| 1 <sup>st</sup> percentile    | 3.945              | 5.846              | 8.652              | 13.490      | 9.538               |
| 2.5 <sup>th</sup> percentile  | 3.972              | 5.884              | 8.683              | 13.590      | 9.698               |
| 5 <sup>th</sup> percentile    | 4.014              | 5.946              | 8.824              | 13.715      | 9.821               |
| 10 <sup>th</sup> percentile   | 4.046              | 6.011              | 8.913              | 13.868      | 9.945               |
| 20 <sup>th</sup> percentile   | 4.094              | 6.095              | 9.016              | 14.020      | 10.139              |
| 30 <sup>th</sup> percentile   | 4.140              | 6.159              | 9.093              | 14.139      | 10.245              |
| 40 <sup>th</sup> percentile   | 4.177              | 6.201              | 9.155              | 14.234      | 10.360              |
| 50 <sup>th</sup> percentile   | 4.205              | 6.247              | 9.216              | 14.328      | 10.471              |
| 60 <sup>th</sup> percentile   | 4.235              | 6.292              | 9.290              | 14.414      | 10.560              |
| 70 <sup>th</sup> percentile   | 4.273              | 6.343              | 9.359              | 14.493      | 10.645              |
| 80 <sup>th</sup> percentile   | 4.317              | 6.403              | 9.445              | 14.579      | 10.791              |
| 90 <sup>th</sup> percentile   | 4.371              | 6.491              | 9.545              | 14.727      | 10.940              |
| 95 <sup>th</sup> percentile   | 4.431              | 6.542              | 9.652              | 14.845      | 11.109              |
| 97.5 <sup>th</sup> percentile | 4.471              | 6.597              | 9.701              | 14.973      | 11.239              |
| 99 <sup>th</sup> percentile   | 4.492              | 6.681              | 9.794              | 15.061      | 11.320              |
| 99.5 <sup>th</sup> percentile | 4.514              | 6.715              | 9.843              | 15.168      | 11.413              |

Table 5.4 – Means, standard deviations and key percentiles of the empirical distributions of deferred/immediate annuity values for *E&W male data* under the *low improvement* assumption and the *log-linear extrapolation method* – level single life annuity of 1 *p.a.* payable annually in advance from age 65 and valued using 4.5% *p.a.* interest – *500 scenarios*

| Quantity                      | Age 35 in  | Age 45 in  | Age 55 in  | Age 65 in   | Age 75 in   |
|-------------------------------|------------|------------|------------|-------------|-------------|
|                               | 2005       | 2005       | 2005       | 2005        | 2005        |
|                               | (Deferred) | (Deferred) | (Deferred) | (Immediate) | (Immediate) |
| Mean                          | 3.329      | 5.008      | 7.624      | 12.546      | 8.803       |
| St. dev.                      | 0.106      | 0.144      | 0.186      | 0.216       | 0.214       |
| St. dev. / Mean               | 3.2%       | 2.9%       | 2.4%       | 1.7%        | 2.4%        |
| 0.5 <sup>th</sup> percentile  | 3.071      | 4.664      | 7.186      | 12.002      | 8.269       |
| 1 <sup>st</sup> percentile    | 3.091      | 4.684      | 7.225      | 12.059      | 8.285       |
| 2.5 <sup>th</sup> percentile  | 3.127      | 4.716      | 7.275      | 12.126      | 8.392       |
| 5 <sup>th</sup> percentile    | 3.155      | 4.778      | 7.317      | 12.185      | 8.446       |
| 10 <sup>th</sup> percentile   | 3.190      | 4.823      | 7.388      | 12.282      | 8.540       |
| 20 <sup>th</sup> percentile   | 3.238      | 4.891      | 7.466      | 12.349      | 8.633       |
| 30 <sup>th</sup> percentile   | 3.269      | 4.935      | 7.518      | 12.427      | 8.692       |
| 40 <sup>th</sup> percentile   | 3.296      | 4.969      | 7.568      | 12.493      | 8.742       |
| 50 <sup>th</sup> percentile   | 3.332      | 4.993      | 7.621      | 12.545      | 8.793       |
| 60 <sup>th</sup> percentile   | 3.357      | 5.037      | 7.670      | 12.597      | 8.847       |
| 70 <sup>th</sup> percentile   | 3.384      | 5.090      | 7.730      | 12.651      | 8.911       |
| 80 <sup>th</sup> percentile   | 3.418      | 5.132      | 7.778      | 12.731      | 8.991       |
| 90 <sup>th</sup> percentile   | 3.464      | 5.187      | 7.860      | 12.826      | 9.077       |
| 95 <sup>th</sup> percentile   | 3.514      | 5.247      | 7.937      | 12.882      | 9.140       |
| 97.5 <sup>th</sup> percentile | 3.542      | 5.296      | 7.987      | 12.975      | 9.240       |
| 99 <sup>th</sup> percentile   | 3.564      | 5.348      | 8.045      | 13.037      | 9.328       |
| 99.5 <sup>th</sup> percentile | 3.571      | 5.360      | 8.102      | 13.102      | 9.394       |

**Table 5.5** – Means, standard deviations and key percentiles of the empirical distributions of deferred/immediate annuity values for *E&W male data* under the *high improvement assumption* and the *log-linear extrapolation method* – level single life annuity of 1 *p.a.* payable annually in advance from age 65 and valued using 4.5% *p.a.* interest – *500 scenarios* 

| Quantity                      | Age 35 in  | Age 45 in  | Age 55 in  | Age 65 in   | Age 75 in   |
|-------------------------------|------------|------------|------------|-------------|-------------|
|                               | 2005       | 2005       | 2005       | 2005        | 2005        |
|                               | (Deferred) | (Deferred) | (Deferred) | (Immediate) | (Immediate) |
| Mean                          | 3.374      | 5.055      | 7.669      | 12.591      | 8.842       |
| St. dev.                      | 0.122      | 0.161      | 0.202      | 0.234       | 0.233       |
| St. dev. / Mean               | 3.6%       | 3.2%       | 2.6%       | 1.9%        | 2.6%        |
| 0.5 <sup>th</sup> percentile  | 3.083      | 4.680      | 7.199      | 12.022      | 8.275       |
| 1 <sup>st</sup> percentile    | 3.104      | 4.705      | 7.234      | 12.070      | 8.301       |
| 2.5 <sup>th</sup> percentile  | 3.149      | 4.742      | 7.295      | 12.148      | 8.406       |
| 5 <sup>th</sup> percentile    | 3.179      | 4.793      | 7.345      | 12.216      | 8.474       |
| 10 <sup>th</sup> percentile   | 3.218      | 4.857      | 7.414      | 12.311      | 8.553       |
| 20 <sup>th</sup> percentile   | 3.268      | 4.925      | 7.493      | 12.374      | 8.659       |
| 30 <sup>th</sup> percentile   | 3.302      | 4.970      | 7.554      | 12.471      | 8.722       |
| 40 <sup>th</sup> percentile   | 3.340      | 5.008      | 7.604      | 12.532      | 8.768       |
| 50 <sup>th</sup> percentile   | 3.377      | 5.044      | 7.659      | 12.592      | 8.827       |
| 60 <sup>th</sup> percentile   | 3.404      | 5.089      | 7.716      | 12.646      | 8.890       |
| 70 <sup>th</sup> percentile   | 3.435      | 5.142      | 7.778      | 12.699      | 8.955       |
| 80 <sup>th</sup> percentile   | 3.474      | 5.190      | 7.840      | 12.783      | 9.035       |
| 90 <sup>th</sup> percentile   | 3.527      | 5.263      | 7.924      | 12.904      | 9.142       |
| 95 <sup>th</sup> percentile   | 3.595      | 5.332      | 8.016      | 12.956      | 9.219       |
| 97.5 <sup>th</sup> percentile | 3.627      | 5.391      | 8.068      | 13.069      | 9.325       |
| 99 <sup>th</sup> percentile   | 3.660      | 5.441      | 8.150      | 13.188      | 9.430       |
| 99.5 <sup>th</sup> percentile | 3.678      | 5.468      | 8.200      | 13.219      | 9.494       |

**Table 5.6** – Means, standard deviations and key percentiles of the empirical distributions of deferred/immediate annuity values for *E&W* female data under the *low improvement* assumption and the *log-linear extrapolation method* – level single life annuity of 1 *p.a.* payable annually in advance from age 65 and valued using 4.5% *p.a.* interest – *500 scenarios* 

| Quantity                      | Age 35 in  | Age 45 in  | Age 55 in  | Age 65 in   | Age 75 in   |
|-------------------------------|------------|------------|------------|-------------|-------------|
|                               | 2005       | 2005       | 2005       | 2005        | 2005        |
|                               | (Deferred) | (Deferred) | (Deferred) | (Immediate) | (Immediate) |
| Mean                          | 3.539      | 5.442      | 8.348      | 13.620      | 9.772       |
| St. dev.                      | 0.115      | 0.153      | 0.198      | 0.236       | 0.239       |
| St. dev. / Mean               | 3.3%       | 2.8%       | 2.4%       | 1.7%        | 2.4%        |
| 0.5 <sup>th</sup> percentile  | 3.249      | 5.068      | 7.845      | 13.030      | 9.178       |
| 1 <sup>st</sup> percentile    | 3.273      | 5.075      | 7.886      | 13.077      | 9.210       |
| 2.5 <sup>th</sup> percentile  | 3.316      | 5.143      | 7.952      | 13.112      | 9.287       |
| 5 <sup>th</sup> percentile    | 3.349      | 5.192      | 8.014      | 13.239      | 9.370       |
| 10 <sup>th</sup> percentile   | 3.393      | 5.247      | 8.110      | 13.327      | 9.452       |
| 20 <sup>th</sup> percentile   | 3.442      | 5.321      | 8.183      | 13.429      | 9.578       |
| 30 <sup>th</sup> percentile   | 3.475      | 5.362      | 8.241      | 13.490      | 9.648       |
| 40 <sup>th</sup> percentile   | 3.509      | 5.399      | 8.286      | 13.561      | 9.709       |
| 50 <sup>th</sup> percentile   | 3.541      | 5.436      | 8.337      | 13.628      | 9.777       |
| 60 <sup>th</sup> percentile   | 3.571      | 5.474      | 8.396      | 13.695      | 9.838       |
| 70 <sup>th</sup> percentile   | 3.599      | 5.527      | 8.458      | 13.743      | 9.897       |
| 80 <sup>th</sup> percentile   | 3.636      | 5.574      | 8.520      | 13.820      | 9.982       |
| 90 <sup>th</sup> percentile   | 3.690      | 5.639      | 8.596      | 13.916      | 10.079      |
| 95 <sup>th</sup> percentile   | 3.736      | 5.691      | 8.671      | 14.014      | 10.145      |
| 97.5 <sup>th</sup> percentile | 3.762      | 5.748      | 8.744      | 14.057      | 10.232      |
| 99 <sup>th</sup> percentile   | 3.784      | 5.787      | 8.796      | 14.133      | 10.320      |
| 99.5 <sup>th</sup> percentile | 3.811      | 5.818      | 8.846      | 14.175      | 10.343      |

**Table 5.7** – Means, standard deviations and key percentiles of the empirical distributions of deferred/immediate annuity values for *E&W* female data under the high improvement assumption and the log-linear extrapolation method – level single life annuity of 1 p.a. payable annually in advance from age 65 and valued using 4.5% p.a. interest – 500 scenarios

| Quantity                      | Age 35 in  | Age 45 in  | Age 55 in  | Age 65 in   | Age 75 in   |
|-------------------------------|------------|------------|------------|-------------|-------------|
| -                             | 2005       | 2005       | 2005       | 2005        | 2005        |
|                               | (Deferred) | (Deferred) | (Deferred) | (Immediate) | (Immediate) |
| Mean                          | 3.566      | 5.473      | 8.379      | 13.656      | 9.805       |
| St. dev.                      | 0.128      | 0.167      | 0.213      | 0.253       | 0.257       |
| St. dev. / Mean               | 3.6%       | 3.1%       | 2.5%       | 1.9%        | 2.6%        |
| 0.5 <sup>th</sup> percentile  | 3.253      | 5.074      | 7.853      | 13.033      | 9.179       |
| 1 <sup>st</sup> percentile    | 3.278      | 5.086      | 7.898      | 13.090      | 9.205       |
| 2.5 <sup>th</sup> percentile  | 3.326      | 5.156      | 7.963      | 13.127      | 9.292       |
| 5 <sup>th</sup> percentile    | 3.359      | 5.207      | 8.037      | 13.248      | 9.382       |
| 10 <sup>th</sup> percentile   | 3.408      | 5.268      | 8.128      | 13.349      | 9.466       |
| 20 <sup>th</sup> percentile   | 3.457      | 5.339      | 8.207      | 13.446      | 9.598       |
| 30 <sup>th</sup> percentile   | 3.494      | 5.383      | 8.265      | 13.516      | 9.665       |
| 40 <sup>th</sup> percentile   | 3.532      | 5.426      | 8.313      | 13.593      | 9.732       |
| 50 <sup>th</sup> percentile   | 3.569      | 5.464      | 8.360      | 13.657      | 9.809       |
| 60 <sup>th</sup> percentile   | 3.599      | 5.508      | 8.429      | 13.730      | 9.877       |
| 70 <sup>th</sup> percentile   | 3.628      | 5.559      | 8.497      | 13.782      | 9.931       |
| 80 <sup>th</sup> percentile   | 3.670      | 5.616      | 8.571      | 13.870      | 10.022      |
| 90 <sup>th</sup> percentile   | 3.732      | 5.685      | 8.644      | 13.979      | 10.142      |
| 95 <sup>th</sup> percentile   | 3.783      | 5.749      | 8.722      | 14.089      | 10.212      |
| 97.5 <sup>th</sup> percentile | 3.827      | 5.807      | 8.815      | 14.139      | 10.305      |
| 99 <sup>th</sup> percentile   | 3.845      | 5.855      | 8.890      | 14.217      | 10.408      |
| 99.5 <sup>th</sup> percentile | 3.874      | 5.892      | 8.919      | 14.305      | 10.440      |

**Table 5.8** – Means, standard deviations and key percentiles of the empirical distributions of deferred/immediate annuity values for *E&W male data* under the *high improvement assumption* and the *log-linear extrapolation method* – level single life annuity of 1 *p.a.* payable annually in advance from age 65 and valued using 4.5% *p.a.* interest – *5,000 scenarios* 

| Quantity                      | Age 35 in  | Age 45 in  | Age 55 in  | Age 65 in   | Age 75 in   |
|-------------------------------|------------|------------|------------|-------------|-------------|
|                               | 2005       | 2005       | 2005       | 2005        | 2005        |
|                               | (Deferred) | (Deferred) | (Deferred) | (Immediate) | (Immediate) |
| Mean                          | 3.371      | 5.051      | 7.662      | 12.580      | 8.829       |
| St. dev.                      | 0.122      | 0.162      | 0.202      | 0.230       | 0.228       |
| St. dev. / Mean               | 3.6%       | 3.2%       | 2.6%       | 1.8%        | 2.6%        |
| 0.5 <sup>th</sup> percentile  | 3.051      | 4.642      | 7.138      | 11.988      | 8.263       |
| 1 <sup>st</sup> percentile    | 3.083      | 4.666      | 7.194      | 12.063      | 8.307       |
| 2.5 <sup>th</sup> percentile  | 3.129      | 4.729      | 7.265      | 12.147      | 8.401       |
| 5 <sup>th</sup> percentile    | 3.167      | 4.778      | 7.334      | 12.211      | 8.462       |
| 10 <sup>th</sup> percentile   | 3.211      | 4.841      | 7.406      | 12.289      | 8.542       |
| 20 <sup>th</sup> percentile   | 3.266      | 4.916      | 7.490      | 12.385      | 8.636       |
| 30 <sup>th</sup> percentile   | 3.309      | 4.968      | 7.556      | 12.456      | 8.707       |
| 40 <sup>th</sup> percentile   | 3.342      | 5.011      | 7.607      | 12.522      | 8.765       |
| 50 <sup>th</sup> percentile   | 3.373      | 5.049      | 7.660      | 12.576      | 8.821       |
| 60 <sup>th</sup> percentile   | 3.402      | 5.093      | 7.715      | 12.638      | 8.880       |
| 70 <sup>th</sup> percentile   | 3.437      | 5.138      | 7.768      | 12.698      | 8.945       |
| 80 <sup>th</sup> percentile   | 3.475      | 5.187      | 7.833      | 12.772      | 9.020       |
| 90 <sup>th</sup> percentile   | 3.527      | 5.260      | 7.923      | 12.880      | 9.127       |
| 95 <sup>th</sup> percentile   | 3.568      | 5.317      | 7.991      | 12.957      | 9.208       |
| 97.5 <sup>th</sup> percentile | 3.606      | 5.362      | 8.053      | 13.035      | 9.291       |
| 99 <sup>th</sup> percentile   | 3.650      | 5.426      | 8.131      | 13.132      | 9.407       |
| 99.5 <sup>th</sup> percentile | 3.677      | 5.466      | 8.172      | 13.192      | 9.449       |

**Table 5.9** – Means, standard deviations and key percentiles of the empirical distributions of deferred/immediate annuity values for *CMI data* under the *low improvement assumption* and the *LifeMetrics extrapolation method* – level single life annuity of 1 *p.a.* payable annually in advance from age 65 and valued using 4.5% *p.a.* interest – *500 scenarios* 

| Quantity                      | Age 35 in  | Age 45 in  | Age 55 in  | Age 65 in   | Age 75 in   |
|-------------------------------|------------|------------|------------|-------------|-------------|
| -                             | 2005       | 2005       | 2005       | 2005        | 2005        |
|                               | (Deferred) | (Deferred) | (Deferred) | (Immediate) | (Immediate) |
| Mean                          | 4.058      | 6.082      | 9.059      | 14.140      | 10.306      |
| St. dev.                      | 0.089      | 0.141      | 0.210      | 0.288       | 0.331       |
| St. dev. / Mean               | 2.2%       | 2.3%       | 2.3%       | 2.0%        | 3.2%        |
| 0.5 <sup>th</sup> percentile  | 3.841      | 5.741      | 8.477      | 13.387      | 9.433       |
| 1 <sup>st</sup> percentile    | 3.865      | 5.756      | 8.549      | 13.458      | 9.480       |
| 2.5 <sup>th</sup> percentile  | 3.889      | 5.802      | 8.613      | 13.493      | 9.638       |
| 5 <sup>th</sup> percentile    | 3.911      | 5.834      | 8.697      | 13.610      | 9.754       |
| 10 <sup>th</sup> percentile   | 3.936      | 5.884      | 8.803      | 13.781      | 9.863       |
| 20 <sup>th</sup> percentile   | 3.980      | 5.967      | 8.891      | 13.900      | 10.046      |
| 30 <sup>th</sup> percentile   | 4.011      | 6.017      | 8.948      | 13.994      | 10.132      |
| 40 <sup>th</sup> percentile   | 4.038      | 6.051      | 9.003      | 14.088      | 10.234      |
| 50 <sup>th</sup> percentile   | 4.060      | 6.081      | 9.058      | 14.160      | 10.319      |
| 60 <sup>th</sup> percentile   | 4.079      | 6.121      | 9.118      | 14.228      | 10.399      |
| 70 <sup>th</sup> percentile   | 4.105      | 6.158      | 9.180      | 14.302      | 10.478      |
| 80 <sup>th</sup> percentile   | 4.138      | 6.209      | 9.238      | 14.380      | 10.592      |
| 90 <sup>th</sup> percentile   | 4.171      | 6.265      | 9.330      | 14.497      | 10.731      |
| 95 <sup>th</sup> percentile   | 4.200      | 6.306      | 9.391      | 14.588      | 10.838      |
| 97.5 <sup>th</sup> percentile | 4.228      | 6.340      | 9.440      | 14.655      | 10.927      |
| 99 <sup>th</sup> percentile   | 4.257      | 6.379      | 9.513      | 14.754      | 11.033      |
| 99.5 <sup>th</sup> percentile | 4.270      | 6.409      | 9.546      | 14.806      | 11.095      |

**Table 5.10** – Means, standard deviations and key percentiles of the empirical distributions of deferred/immediate annuity values for *CMI data* under the *high improvement assumption* and the *LifeMetrics extrapolation method* – level single life annuity of 1 *p.a.* payable annually in advance from age 65 and valued using 4.5% *p.a.* interest – *500 scenarios* 

| Quantity                      | Ago 25 in  | Ago 45 in  | Ago 55 in  | Ago 65 in   | Ago 75 in   |
|-------------------------------|------------|------------|------------|-------------|-------------|
| Quantity                      | Age 55 m   | Age 45 III | Age 55 m   | Age 05 III  | Age 75 III  |
|                               | 2005       | 2005       | 2005       | 2005        | 2005        |
|                               | (Deferred) | (Deferred) | (Deferred) | (Immediate) | (Immediate) |
| Mean                          | 4.209      | 6.247      | 9.216      | 14.290      | 10.432      |
| St. dev.                      | 0.128      | 0.185      | 0.257      | 0.342       | 0.388       |
| St. dev. / Mean               | 3.0%       | 3.0%       | 2.8%       | 2.4%        | 3.7%        |
| 0.5 <sup>th</sup> percentile  | 3.896      | 5.793      | 8.556      | 13.443      | 9.479       |
| 1 <sup>st</sup> percentile    | 3.943      | 5.839      | 8.645      | 13.475      | 9.517       |
| 2.5 <sup>th</sup> percentile  | 3.970      | 5.877      | 8.671      | 13.575      | 9.676       |
| 5 <sup>th</sup> percentile    | 4.010      | 5.941      | 8.814      | 13.700      | 9.797       |
| 10 <sup>th</sup> percentile   | 4.044      | 6.006      | 8.903      | 13.852      | 9.921       |
| 20 <sup>th</sup> percentile   | 4.093      | 6.090      | 9.009      | 14.006      | 10.113      |
| 30 <sup>th</sup> percentile   | 4.140      | 6.155      | 9.085      | 14.122      | 10.220      |
| 40 <sup>th</sup> percentile   | 4.176      | 6.199      | 9.147      | 14.219      | 10.332      |
| 50 <sup>th</sup> percentile   | 4.205      | 6.244      | 9.210      | 14.311      | 10.444      |
| 60 <sup>th</sup> percentile   | 4.236      | 6.290      | 9.281      | 14.400      | 10.535      |
| 70 <sup>th</sup> percentile   | 4.274      | 6.342      | 9.352      | 14.481      | 10.618      |
| 80 <sup>th</sup> percentile   | 4.319      | 6.401      | 9.440      | 14.566      | 10.762      |
| 90 <sup>th</sup> percentile   | 4.372      | 6.491      | 9.540      | 14.713      | 10.914      |
| 95 <sup>th</sup> percentile   | 4.433      | 6.550      | 9.648      | 14.830      | 11.082      |
| 97.5 <sup>th</sup> percentile | 4.468      | 6.600      | 9.700      | 14.965      | 11.220      |
| 99 <sup>th</sup> percentile   | 4.496      | 6.686      | 9.787      | 15.060      | 11.295      |
| 99.5 <sup>th</sup> percentile | 4.516      | 6.720      | 9.848      | 15.159      | 11.394      |

Table 5.11 – Means, standard deviations and key percentiles of the empirical distributions of deferred/immediate annuity values for *E&W male data* under the *low improvement assumption* and the *LifeMetrics extrapolation method* – level single life annuity of 1 *p.a.* payable annually in advance from age 65 and valued using 4.5% *p.a.* interest – *500 scenarios* 

| Quantity                      | Age 35 in  | Age 45 in  | Age 55 in  | Age 65 in   | Age 75 in   |
|-------------------------------|------------|------------|------------|-------------|-------------|
|                               | 2005       | 2005       | 2005       | 2005        | 2005        |
|                               | (Deferred) | (Deferred) | (Deferred) | (Immediate) | (Immediate) |
| Mean                          | 3.334      | 5.015      | 7.632      | 12.556      | 8.815       |
| St. dev.                      | 0.107      | 0.146      | 0.187      | 0.219       | 0.217       |
| St. dev. / Mean               | 3.2%       | 2.9%       | 2.5%       | 1.7%        | 2.5%        |
| 0.5 <sup>th</sup> percentile  | 3.073      | 4.667      | 7.190      | 12.008      | 8.276       |
| 1 <sup>st</sup> percentile    | 3.094      | 4.688      | 7.228      | 12.065      | 8.292       |
| 2.5 <sup>th</sup> percentile  | 3.130      | 4.720      | 7.280      | 12.133      | 8.400       |
| 5 <sup>th</sup> percentile    | 3.158      | 4.781      | 7.323      | 12.192      | 8.454       |
| 10 <sup>th</sup> percentile   | 3.194      | 4.827      | 7.393      | 12.289      | 8.550       |
| 20 <sup>th</sup> percentile   | 3.243      | 4.899      | 7.472      | 12.357      | 8.642       |
| 30 <sup>th</sup> percentile   | 3.273      | 4.940      | 7.525      | 12.438      | 8.703       |
| 40 <sup>th</sup> percentile   | 3.302      | 4.975      | 7.575      | 12.503      | 8.753       |
| 50 <sup>th</sup> percentile   | 3.338      | 5.000      | 7.628      | 12.556      | 8.805       |
| 60 <sup>th</sup> percentile   | 3.363      | 5.044      | 7.680      | 12.608      | 8.861       |
| 70 <sup>th</sup> percentile   | 3.390      | 5.097      | 7.738      | 12.662      | 8.927       |
| 80 <sup>th</sup> percentile   | 3.424      | 5.140      | 7.788      | 12.743      | 9.006       |
| 90 <sup>th</sup> percentile   | 3.472      | 5.195      | 7.870      | 12.839      | 9.091       |
| 95 <sup>th</sup> percentile   | 3.521      | 5.259      | 7.949      | 12.895      | 9.156       |
| 97.5 <sup>th</sup> percentile | 3.550      | 5.305      | 7.997      | 12.990      | 9.259       |
| 99 <sup>th</sup> percentile   | 3.573      | 5.359      | 8.057      | 13.058      | 9.350       |
| 99.5 <sup>th</sup> percentile | 3.580      | 5.372      | 8.114      | 13.119      | 9.415       |

Table 5.12 – Means, standard deviations and key percentiles of the empirical distributions of deferred/immediate annuity values for *E&W male data* under the *high improvement assumption* and the *LifeMetrics extrapolation method* – level single life annuity of 1 *p.a.* payable annually in advance from age 65 and valued using 4.5% *p.a.* interest – *500 scenarios*

| Quantity                      | Age 35 in  | Age 45 in  | Age 55 in  | Age 65 in   | Age 75 in   |
|-------------------------------|------------|------------|------------|-------------|-------------|
|                               | 2005       | 2005       | 2005       | 2005        | 2005        |
|                               | (Deferred) | (Deferred) | (Deferred) | (Immediate) | (Immediate) |
| Mean                          | 3.389      | 5.071      | 7.684      | 12.609      | 8.860       |
| St. dev.                      | 0.127      | 0.166      | 0.207      | 0.240       | 0.239       |
| St. dev. / Mean               | 3.7%       | 3.3%       | 2.7%       | 1.9%        | 2.7%        |
| 0.5 <sup>th</sup> percentile  | 3.087      | 4.687      | 7.204      | 12.031      | 8.281       |
| 1 <sup>st</sup> percentile    | 3.109      | 4.711      | 7.239      | 12.077      | 8.312       |
| 2.5 <sup>th</sup> percentile  | 3.159      | 4.750      | 7.303      | 12.157      | 8.415       |
| 5 <sup>th</sup> percentile    | 3.190      | 4.801      | 7.354      | 12.224      | 8.482       |
| 10 <sup>th</sup> percentile   | 3.229      | 4.866      | 7.425      | 12.319      | 8.563       |
| 20 <sup>th</sup> percentile   | 3.279      | 4.935      | 7.505      | 12.391      | 8.674       |
| 30 <sup>th</sup> percentile   | 3.315      | 4.978      | 7.562      | 12.484      | 8.737       |
| 40 <sup>th</sup> percentile   | 3.352      | 5.022      | 7.621      | 12.546      | 8.786       |
| 50 <sup>th</sup> percentile   | 3.391      | 5.058      | 7.674      | 12.610      | 8.843       |
| 60 <sup>th</sup> percentile   | 3.418      | 5.105      | 7.733      | 12.663      | 8.910       |
| 70 <sup>th</sup> percentile   | 3.452      | 5.159      | 7.797      | 12.717      | 8.976       |
| 80 <sup>th</sup> percentile   | 3.492      | 5.206      | 7.861      | 12.800      | 9.057       |
| 90 <sup>th</sup> percentile   | 3.549      | 5.286      | 7.942      | 12.929      | 9.168       |
| 95 <sup>th</sup> percentile   | 3.620      | 5.360      | 8.036      | 12.987      | 9.250       |
| 97.5 <sup>th</sup> percentile | 3.651      | 5.419      | 8.095      | 13.101      | 9.358       |
| 99 <sup>th</sup> percentile   | 3.686      | 5.469      | 8.184      | 13.229      | 9.466       |
| 99.5 <sup>th</sup> percentile | 3.705      | 5.500      | 8.232      | 13.268      | 9.531       |

Table 5.13 – Means, standard deviations and key percentiles of the empirical distributions of deferred/immediate annuity values for *E&W* female data under the *low improvement* assumption and the *LifeMetrics extrapolation method* – level single life annuity of 1 *p.a.* payable annually in advance from age 65 and valued using 4.5% *p.a.* interest – *500 scenarios* 

| Quantity                      | Age 35 in  | Age 45 in  | Age 55 in  | Age 65 in   | Age 75 in   |
|-------------------------------|------------|------------|------------|-------------|-------------|
| -                             | 2005       | 2005       | 2005       | 2005        | 2005        |
|                               | (Deferred) | (Deferred) | (Deferred) | (Immediate) | (Immediate) |
| Mean                          | 3.547      | 5.452      | 8.360      | 13.638      | 9.794       |
| St. dev.                      | 0.117      | 0.155      | 0.201      | 0.240       | 0.244       |
| St. dev. / Mean               | 3.3%       | 2.9%       | 2.4%       | 1.8%        | 2.5%        |
| 0.5 <sup>th</sup> percentile  | 3.252      | 5.073      | 7.852      | 13.038      | 9.189       |
| 1 <sup>st</sup> percentile    | 3.277      | 5.079      | 7.894      | 13.089      | 9.220       |
| 2.5 <sup>th</sup> percentile  | 3.321      | 5.150      | 7.959      | 13.125      | 9.300       |
| 5 <sup>th</sup> percentile    | 3.354      | 5.199      | 8.024      | 13.250      | 9.386       |
| 10 <sup>th</sup> percentile   | 3.399      | 5.254      | 8.118      | 13.341      | 9.469       |
| 20 <sup>th</sup> percentile   | 3.448      | 5.330      | 8.194      | 13.444      | 9.594       |
| 30 <sup>th</sup> percentile   | 3.482      | 5.370      | 8.251      | 13.507      | 9.665       |
| 40 <sup>th</sup> percentile   | 3.516      | 5.409      | 8.298      | 13.578      | 9.730       |
| 50 <sup>th</sup> percentile   | 3.549      | 5.445      | 8.348      | 13.644      | 9.797       |
| 60 <sup>th</sup> percentile   | 3.579      | 5.485      | 8.408      | 13.714      | 9.864       |
| 70 <sup>th</sup> percentile   | 3.607      | 5.538      | 8.472      | 13.763      | 9.920       |
| 80 <sup>th</sup> percentile   | 3.645      | 5.586      | 8.534      | 13.844      | 10.008      |
| 90 <sup>th</sup> percentile   | 3.700      | 5.653      | 8.613      | 13.941      | 10.111      |
| 95 <sup>th</sup> percentile   | 3.748      | 5.706      | 8.687      | 14.043      | 10.178      |
| 97.5 <sup>th</sup> percentile | 3.774      | 5.763      | 8.762      | 14.085      | 10.264      |
| 99 <sup>th</sup> percentile   | 3.795      | 5.803      | 8.818      | 14.162      | 10.355      |
| 99.5 <sup>th</sup> percentile | 3.823      | 5.834      | 8.866      | 14.205      | 10.380      |

Table 5.14 – Means, standard deviations and key percentiles of the empirical distributions of deferred/immediate annuity values for *E&W female data* under the *high improvement assumption* and the *LifeMetrics extrapolation method* – level single life annuity of 1 *p.a.* payable annually in advance from age 65 and valued using 4.5% *p.a.* interest – *500 scenarios* 

| Quantity                      | Age 35 in  | Age 45 in  | Age 55 in  | Age 65 in   | Age 75 in   |
|-------------------------------|------------|------------|------------|-------------|-------------|
|                               | 2005       | 2005       | 2005       | 2005        | 2005        |
|                               | (Deferred) | (Deferred) | (Deferred) | (Immediate) | (Immediate) |
| Mean                          | 3.580      | 5.490      | 8.398      | 13.682      | 9.834       |
| St. dev.                      | 0.133      | 0.173      | 0.220      | 0.261       | 0.265       |
| St. dev. / Mean               | 3.7%       | 3.2%       | 2.6%       | 1.9%        | 2.7%        |
| 0.5 <sup>th</sup> percentile  | 3.257      | 5.078      | 7.861      | 13.042      | 9.191       |
| 1 <sup>st</sup> percentile    | 3.282      | 5.096      | 7.908      | 13.102      | 9.215       |
| 2.5 <sup>th</sup> percentile  | 3.333      | 5.164      | 7.970      | 13.143      | 9.306       |
| 5 <sup>th</sup> percentile    | 3.366      | 5.216      | 8.051      | 13.259      | 9.397       |
| 10 <sup>th</sup> percentile   | 3.416      | 5.279      | 8.137      | 13.369      | 9.486       |
| 20 <sup>th</sup> percentile   | 3.466      | 5.350      | 8.218      | 13.463      | 9.623       |
| 30 <sup>th</sup> percentile   | 3.504      | 5.395      | 8.279      | 13.539      | 9.689       |
| 40 <sup>th</sup> percentile   | 3.543      | 5.441      | 8.327      | 13.617      | 9.758       |
| 50 <sup>th</sup> percentile   | 3.583      | 5.480      | 8.378      | 13.683      | 9.838       |
| 60 <sup>th</sup> percentile   | 3.612      | 5.526      | 8.447      | 13.753      | 9.905       |
| 70 <sup>th</sup> percentile   | 3.643      | 5.579      | 8.523      | 13.812      | 9.962       |
| 80 <sup>th</sup> percentile   | 3.688      | 5.636      | 8.592      | 13.904      | 10.053      |
| 90 <sup>th</sup> percentile   | 3.753      | 5.710      | 8.671      | 14.012      | 10.178      |
| 95 <sup>th</sup> percentile   | 3.810      | 5.783      | 8.750      | 14.130      | 10.255      |
| 97.5 <sup>th</sup> percentile | 3.855      | 5.844      | 8.843      | 14.185      | 10.357      |
| 99 <sup>th</sup> percentile   | 3.875      | 5.888      | 8.929      | 14.263      | 10.466      |
| 99.5 <sup>th</sup> percentile | 3.902      | 5.926      | 8.955      | 14.375      | 10.493      |

Table 5.15 – Deferred/immediate annuity values based on previously published projectionsapplied to the 2005 CMI mortality curve fitted in this thesis, using the log-linear extrapolationmethod – level single life annuity of 1 *p.a.* payable annually in advance from age 65 andvalued using 4.5% *p.a.* interest

| Age in 2005    | Medium Cohort | Medium Cohort<br>subject to 1%<br>underpin | Long Cohort |
|----------------|---------------|--|-------------|
| 35 (Deferred)  | 3.717         | 3.820                                      | 3.836       |
| 45 (Deferred)  | 5.673         | 5.780                                      | 5.857       |
| 55 (Deferred)  | 8.608         | 8.706                                      | 8.892       |
| 65 (Immediate) | 13.723        | 13.808                                     | 14.182      |
| 75 (Immediate) | 10.130        | 10.189                                     | 10.642      |

Table 5.16 – Deferred/immediate annuity values based on GAD projections applied to the2005 England and Wales male mortality curve fitted in this thesis, using the log-linearextrapolation method – level single life annuity of 1 *p.a.* payable annually in advance from age65 and valued using 4.5% *p.a.* interest

| Age in 2005    | LLE    | Principal | HLE    |
|----------------|--------|-----------|--------|
| 35 (Deferred)  | 2.984  | 3.298     | 3.625  |
| 45 (Deferred)  | 4.705  | 5.034     | 5.384  |
| 55 (Deferred)  | 7.515  | 7.812     | 8.132  |
| 65 (Immediate) | 12.633 | 12.850    | 13.085 |
| 75 (Immediate) | 8.817  | 8.930     | 9.050  |

Table 5.17 – Deferred/immediate annuity values based on GAD projections applied to the2005 England and Wales female mortality curve fitted in this thesis, using the log-linearextrapolation method – level single life annuity of 1 *p.a.* payable annually in advance from age65 and valued using 4.5% *p.a.* interest

| Age in 2005    | LLE    | Principal | HLE    |
|----------------|--------|-----------|--------|
| 35 (Deferred)  | 3.429  | 3.672     | 3.924  |
| 45 (Deferred)  | 5.393  | 5.641     | 5.904  |
| 55 (Deferred)  | 8.444  | 8.661     | 8.893  |
| 65 (Immediate) | 13.873 | 14.026    | 14.191 |
| 75 (Immediate) | 9.820  | 9.894     | 9.971  |

**Table 5.18** – ICA capital calculations for deferred/immediate annuities and comparisons with the annuity value distributions obtained in Section 5.5, as shown in Table 5.3 for CMI data, Table 5.5 for England and Wales male data and Table 5.7 for England and Wales female data

| Data set   | Age in<br>2005 | Mean<br>annuity<br>value from<br>Section 5.5 | ICA capital | Mean<br>annuity<br>value +<br>ICA capital | Percentile<br>of<br>distribution<br>in Section<br>5.5 | 95 <sup>th</sup><br>percentile |
|------------|----------------|--|-------------|---|---|--------------------------------|
| CMI        | 35             | 4.209  | 0.047       | 4.256                                     | 64%   | 4.431                          |
| E&W Male   | 35             | 3.374  | 0.055       | 3.430                                     | 69%   | 3.595                          |
| E&W Female | 35             | 3.566  | 0.057       | 3.623                                     | 69%   | 3.783                          |
| CMI        | 45             | 6.249  | 0.075       | 6.323                                     | 67%   | 6.542                          |
| E&W Male   | 45             | 5.055  | 0.084       | 5.140                                     | 69%   | 5.332                          |
| E&W Female | 45             | 5.473  | 0.083       | 5.557                                     | 69%   | 5.749                          |
| CMI        | 55             | 9.224  | 0.125       | 9.349                                     | 69%   | 9.652                          |
| E&W Male   | 55             | 7.669  | 0.128       | 7.797                                     | 73%   | 8.016                          |
| E&W Female | 55             | 8.379  | 0.126       | 8.505                                     | 71%   | 8.722                          |
| CMI        | 65             | 14.305                                       | 0.218       | 14.523                                    | 73%   | 14.845                         |
| E&W Male   | 65             | 12.591                                       | 0.184       | 12.775                                    | 80%   | 12.956                         |
| E&W Female | 65             | 13.656                                       | 0.185       | 13.841                                    | 76%   | 14.089                         |
| CMI        | 75             | 10.458                                       | 0.336       | 10.794                                    | 80%   | 11.109                         |
| E&W Male   | 75             | 8.842  | 0.234       | 9.076                                     | 85%   | 9.219                          |
| E&W Female | 75             | 9.805  | 0.246       | 10.051                                    | 83%   | 10.212                         |

**Table 5.19** – Coefficients of the  $Z_{2006}^{(i)}$ , for *i* = 0,3,4,5, in the expressions for *C*(*x*) for each age *x* and each data set

| Data set   | Age in         | Coefficient of   | Coefficient of   | Coefficient of   | Coefficient of   |
|------------|----------------|------------------|------------------|------------------|------------------|
|            | 2005, <i>x</i> | $Z^{(0)}_{2006}$ | $Z^{(3)}_{2006}$ | $Z_{2006}^{(4)}$ | $Z_{2006}^{(5)}$ |
| CMI        | 35             | -0.00127         | -0.02316         | -0.00689         | 0.00097          |
| E&W Male   | 35             | -0.00280         | -0.02603         | -0.00282         | 0.00213          |
| E&W Female | 35             | -0.00490         | -0.02883         | -0.00427         | 0.00136          |
| CMI        | 45             | -0.00189         | -0.03861         | -0.01728         | 0.00377          |
| E&W Male   | 45             | -0.00381         | -0.04098         | -0.00768         | 0.00589          |
| E&W Female | 45             | -0.00664         | -0.04391         | -0.01196         | 0.00373          |
| CMI        | 55             | -0.00285         | -0.06245         | -0.04386         | 0.00924          |
| E&W Male   | 55             | -0.00478         | -0.06209         | -0.02050         | 0.01149          |
| E&W Female | 55             | -0.00849         | -0.06567         | -0.03047         | 0.00632          |
| CMI        | 65             | -0.00331         | -0.08950         | -0.10241         | -0.00180         |
| E&W Male   | 65             | -0.00422         | -0.08154         | -0.04728         | 0.00479          |
| E&W Female | 65             | -0.00744         | -0.08528         | -0.06634         | -0.00197         |
| CMI        | 75             | -0.00219         | -0.10451         | -0.18576         | -0.06768         |
| E&W Male   | 75             | -0.00233         | -0.08440         | -0.07882         | -0.03246         |
| E&W Female | 75             | -0.00433         | -0.09226         | -0.11122         | -0.03176         |

Table 5.20 – Diversification benefits from a portfolio of deferred/immediate annuities withequal numbers of annuitants aged 35, 45, 55, 65 and 75 in 2005, compared with each ageseparately

| Data set   | Undiversified ICA capital | Diversified ICA capital | Diversification benefit |
|------------|---------------------------|-------------------------|-------------------------|
| CMI        | 0.8016                    | 0.7526                  | 0.0490                  |
| E&W Male   | 0.6852                    | 0.6670                  | 0.0182                  |
| E&W Female | 0.6975                    | 0.6746                  | 0.0229                  |

## **Table 5.21** – Correlation coefficients between C(x) and C(y) for different ages x and y, for<br/>CMI data

| x/y | 35      | 45     | 55     | 65     | 75 |
|-----|---------|--------|--------|--------|----|
| 35  | 1       |        |        |        |    |
| 45  | 0.8056  | 1      |        |        |    |
| 55  | 0.4206  | 0.6667 | 1      |        |    |
| 65  | 0.0477  | 0.2615 | 0.6021 | 1      |    |
| 75  | -0.0482 | 0.0421 | 0.2237 | 0.5722 | 1  |

| Table 5.22 – Correlation coefficients between $C(x)$ and $C(y)$ for different ages x and y, for |
|---|
| England and Wales male data   |

| x/y | 35     | 45     | 55     | 65     | 75 |
|-----|--------|--------|--------|--------|----|
| 35  | 1      |        |        |        |    |
| 45  | 0.9012 | 1      |        |        |    |
| 55  | 0.7474 | 0.8620 | 1      |        |    |
| 65  | 0.6395 | 0.7398 | 0.8734 | 1      |    |
| 75  | 0.6867 | 0.7296 | 0.7905 | 0.8953 | 1  |

# **Table 5.23** – Correlation coefficients between C(x) and C(y) for different ages x and y, forEngland and Wales female data

| x/y | 35     | 45     | 55     | 65     | 75 |
|-----|--------|--------|--------|--------|----|
| 35  | 1      |        |        |        |    |
| 45  | 0.8557 | 1      |        |        |    |
| 55  | 0.6339 | 0.8016 | 1      |        |    |
| 65  | 0.4216 | 0.5751 | 0.7796 | 1      |    |
| 75  | 0.3505 | 0.4513 | 0.5999 | 0.8025 | 1  |

**Table 5.24** – Investigation of the accuracy of Approximation 1 for CMI data – age in 2005 (i) 35, (ii) 45, (iii) 55 – column (1) = 'deterministic' probability of surviving 2006, (2) = linear approximation to conditional probability of surviving 2006, (3) = actual conditional probability of surviving 2006, (4) = (2) – (1), (5) = (3) – (1), (6) = absolute error in approximation of (3) by (2), (7) = percentage error in approximation of (3) by (2)

|    | ٠ | • |
|----|---|---|
| 1  |   | ۱ |
|    | L |   |
| ۰. | ۰ | , |

| Test | (1)       | (2)       | (3)       | (4)        | (5)        | (6)       | (7)      |
|------|-----------|-----------|-----------|------------|------------|-----------|----------|
| 1    | 0.9994846 | 0.9994813 | 0.9994807 | -0.0000033 | -0.0000040 | 0.000007  | 0.00007% |
| 2    | 0.9994846 | 0.9994822 | 0.9994806 | -0.0000024 | -0.0000040 | 0.0000016 | 0.00016% |
| 3    | 0.9994846 | 0.9995019 | 0.9995019 | 0.0000173  | 0.0000173  | 0.0000000 | 0.00000% |
| 4    | 0.9994846 | 0.9994488 | 0.9994484 | -0.0000358 | -0.0000362 | 0.0000004 | 0.00004% |
| 5    | 0.9994846 | 0.9995180 | 0.9995179 | 0.0000333  | 0.0000332  | 0.0000001 | 0.00001% |
| 6    | 0.9994846 | 0.9993800 | 0.9993797 | -0.0001046 | -0.0001049 | 0.000003  | 0.00003% |
| 7    | 0.9994846 | 0.9994415 | 0.9994404 | -0.0000431 | -0.0000442 | 0.0000011 | 0.00011% |
| 8    | 0.9994846 | 0.9994569 | 0.9994568 | -0.0000278 | -0.0000278 | 0.0000000 | 0.00000% |
| 9    | 0.9994846 | 0.9996093 | 0.9996009 | 0.0001247  | 0.0001162  | 0.000085  | 0.00085% |
| 10   | 0.9994846 | 0.9994455 | 0.9994455 | -0.0000391 | -0.0000391 | 0.0000000 | 0.00000% |
| 11   | 0.9994846 | 0.9994770 | 0.9994770 | -0.0000076 | -0.0000076 | 0.0000000 | 0.00000% |
| 12   | 0.9994846 | 0.9994840 | 0.9994840 | -0.0000006 | -0.0000006 | 0.0000000 | 0.00000% |
| 13   | 0.9994846 | 0.9994856 | 0.9994856 | 0.0000010  | 0.0000009  | 0.0000000 | 0.00000% |
| 14   | 0.9994846 | 0.9994841 | 0.9994841 | -0.0000005 | -0.0000005 | 0.0000000 | 0.00000% |

(ii)

| Test | (1)       | (2)       | (3)       | (4)        | (5)        | (6)       | (7)      |
|------|-----------|-----------|-----------|------------|------------|-----------|----------|
| 1    | 0.9990114 | 0.9989917 | 0.9989910 | -0.0000197 | -0.0000204 | 0.0000007 | 0.00007% |
| 2    | 0.9990114 | 0.9989708 | 0.9989679 | -0.0000406 | -0.0000435 | 0.0000029 | 0.00029% |
| 3    | 0.9990114 | 0.9990130 | 0.9990129 | 0.0000016  | 0.0000014  | 0.0000002 | 0.00002% |
| 4    | 0.9990114 | 0.9989603 | 0.9989597 | -0.0000511 | -0.0000517 | 0.0000005 | 0.00005% |
| 5    | 0.9990114 | 0.9990417 | 0.9990417 | 0.0000303  | 0.0000303  | 0.0000000 | 0.00000% |
| 6    | 0.9990114 | 0.9988982 | 0.9988980 | -0.0001132 | -0.0001134 | 0.0000002 | 0.00002% |
| 7    | 0.9990114 | 0.9989439 | 0.9989424 | -0.0000675 | -0.0000690 | 0.0000015 | 0.00015% |
| 8    | 0.9990114 | 0.9990068 | 0.9990063 | -0.0000046 | -0.0000051 | 0.0000005 | 0.00005% |
| 9    | 0.9990114 | 0.9990779 | 0.9990686 | 0.0000665  | 0.0000572  | 0.0000094 | 0.00094% |
| 10   | 0.9990114 | 0.9989819 | 0.9989818 | -0.0000295 | -0.0000296 | 0.0000001 | 0.00001% |
| 11   | 0.9990114 | 0.9990038 | 0.9990038 | -0.0000076 | -0.0000076 | 0.0000000 | 0.00000% |
| 12   | 0.9990114 | 0.9990089 | 0.9990089 | -0.0000025 | -0.0000025 | 0.0000000 | 0.00000% |
| 13   | 0.9990114 | 0.9990139 | 0.9990139 | 0.0000025  | 0.0000025  | 0.0000000 | 0.00000% |
| 14   | 0.9990114 | 0.9990120 | 0.9990120 | 0.0000006  | 0.0000006  | 0.0000000 | 0.00000% |

(iii)

| Test | (1)       | (2)       | (3)       | (4)        | (5)        | (6)       | (7)      |
|------|-----------|-----------|-----------|------------|------------|-----------|----------|
| 1    | 0.9970823 | 0.9970386 | 0.9970381 | -0.0000436 | -0.0000442 | 0.0000006 | 0.00006% |
| 2    | 0.9970823 | 0.9969422 | 0.9969371 | -0.0001400 | -0.0001451 | 0.0000051 | 0.00051% |
| 3    | 0.9970823 | 0.9969787 | 0.9969758 | -0.0001036 | -0.0001064 | 0.0000029 | 0.00029% |
| 4    | 0.9970823 | 0.9970164 | 0.9970161 | -0.0000659 | -0.0000662 | 0.000003  | 0.00003% |
| 5    | 0.9970823 | 0.9970599 | 0.9970594 | -0.0000223 | -0.0000229 | 0.0000005 | 0.00005% |
| 6    | 0.9970823 | 0.9969642 | 0.9969641 | -0.0001181 | -0.0001181 | 0.0000001 | 0.00001% |
| 7    | 0.9970823 | 0.9969905 | 0.9969896 | -0.0000918 | -0.0000926 | 0.000008  | 0.00008% |
| 8    | 0.9970823 | 0.9971752 | 0.9971722 | 0.0000930  | 0.0000899  | 0.0000030 | 0.00030% |
| 9    | 0.9970823 | 0.9971156 | 0.9971114 | 0.0000333  | 0.0000292  | 0.0000041 | 0.00041% |
| 10   | 0.9970823 | 0.9970901 | 0.9970896 | 0.0000079  | 0.0000074  | 0.0000005 | 0.00005% |
| 11   | 0.9970823 | 0.9970745 | 0.9970745 | -0.0000078 | -0.0000078 | 0.0000000 | 0.00000% |
| 12   | 0.9970823 | 0.9970719 | 0.9970718 | -0.0000104 | -0.0000104 | 0.0000000 | 0.00000% |
| 13   | 0.9970823 | 0.9970857 | 0.9970857 | 0.0000035  | 0.0000034  | 0.0000000 | 0.00000% |
| 14   | 0.9970823 | 0.9970904 | 0.9970904 | 0.000082   | 0.0000082  | 0.0000000 | 0.00000% |

**Table 5.24** (continued) – Investigation of the accuracy of Approximation 1 for CMI data – age in 2005 (iv) 65, (v) 75 – column (1) = 'deterministic' probability of surviving 2006, (2) = linear approximation to conditional probability of surviving 2006, (3) = actual conditional probability of surviving 2006, (4) = (2) – (1), (5) = (3) – (1), (6) = absolute error in approximation of (3) by (2), (7) = percentage error in approximation of (3) by (2)

| 1  | •  |   | • |
|----|----|---|---|
| 1  | I) | 1 | ۱ |
| ١. | L  | v | 1 |

| Test | (1)       | (2)       | (3)       | (4)        | (5)        | (6)       | (7)      |
|------|-----------|-----------|-----------|------------|------------|-----------|----------|
| 1    | 0.9908489 | 0.9907333 | 0.9907324 | -0.0001156 | -0.0001165 | 0.0000009 | 0.00009% |
| 2    | 0.9908489 | 0.9904451 | 0.9904348 | -0.0004039 | -0.0004141 | 0.0000103 | 0.00104% |
| 3    | 0.9908489 | 0.9903276 | 0.9903109 | -0.0005213 | -0.0005380 | 0.0000167 | 0.00168% |
| 4    | 0.9908489 | 0.9908671 | 0.9908670 | 0.0000182  | 0.0000181  | 0.0000001 | 0.00001% |
| 5    | 0.9908489 | 0.9905800 | 0.9905750 | -0.0002689 | -0.0002740 | 0.0000051 | 0.00051% |
| 6    | 0.9908489 | 0.9907531 | 0.9907531 | -0.0000958 | -0.0000958 | 0.0000000 | 0.00000% |
| 7    | 0.9908489 | 0.9908468 | 0.9908468 | -0.0000021 | -0.0000022 | 0.0000000 | 0.00000% |
| 8    | 0.9908489 | 0.9911812 | 0.9911738 | 0.0003322  | 0.0003248  | 0.0000074 | 0.00074% |
| 9    | 0.9908489 | 0.9911045 | 0.9911041 | 0.0002555  | 0.0002551  | 0.0000004 | 0.00004% |
| 10   | 0.9908489 | 0.9909558 | 0.9909546 | 0.0001069  | 0.0001057  | 0.0000012 | 0.00012% |
| 11   | 0.9908489 | 0.9908415 | 0.9908415 | -0.0000075 | -0.0000075 | 0.0000000 | 0.00000% |
| 12   | 0.9908489 | 0.9908130 | 0.9908129 | -0.0000360 | -0.0000361 | 0.0000001 | 0.00001% |
| 13   | 0.9908489 | 0.9908370 | 0.9908370 | -0.0000119 | -0.0000120 | 0.0000000 | 0.00000% |
| 14   | 0.9908489 | 0.9908772 | 0.9908772 | 0.0000283  | 0.0000282  | 0.0000000 | 0.00000% |

(v)

| Test | (1)       | (2)       | (3)       | (4)        | (5)        | (6)       | (7)      |
|------|-----------|-----------|-----------|------------|------------|-----------|----------|
| 1    | 0.9713826 | 0.9706270 | 0.9706167 | -0.0007555 | -0.0007658 | 0.0000103 | 0.00106% |
| 2    | 0.9713826 | 0.9699484 | 0.9699110 | -0.0014342 | -0.0014716 | 0.0000374 | 0.00385% |
| 3    | 0.9713826 | 0.9695911 | 0.9695330 | -0.0017915 | -0.0018496 | 0.0000581 | 0.00599% |
| 4    | 0.9713826 | 0.9719137 | 0.9719083 | 0.0005311  | 0.0005257  | 0.0000054 | 0.00055% |
| 5    | 0.9713826 | 0.9703751 | 0.9703562 | -0.0010075 | -0.0010264 | 0.0000189 | 0.00195% |
| 6    | 0.9713826 | 0.9711205 | 0.9711200 | -0.0002621 | -0.0002625 | 0.0000005 | 0.00005% |
| 7    | 0.9713826 | 0.9718347 | 0.9718308 | 0.0004522  | 0.0004482  | 0.0000040 | 0.00041% |
| 8    | 0.9713826 | 0.9718817 | 0.9718769 | 0.0004992  | 0.0004943  | 0.0000049 | 0.00050% |
| 9    | 0.9713826 | 0.9723751 | 0.9723634 | 0.0009925  | 0.0009808  | 0.0000117 | 0.00120% |
| 10   | 0.9713826 | 0.9716112 | 0.9716099 | 0.0002286  | 0.0002274  | 0.0000013 | 0.00013% |
| 11   | 0.9713826 | 0.9713753 | 0.9713753 | -0.0000073 | -0.0000073 | 0.0000000 | 0.00000% |
| 12   | 0.9713826 | 0.9712679 | 0.9712676 | -0.0001147 | -0.0001149 | 0.0000002 | 0.00002% |
| 13   | 0.9713826 | 0.9712684 | 0.9712681 | -0.0001142 | -0.0001144 | 0.0000002 | 0.00002% |
| 14   | 0.9713826 | 0.9714095 | 0.9714094 | 0.0000269  | 0.0000269  | 0.0000000 | 0.00000% |

Table 5.25 – Investigation of the accuracy of Approximation 1 for England and Wales maledata – age in 2005 (i) 35, (ii) 45, (iii) 55 – column (1) = 'deterministic' probability of surviving2006, (2) = linear approximation to conditional probability of surviving 2006, (3) = actualconditional probability of surviving 2006, (4) = (2) – (1), (5) = (3) – (1), (6) = absolute error inapproximation of (3) by (2), (7) = percentage error in approximation of (3) by (2)

|    | ٠ | ۰. |
|----|---|----|
|    |   | ۱. |
|    | L |    |
| ۰. | ۰ | ,  |

| Test | (1)       | (2)       | (3)       | (4)        | (5)        | (6)       | (7)      |
|------|-----------|-----------|-----------|------------|------------|-----------|----------|
| 1    | 0.9989185 | 0.9989045 | 0.9989042 | -0.0000140 | -0.0000144 | 0.000003  | 0.00003% |
| 2    | 0.9989185 | 0.9989009 | 0.9989002 | -0.0000176 | -0.0000184 | 0.0000007 | 0.00007% |
| 3    | 0.9989185 | 0.9989362 | 0.9989361 | 0.0000176  | 0.0000176  | 0.0000000 | 0.00000% |
| 4    | 0.9989185 | 0.9988888 | 0.9988887 | -0.0000297 | -0.0000298 | 0.0000001 | 0.00001% |
| 5    | 0.9989185 | 0.9989448 | 0.9989448 | 0.0000263  | 0.0000262  | 0.0000000 | 0.00000% |
| 6    | 0.9989185 | 0.9988576 | 0.9988575 | -0.0000610 | -0.0000610 | 0.0000000 | 0.00000% |
| 7    | 0.9989185 | 0.9988778 | 0.9988774 | -0.0000408 | -0.0000412 | 0.0000004 | 0.00004% |
| 8    | 0.9989185 | 0.9989038 | 0.9989038 | -0.0000147 | -0.0000148 | 0.0000000 | 0.00000% |
| 9    | 0.9989185 | 0.9989290 | 0.9989190 | 0.0000105  | 0.0000005  | 0.0000100 | 0.00100% |
| 10   | 0.9989185 | 0.9988994 | 0.9988993 | -0.0000191 | -0.0000192 | 0.0000001 | 0.00001% |
| 11   | 0.9989185 | 0.9989131 | 0.9989131 | -0.0000054 | -0.0000054 | 0.0000000 | 0.00000% |
| 12   | 0.9989185 | 0.9989167 | 0.9989167 | -0.0000018 | -0.0000018 | 0.0000000 | 0.00000% |
| 13   | 0.9989185 | 0.9989205 | 0.9989205 | 0.0000020  | 0.0000020  | 0.0000000 | 0.00000% |
| 14   | 0.9989185 | 0.9989173 | 0.9989173 | -0.0000012 | -0.0000012 | 0.0000000 | 0.00000% |

(ii)

| Test | (1)       | (2)       | (3)       | (4)        | (5)        | (6)       | (7)      |
|------|-----------|-----------|-----------|------------|------------|-----------|----------|
| 1    | 0.9977385 | 0.9977071 | 0.9977066 | -0.0000315 | -0.0000319 | 0.0000004 | 0.00004% |
| 2    | 0.9977385 | 0.9976738 | 0.9976721 | -0.0000648 | -0.0000664 | 0.0000017 | 0.00017% |
| 3    | 0.9977385 | 0.9977301 | 0.9977300 | -0.0000084 | -0.0000085 | 0.0000001 | 0.00001% |
| 4    | 0.9977385 | 0.9976909 | 0.9976906 | -0.0000476 | -0.0000479 | 0.0000002 | 0.00002% |
| 5    | 0.9977385 | 0.9977569 | 0.9977569 | 0.0000184  | 0.0000184  | 0.0000000 | 0.00000% |
| 6    | 0.9977385 | 0.9976667 | 0.9976667 | -0.0000718 | -0.0000718 | 0.0000000 | 0.00000% |
| 7    | 0.9977385 | 0.9976700 | 0.9976693 | -0.0000685 | -0.0000692 | 0.0000007 | 0.00007% |
| 8    | 0.9977385 | 0.9977574 | 0.9977569 | 0.0000188  | 0.0000184  | 0.0000004 | 0.00004% |
| 9    | 0.9977385 | 0.9976922 | 0.9976843 | -0.0000463 | -0.0000542 | 0.0000079 | 0.00079% |
| 10   | 0.9977385 | 0.9977321 | 0.9977320 | -0.0000064 | -0.0000065 | 0.0000001 | 0.00001% |
| 11   | 0.9977385 | 0.9977331 | 0.9977331 | -0.0000054 | -0.0000054 | 0.0000000 | 0.00000% |
| 12   | 0.9977385 | 0.9977334 | 0.9977334 | -0.0000051 | -0.0000051 | 0.0000000 | 0.00000% |
| 13   | 0.9977385 | 0.9977417 | 0.9977417 | 0.0000032  | 0.0000032  | 0.0000000 | 0.00000% |
| 14   | 0.9977385 | 0.9977395 | 0.9977395 | 0.0000010  | 0.0000010  | 0.0000000 | 0.00000% |

(iii)

| Test | (1)       | (2)       | (3)       | (4)        | (5)        | (6)       | (7)      |
|------|-----------|-----------|-----------|------------|------------|-----------|----------|
| 1    | 0.9943192 | 0.9942376 | 0.9942369 | -0.0000816 | -0.0000824 | 0.0000008 | 0.00008% |
| 2    | 0.9943192 | 0.9941118 | 0.9941071 | -0.0002074 | -0.0002121 | 0.0000047 | 0.00047% |
| 3    | 0.9943192 | 0.9941936 | 0.9941918 | -0.0001256 | -0.0001274 | 0.0000018 | 0.00018% |
| 4    | 0.9943192 | 0.9942373 | 0.9942369 | -0.0000819 | -0.0000823 | 0.0000004 | 0.00004% |
| 5    | 0.9943192 | 0.9942916 | 0.9942914 | -0.0000276 | -0.0000279 | 0.0000002 | 0.00002% |
| 6    | 0.9943192 | 0.9941949 | 0.9941946 | -0.0001243 | -0.0001246 | 0.000003  | 0.00003% |
| 7    | 0.9943192 | 0.9941944 | 0.9941933 | -0.0001248 | -0.0001259 | 0.0000011 | 0.00011% |
| 8    | 0.9943192 | 0.9944371 | 0.9944353 | 0.0001179  | 0.0001161  | 0.0000018 | 0.00019% |
| 9    | 0.9943192 | 0.9942305 | 0.9942261 | -0.0000888 | -0.0000931 | 0.0000044 | 0.00044% |
| 10   | 0.9943192 | 0.9943429 | 0.9943426 | 0.0000237  | 0.0000234  | 0.000003  | 0.00003% |
| 11   | 0.9943192 | 0.9943137 | 0.9943137 | -0.0000055 | -0.0000055 | 0.0000000 | 0.00000% |
| 12   | 0.9943192 | 0.9943046 | 0.9943046 | -0.0000146 | -0.0000146 | 0.0000000 | 0.00000% |
| 13   | 0.9943192 | 0.9943221 | 0.9943221 | 0.0000028  | 0.0000028  | 0.0000000 | 0.00000% |
| 14   | 0.9943192 | 0.9943282 | 0.9943282 | 0.0000090  | 0.0000090  | 0.0000000 | 0.00000% |

**Table 5.25** (continued) – Investigation of the accuracy of Approximation 1 for England and Wales male data – age in 2005 (iv) 65, (v) 75 – column (1) = 'deterministic' probability of surviving 2006, (2) = linear approximation to conditional probability of surviving 2006, (3) = actual conditional probability of surviving 2006, (4) = (2) – (1), (5) = (3) – (1), (6) = absolute error in approximation of (3) by (2), (7) = percentage error in approximation of (3) by (2)

| Test | (1)       | (2)       | (3)       | (4)        | (5)        | (6)       | (7)      |
|------|-----------|-----------|-----------|------------|------------|-----------|----------|
| 1    | 0.9854343 | 0.9851603 | 0.9851575 | -0.0002739 | -0.0002767 | 0.0000028 | 0.00028% |
| 2    | 0.9854343 | 0.9848010 | 0.9847861 | -0.0006333 | -0.0006481 | 0.0000148 | 0.00151% |
| 3    | 0.9854343 | 0.9849282 | 0.9849187 | -0.0005061 | -0.0005156 | 0.0000095 | 0.00096% |
| 4    | 0.9854343 | 0.9853371 | 0.9853369 | -0.0000971 | -0.0000974 | 0.0000002 | 0.00002% |
| 5    | 0.9854343 | 0.9852346 | 0.9852329 | -0.0001996 | -0.0002013 | 0.0000017 | 0.00017% |
| 6    | 0.9854343 | 0.9851685 | 0.9851671 | -0.0002658 | -0.0002672 | 0.0000014 | 0.00014% |
| 7    | 0.9854343 | 0.9852360 | 0.9852348 | -0.0001983 | -0.0001994 | 0.0000011 | 0.00012% |
| 8    | 0.9854343 | 0.9857621 | 0.9857579 | 0.0003279  | 0.0003236  | 0.0000042 | 0.00043% |
| 9    | 0.9854343 | 0.9853184 | 0.9853165 | -0.0001158 | -0.0001178 | 0.0000019 | 0.00020% |
| 10   | 0.9854343 | 0.9855286 | 0.9855281 | 0.0000944  | 0.0000938  | 0.0000005 | 0.00005% |
| 11   | 0.9854343 | 0.9854290 | 0.9854290 | -0.0000052 | -0.0000052 | 0.0000000 | 0.00000% |
| 12   | 0.9854343 | 0.9853952 | 0.9853951 | -0.0000391 | -0.0000391 | 0.0000001 | 0.00001% |
| 13   | 0.9854343 | 0.9854250 | 0.9854250 | -0.000093  | -0.000093  | 0.0000000 | 0.00000% |
| 14   | 0.9854343 | 0.9854574 | 0.9854574 | 0.0000232  | 0.0000232  | 0.0000000 | 0.00000% |

(v)

(iv)

| Test | (1)       | (2)       | (3)       | (4)        | (5)        | (6)       | (7)      |
|------|-----------|-----------|-----------|------------|------------|-----------|----------|
| 1    | 0.9592526 | 0.9580728 | 0.9580556 | -0.0011797 | -0.0011970 | 0.0000173 | 0.00180% |
| 2    | 0.9592526 | 0.9571410 | 0.9570853 | -0.0021116 | -0.0021673 | 0.0000557 | 0.00582% |
| 3    | 0.9592526 | 0.9575823 | 0.9575475 | -0.0016703 | -0.0017050 | 0.0000348 | 0.00363% |
| 4    | 0.9592526 | 0.9593683 | 0.9593681 | 0.0001157  | 0.0001155  | 0.0000002 | 0.00002% |
| 5    | 0.9592526 | 0.9584759 | 0.9584681 | -0.0007767 | -0.0007845 | 0.0000078 | 0.00081% |
| 6    | 0.9592526 | 0.9586729 | 0.9586698 | -0.0005797 | -0.0005828 | 0.0000032 | 0.00033% |
| 7    | 0.9592526 | 0.9590387 | 0.9590382 | -0.0002139 | -0.0002144 | 0.0000005 | 0.00005% |
| 8    | 0.9592526 | 0.9598799 | 0.9598748 | 0.0006273  | 0.0006223  | 0.0000051 | 0.00053% |
| 9    | 0.9592526 | 0.9585970 | 0.9585896 | -0.0006556 | -0.0006630 | 0.0000074 | 0.00077% |
| 10   | 0.9592526 | 0.9595428 | 0.9595416 | 0.0002902  | 0.0002890  | 0.0000012 | 0.00013% |
| 11   | 0.9592526 | 0.9592474 | 0.9592474 | -0.0000052 | -0.0000052 | 0.0000000 | 0.00000% |
| 12   | 0.9592526 | 0.9591427 | 0.9591426 | -0.0001099 | -0.0001100 | 0.0000001 | 0.00002% |
| 13   | 0.9592526 | 0.9591790 | 0.9591789 | -0.0000736 | -0.0000737 | 0.0000001 | 0.00001% |
| 14   | 0.9592526 | 0.9592670 | 0.9592670 | 0.0000144  | 0.0000144  | 0.0000000 | 0.00000% |

**Table 5.26 –** Investigation of the accuracy of Approximation 1 for England and Wales femaledata – age in 2005 (i) 35, (ii) 45, (iii) 55 – column (1) = 'deterministic' probability of surviving2006, (2) = linear approximation to conditional probability of surviving 2006, (3) = actualconditional probability of surviving 2006, (4) = (2) – (1), (5) = (3) – (1), (6) = absolute error inapproximation of (3) by (2), (7) = percentage error in approximation of (3) by (2)

|    | ٠ | ۰. |
|----|---|----|
|    |   | ۱. |
|    | L |    |
| ۰. | ۰ | ,  |

| Test | (1)       | (2)       | (3)       | (4)        | (5)        | (6)       | (7)      |
|------|-----------|-----------|-----------|------------|------------|-----------|----------|
| 1    | 0.9994206 | 0.9994049 | 0.9994046 | -0.0000157 | -0.0000160 | 0.0000004 | 0.00004% |
| 2    | 0.9994206 | 0.9994011 | 0.9994004 | -0.0000195 | -0.0000203 | 0.000008  | 0.00008% |
| 3    | 0.9994206 | 0.9994415 | 0.9994415 | 0.0000209  | 0.0000208  | 0.0000000 | 0.00000% |
| 4    | 0.9994206 | 0.9993899 | 0.9993899 | -0.0000307 | -0.0000307 | 0.0000000 | 0.00000% |
| 5    | 0.9994206 | 0.9994481 | 0.9994481 | 0.0000275  | 0.0000275  | 0.0000000 | 0.00000% |
| 6    | 0.9994206 | 0.9993654 | 0.9993642 | -0.0000552 | -0.0000565 | 0.0000012 | 0.00012% |
| 7    | 0.9994206 | 0.9993778 | 0.9993776 | -0.0000429 | -0.0000430 | 0.0000001 | 0.00001% |
| 8    | 0.9994206 | 0.9994071 | 0.9994069 | -0.0000136 | -0.0000138 | 0.0000002 | 0.00002% |
| 9    | 0.9994206 | 0.9994087 | 0.9993872 | -0.0000119 | -0.0000334 | 0.0000214 | 0.00215% |
| 10   | 0.9994206 | 0.9994041 | 0.9994037 | -0.0000165 | -0.0000169 | 0.0000004 | 0.00004% |
| 11   | 0.9994206 | 0.9994123 | 0.9994123 | -0.0000083 | -0.0000083 | 0.0000000 | 0.00000% |
| 12   | 0.9994206 | 0.9994178 | 0.9994178 | -0.0000028 | -0.0000028 | 0.0000000 | 0.00000% |
| 13   | 0.9994206 | 0.9994249 | 0.9994249 | 0.0000043  | 0.0000043  | 0.0000000 | 0.00000% |
| 14   | 0.9994206 | 0.9994190 | 0.9994190 | -0.0000016 | -0.0000016 | 0.0000000 | 0.00000% |

(ii)

| Test | (1)       | (2)       | (3)       | (4)        | (5)        | (6)       | (7)      |
|------|-----------|-----------|-----------|------------|------------|-----------|----------|
| 1    | 0.9985563 | 0.9985340 | 0.9985337 | -0.0000223 | -0.0000226 | 0.0000003 | 0.00003% |
| 2    | 0.9985563 | 0.9985094 | 0.9985082 | -0.0000469 | -0.0000481 | 0.0000012 | 0.00012% |
| 3    | 0.9985563 | 0.9985548 | 0.9985547 | -0.0000015 | -0.0000016 | 0.0000001 | 0.00001% |
| 4    | 0.9985563 | 0.9985154 | 0.9985154 | -0.0000409 | -0.0000409 | 0.0000000 | 0.00000% |
| 5    | 0.9985563 | 0.9985749 | 0.9985749 | 0.0000187  | 0.0000186  | 0.0000000 | 0.00000% |
| 6    | 0.9985563 | 0.9984948 | 0.9984944 | -0.0000614 | -0.0000619 | 0.0000005 | 0.00005% |
| 7    | 0.9985563 | 0.9984991 | 0.9984988 | -0.0000572 | -0.0000574 | 0.0000002 | 0.00002% |
| 8    | 0.9985563 | 0.9985689 | 0.9985684 | 0.0000127  | 0.0000121  | 0.0000006 | 0.00006% |
| 9    | 0.9985563 | 0.9985260 | 0.9985131 | -0.0000303 | -0.0000431 | 0.0000128 | 0.00128% |
| 10   | 0.9985563 | 0.9985486 | 0.9985483 | -0.0000077 | -0.0000080 | 0.0000004 | 0.00004% |
| 11   | 0.9985563 | 0.9985482 | 0.9985482 | -0.0000081 | -0.0000081 | 0.0000000 | 0.00000% |
| 12   | 0.9985563 | 0.9985510 | 0.9985510 | -0.0000053 | -0.0000053 | 0.0000000 | 0.00000% |
| 13   | 0.9985563 | 0.9985610 | 0.9985610 | 0.0000047  | 0.0000047  | 0.0000000 | 0.00000% |
| 14   | 0.9985563 | 0.9985571 | 0.9985571 | 0.0000009  | 0.0000009  | 0.0000000 | 0.00000% |

(iii)

| Test | (1)       | (2)       | (3)       | (4)        | (5)        | (6)       | (7)      |
|------|-----------|-----------|-----------|------------|------------|-----------|----------|
| 1    | 0.9962834 | 0.9962303 | 0.9962298 | -0.0000530 | -0.0000535 | 0.0000005 | 0.00005% |
| 2    | 0.9962834 | 0.9961500 | 0.9961471 | -0.0001334 | -0.0001363 | 0.0000029 | 0.00029% |
| 3    | 0.9962834 | 0.9962115 | 0.9962105 | -0.0000719 | -0.0000729 | 0.0000010 | 0.00010% |
| 4    | 0.9962834 | 0.9962185 | 0.9962183 | -0.0000649 | -0.0000651 | 0.0000002 | 0.00002% |
| 5    | 0.9962834 | 0.9962773 | 0.9962771 | -0.0000061 | -0.0000063 | 0.0000002 | 0.00002% |
| 6    | 0.9962834 | 0.9961764 | 0.9961764 | -0.0001070 | -0.0001070 | 0.0000000 | 0.00000% |
| 7    | 0.9962834 | 0.9961883 | 0.9961877 | -0.0000951 | -0.0000957 | 0.0000005 | 0.00005% |
| 8    | 0.9962834 | 0.9963521 | 0.9963508 | 0.0000687  | 0.0000674  | 0.0000013 | 0.00013% |
| 9    | 0.9962834 | 0.9962487 | 0.9962425 | -0.0000347 | -0.0000409 | 0.0000062 | 0.00062% |
| 10   | 0.9962834 | 0.9962886 | 0.9962883 | 0.0000052  | 0.0000049  | 0.000003  | 0.00003% |
| 11   | 0.9962834 | 0.9962750 | 0.9962750 | -0.000084  | -0.0000084 | 0.0000000 | 0.00000% |
| 12   | 0.9962834 | 0.9962715 | 0.9962715 | -0.0000119 | -0.0000119 | 0.0000000 | 0.00000% |
| 13   | 0.9962834 | 0.9962867 | 0.9962867 | 0.000033   | 0.0000033  | 0.0000000 | 0.00000% |
| 14   | 0.9962834 | 0.9962895 | 0.9962895 | 0.0000062  | 0.0000062  | 0.0000000 | 0.00000% |

**Table 5.26** (continued) – Investigation of the accuracy of Approximation 1 for England and Wales female data – age in 2005 (iv) 65, (v) 75 – column (1) = 'deterministic' probability of surviving 2006, (2) = linear approximation to conditional probability of surviving 2006, (3) = actual conditional probability of surviving 2006, (4) = (2) – (1), (5) = (3) – (1), (6) = absolute error in approximation of (3) by (2), (7) = percentage error in approximation of (3) by (2)

| Test | (1)       | (2)       | (3)       | (4)        | (5)        | (6)       | (7)      |
|------|-----------|-----------|-----------|------------|------------|-----------|----------|
| 1    | 0.9908891 | 0.9907113 | 0.9907095 | -0.0001778 | -0.0001797 | 0.0000019 | 0.00019% |
| 2    | 0.9908891 | 0.9904959 | 0.9904869 | -0.0003932 | -0.0004022 | 0.0000090 | 0.00091% |
| 3    | 0.9908891 | 0.9906073 | 0.9906026 | -0.0002818 | -0.0002865 | 0.0000047 | 0.00047% |
| 4    | 0.9908891 | 0.9907928 | 0.9907925 | -0.0000963 | -0.0000966 | 0.000003  | 0.00003% |
| 5    | 0.9908891 | 0.9908061 | 0.9908055 | -0.0000830 | -0.0000837 | 0.0000007 | 0.00007% |
| 6    | 0.9908891 | 0.9906266 | 0.9906253 | -0.0002625 | -0.0002639 | 0.0000014 | 0.00014% |
| 7    | 0.9908891 | 0.9907205 | 0.9907194 | -0.0001686 | -0.0001697 | 0.0000011 | 0.00011% |
| 8    | 0.9908891 | 0.9910552 | 0.9910531 | 0.0001660  | 0.0001640  | 0.0000020 | 0.00021% |
| 9    | 0.9908891 | 0.9909036 | 0.9909022 | 0.0000145  | 0.0000131  | 0.0000014 | 0.00014% |
| 10   | 0.9908891 | 0.9909085 | 0.9909083 | 0.0000194  | 0.0000192  | 0.0000002 | 0.00002% |
| 11   | 0.9908891 | 0.9908813 | 0.9908813 | -0.0000078 | -0.0000078 | 0.0000000 | 0.00000% |
| 12   | 0.9908891 | 0.9908620 | 0.9908619 | -0.0000272 | -0.0000272 | 0.0000000 | 0.00000% |
| 13   | 0.9908891 | 0.9908800 | 0.9908800 | -0.0000092 | -0.0000092 | 0.0000000 | 0.00000% |
| 14   | 0.9908891 | 0.9909027 | 0.9909027 | 0.0000136  | 0.0000136  | 0.0000000 | 0.00000% |

(v)

(iv)

| Test | (1)       | (2)       | (3)       | (4)        | (5)        | (6)       | (7)      |
|------|-----------|-----------|-----------|------------|------------|-----------|----------|
| 1    | 0.9730089 | 0.9721838 | 0.9721711 | -0.0008252 | -0.0008379 | 0.0000127 | 0.00131% |
| 2    | 0.9730089 | 0.9715429 | 0.9715025 | -0.0014660 | -0.0015064 | 0.0000404 | 0.00416% |
| 3    | 0.9730089 | 0.9719583 | 0.9719375 | -0.0010506 | -0.0010714 | 0.0000208 | 0.00214% |
| 4    | 0.9730089 | 0.9729045 | 0.9729044 | -0.0001044 | -0.0001046 | 0.0000001 | 0.00001% |
| 5    | 0.9730089 | 0.9726424 | 0.9726396 | -0.0003665 | -0.0003694 | 0.0000029 | 0.00029% |
| 6    | 0.9730089 | 0.9721282 | 0.9721173 | -0.0008807 | -0.0008917 | 0.0000109 | 0.00112% |
| 7    | 0.9730089 | 0.9726394 | 0.9726373 | -0.0003696 | -0.0003717 | 0.0000021 | 0.00022% |
| 8    | 0.9730089 | 0.9733210 | 0.9733189 | 0.0003121  | 0.0003099  | 0.0000021 | 0.00022% |
| 9    | 0.9730089 | 0.9730742 | 0.9730740 | 0.0000653  | 0.0000650  | 0.0000002 | 0.00003% |
| 10   | 0.9730089 | 0.9730318 | 0.9730317 | 0.0000228  | 0.0000227  | 0.0000001 | 0.00001% |
| 11   | 0.9730089 | 0.9730009 | 0.9730009 | -0.0000081 | -0.0000081 | 0.0000000 | 0.00000% |
| 12   | 0.9730089 | 0.9729313 | 0.9729312 | -0.0000776 | -0.0000778 | 0.0000001 | 0.00001% |
| 13   | 0.9730089 | 0.9729352 | 0.9729351 | -0.0000738 | -0.0000739 | 0.0000001 | 0.00001% |
| 14   | 0.9730089 | 0.9730175 | 0.9730175 | 0.0000086  | 0.0000086  | 0.0000000 | 0.00000% |
**Table 5.27 –** Investigation of the accuracy of Approximation 2 for CMI data – age in 2005 (i) 35, (ii) 45, (iii) 55 – column (1) = 'deterministic' mean annuity value at end of 2006, (2) = linear approximation to conditional mean annuity value at end of 2006, (3) = actual conditional mean annuity value at end of 2006, (3) = absolute error in approximation of (3) by (2), (7) = percentage error in approximation of (3) by (2)

|    | ٠ | • |
|----|---|---|
|    |   | ۰ |
|    |   |   |
| ۰. |   | , |
| •  |   |   |

| Test | (1)     | (2)     | (3)     | (4)      | (5)      | (6)     | (7)     |
|------|---------|---------|---------|----------|----------|---------|---------|
| 1    | 4.38954 | 4.36616 | 4.36608 | -0.02338 | -0.02346 | 0.00008 | 0.0017% |
| 2    | 4.38954 | 4.34827 | 4.34797 | -0.04127 | -0.04157 | 0.00029 | 0.0067% |
| 3    | 4.38954 | 4.36363 | 4.36346 | -0.02591 | -0.02608 | 0.00017 | 0.0039% |
| 4    | 4.38954 | 4.38721 | 4.38719 | -0.00233 | -0.00235 | 0.00002 | 0.0004% |
| 5    | 4.38954 | 4.37909 | 4.37905 | -0.01045 | -0.01049 | 0.00004 | 0.0010% |
| 6    | 4.38954 | 4.37794 | 4.37793 | -0.01160 | -0.01161 | 0.00001 | 0.0003% |
| 7    | 4.38954 | 4.37931 | 4.37926 | -0.01023 | -0.01028 | 0.00005 | 0.0011% |
| 8    | 4.38954 | 4.40163 | 4.40155 | 0.01209  | 0.01201  | 0.00008 | 0.0018% |
| 9    | 4.38954 | 4.36003 | 4.35970 | -0.02951 | -0.02984 | 0.00033 | 0.0077% |
| 10   | 4.38954 | 4.39538 | 4.39537 | 0.00584  | 0.00583  | 0.00002 | 0.0003% |
| 11   | 4.38954 | 4.38945 | 4.38945 | -0.00009 | -0.00009 | 0.00000 | 0.0000% |
| 12   | 4.38954 | 4.38723 | 4.38722 | -0.00231 | -0.00232 | 0.00000 | 0.0000% |
| 13   | 4.38954 | 4.38885 | 4.38885 | -0.00069 | -0.00069 | 0.00000 | 0.0000% |
| 14   | 4.38954 | 4.38964 | 4.38964 | 0.00010  | 0.00010  | 0.00000 | 0.0000% |

(ii)

| Test | (1)     | (2)     | (3)     | (4)      | (5)      | (6)     | (7)     |
|------|---------|---------|---------|----------|----------|---------|---------|
| 1    | 6.51325 | 6.47773 | 6.47768 | -0.03552 | -0.03556 | 0.00004 | 0.0007% |
| 2    | 6.51325 | 6.44945 | 6.44918 | -0.06379 | -0.06406 | 0.00027 | 0.0042% |
| 3    | 6.51325 | 6.46611 | 6.46579 | -0.04714 | -0.04745 | 0.00032 | 0.0049% |
| 4    | 6.51325 | 6.51441 | 6.51439 | 0.00116  | 0.00114  | 0.00002 | 0.0003% |
| 5    | 6.51325 | 6.49189 | 6.49180 | -0.02135 | -0.02144 | 0.00009 | 0.0014% |
| 6    | 6.51325 | 6.49701 | 6.49701 | -0.01623 | -0.01623 | 0.00000 | 0.0000% |
| 7    | 6.51325 | 6.50376 | 6.50374 | -0.00948 | -0.00951 | 0.00003 | 0.0004% |
| 8    | 6.51325 | 6.53304 | 6.53286 | 0.01979  | 0.01961  | 0.00018 | 0.0027% |
| 9    | 6.51325 | 6.48255 | 6.48243 | -0.03070 | -0.03082 | 0.00012 | 0.0019% |
| 10   | 6.51325 | 6.52278 | 6.52276 | 0.00954  | 0.00951  | 0.00003 | 0.0004% |
| 11   | 6.51325 | 6.51311 | 6.51311 | -0.00014 | -0.00014 | 0.00000 | 0.0000% |
| 12   | 6.51325 | 6.50940 | 6.50940 | -0.00385 | -0.00385 | 0.00000 | 0.0000% |
| 13   | 6.51325 | 6.51150 | 6.51150 | -0.00175 | -0.00175 | 0.00000 | 0.0000% |
| 14   | 6.51325 | 6.51362 | 6.51362 | 0.00037  | 0.00037  | 0.00000 | 0.0000% |

(iii)

| Test | (1)     | (2)     | (3)     | (4)      | (5)      | (6)      | (7)      |
|------|---------|---------|---------|----------|----------|----------|----------|
| 1    | 9.63762 | 9.58712 | 9.58716 | -0.05049 | -0.05046 | -0.00003 | -0.0003% |
| 2    | 9.63762 | 9.54618 | 9.54609 | -0.09144 | -0.09153 | 0.00009  | 0.0009%  |
| 3    | 9.63762 | 9.55252 | 9.55191 | -0.08509 | -0.08571 | 0.00062  | 0.0065%  |
| 4    | 9.63762 | 9.65246 | 9.65242 | 0.01484  | 0.01481  | 0.00003  | 0.0004%  |
| 5    | 9.63762 | 9.59406 | 9.59384 | -0.04356 | -0.04378 | 0.00022  | 0.0023%  |
| 6    | 9.63762 | 9.61768 | 9.61769 | -0.01993 | -0.01992 | -0.00001 | -0.0001% |
| 7    | 9.63762 | 9.64107 | 9.64105 | 0.00346  | 0.00343  | 0.00002  | 0.0003%  |
| 8    | 9.63762 | 9.66625 | 9.66591 | 0.02863  | 0.02830  | 0.00034  | 0.0035%  |
| 9    | 9.63762 | 9.63313 | 9.63305 | -0.00448 | -0.00456 | 0.00008  | 0.0008%  |
| 10   | 9.63762 | 9.65163 | 9.65158 | 0.01402  | 0.01397  | 0.00005  | 0.0005%  |
| 11   | 9.63762 | 9.63741 | 9.63741 | -0.00021 | -0.00021 | 0.00000  | 0.0000%  |
| 12   | 9.63762 | 9.63145 | 9.63145 | -0.00616 | -0.00617 | 0.00000  | 0.0000%  |
| 13   | 9.63762 | 9.63318 | 9.63318 | -0.00443 | -0.00443 | 0.00000  | 0.0000%  |
| 14   | 9.63762 | 9.63846 | 9.63846 | 0.00085  | 0.00085  | 0.00000  | 0.0000%  |

**Table 5.27** (continued) – Investigation of the accuracy of Approximation 2 for CMI data – age in 2005 (iv) 65, (v) 75 – column (1) = 'deterministic' mean annuity value at end of 2006, (2) = linear approximation to conditional mean annuity value at end of 2006, (3) = actual conditional mean annuity value at end of 2006, (4) = (2) – (1), (5) = (3) – (1), (6) = absolute error in approximation of (3) by (2), (7) = percentage error in approximation of (3) by (2)

(iv)

| Test | (1)      | (2)      | (3)      | (4)      | (5)      | (6)      | (7)      |
|------|----------|----------|----------|----------|----------|----------|----------|
| 1    | 14.01838 | 13.94327 | 13.94344 | -0.07510 | -0.07493 | -0.00017 | -0.0012% |
| 2    | 14.01838 | 13.89906 | 13.89931 | -0.11931 | -0.11906 | -0.00025 | -0.0018% |
| 3    | 14.01838 | 13.88801 | 13.88709 | -0.13037 | -0.13128 | 0.00092  | 0.0066%  |
| 4    | 14.01838 | 14.06519 | 14.06506 | 0.04681  | 0.04668  | 0.00013  | 0.0009%  |
| 5    | 14.01838 | 13.94582 | 13.94541 | -0.07255 | -0.07297 | 0.00042  | 0.0030%  |
| 6    | 14.01838 | 13.99219 | 13.99221 | -0.02618 | -0.02616 | -0.00002 | -0.0001% |
| 7    | 14.01838 | 14.05461 | 14.05440 | 0.03623  | 0.03603  | 0.00021  | 0.0015%  |
| 8    | 14.01838 | 14.03529 | 14.03498 | 0.01691  | 0.01661  | 0.00030  | 0.0022%  |
| 9    | 14.01838 | 14.07293 | 14.07174 | 0.05455  | 0.05337  | 0.00118  | 0.0084%  |
| 10   | 14.01838 | 14.03039 | 14.03035 | 0.01202  | 0.01197  | 0.00005  | 0.0003%  |
| 11   | 14.01838 | 14.01815 | 14.01815 | -0.00023 | -0.00023 | 0.00000  | 0.0000%  |
| 12   | 14.01838 | 14.00985 | 14.00985 | -0.00852 | -0.00853 | 0.00000  | 0.0000%  |
| 13   | 14.01838 | 14.00821 | 14.00821 | -0.01017 | -0.01017 | 0.00000  | 0.0000%  |
| 14   | 14.01838 | 14.01779 | 14.01779 | -0.00058 | -0.00059 | 0.00000  | 0.0000%  |

(v)

| Test | (1)      | (2)      | (3)      | (4)      | (5)      | (6)      | (7)      |
|------|----------|----------|----------|----------|----------|----------|----------|
| 1    | 10.17410 | 10.04807 | 10.04868 | -0.12603 | -0.12542 | -0.00061 | -0.0061% |
| 2    | 10.17410 | 10.02132 | 10.02215 | -0.15277 | -0.15195 | -0.00083 | -0.0082% |
| 3    | 10.17410 | 10.02491 | 10.02438 | -0.14919 | -0.14972 | 0.00053  | 0.0053%  |
| 4    | 10.17410 | 10.25984 | 10.25963 | 0.08574  | 0.08554  | 0.00020  | 0.0020%  |
| 5    | 10.17410 | 10.09007 | 10.08975 | -0.08402 | -0.08435 | 0.00032  | 0.0032%  |
| 6    | 10.17410 | 10.12492 | 10.12505 | -0.04918 | -0.04904 | -0.00014 | -0.0014% |
| 7    | 10.17410 | 10.24239 | 10.24200 | 0.06829  | 0.06791  | 0.00038  | 0.0037%  |
| 8    | 10.17410 | 10.13223 | 10.13226 | -0.04187 | -0.04184 | -0.00003 | -0.0003% |
| 9    | 10.17410 | 10.25606 | 10.25430 | 0.08196  | 0.08020  | 0.00176  | 0.0171%  |
| 10   | 10.17410 | 10.16868 | 10.16867 | -0.00542 | -0.00543 | 0.00001  | 0.0001%  |
| 11   | 10.17410 | 10.17395 | 10.17395 | -0.00015 | -0.00015 | 0.00000  | 0.0000%  |
| 12   | 10.17410 | 10.16454 | 10.16454 | -0.00956 | -0.00956 | 0.00000  | 0.0000%  |
| 13   | 10.17410 | 10.15617 | 10.15617 | -0.01793 | -0.01793 | 0.00000  | 0.0000%  |
| 14   | 10.17410 | 10.16685 | 10.16685 | -0.00725 | -0.00725 | 0.00000  | 0.0000%  |

**Table 5.28** – Investigation of the accuracy of Approximation 2 for England and Wales male data – age in 2005 (i) 35, (ii) 45, (iii) 55 – column (1) = 'deterministic' mean annuity value at end of 2006, (2) = linear approximation to conditional mean annuity value at end of 2006, (3) = actual conditional mean annuity value at end of 2006, (4) = (2) - (1), (5) = (3) - (1), (6) = absolute error in approximation of (3) by (2), (7) = percentage error in approximation of (3) by (2)

(i)

| Test | (1)     | (2)     | (3)     | (4)      | (5)      | (6)     | (7)     |
|------|---------|---------|---------|----------|----------|---------|---------|
| 1    | 3.52276 | 3.49756 | 3.49751 | -0.02521 | -0.02525 | 0.00005 | 0.0013% |
| 2    | 3.52276 | 3.47717 | 3.47697 | -0.04559 | -0.04580 | 0.00021 | 0.0060% |
| 3    | 3.52276 | 3.49719 | 3.49710 | -0.02558 | -0.02567 | 0.00009 | 0.0025% |
| 4    | 3.52276 | 3.51767 | 3.51766 | -0.00509 | -0.00510 | 0.00001 | 0.0003% |
| 5    | 3.52276 | 3.51297 | 3.51296 | -0.00979 | -0.00981 | 0.00002 | 0.0005% |
| 6    | 3.52276 | 3.51260 | 3.51260 | -0.01016 | -0.01016 | 0.00000 | 0.0001% |
| 7    | 3.52276 | 3.50835 | 3.50832 | -0.01441 | -0.01445 | 0.00003 | 0.0009% |
| 8    | 3.52276 | 3.53829 | 3.53823 | 0.01552  | 0.01547  | 0.00006 | 0.0016% |
| 9    | 3.52276 | 3.47718 | 3.47681 | -0.04558 | -0.04596 | 0.00037 | 0.0107% |
| 10   | 3.52276 | 3.53076 | 3.53074 | 0.00799  | 0.00798  | 0.00001 | 0.0004% |
| 11   | 3.52276 | 3.52250 | 3.52250 | -0.00026 | -0.00026 | 0.00000 | 0.0000% |
| 12   | 3.52276 | 3.52016 | 3.52016 | -0.00260 | -0.00260 | 0.00000 | 0.0000% |
| 13   | 3.52276 | 3.52247 | 3.52247 | -0.00029 | -0.00029 | 0.00000 | 0.0000% |
| 14   | 3.52276 | 3.52298 | 3.52298 | 0.00022  | 0.00022  | 0.00000 | 0.0000% |

(ii)

| Test | (1)     | (2)     | (3)     | (4)      | (5)      | (6)     | (7)     |
|------|---------|---------|---------|----------|----------|---------|---------|
| 1    | 5.27308 | 5.23410 | 5.23407 | -0.03897 | -0.03901 | 0.00003 | 0.0006% |
| 2    | 5.27308 | 5.20116 | 5.20091 | -0.07191 | -0.07216 | 0.00025 | 0.0048% |
| 3    | 5.27308 | 5.22883 | 5.22868 | -0.04425 | -0.04439 | 0.00014 | 0.0027% |
| 4    | 5.27308 | 5.26651 | 5.26649 | -0.00657 | -0.00658 | 0.00002 | 0.0003% |
| 5    | 5.27308 | 5.25532 | 5.25529 | -0.01775 | -0.01778 | 0.00003 | 0.0005% |
| 6    | 5.27308 | 5.25576 | 5.25575 | -0.01731 | -0.01732 | 0.00001 | 0.0002% |
| 7    | 5.27308 | 5.25260 | 5.25256 | -0.02047 | -0.02052 | 0.00004 | 0.0008% |
| 8    | 5.27308 | 5.29854 | 5.29843 | 0.02546  | 0.02535  | 0.00011 | 0.0020% |
| 9    | 5.27308 | 5.21260 | 5.21242 | -0.06048 | -0.06065 | 0.00018 | 0.0034% |
| 10   | 5.27308 | 5.28536 | 5.28534 | 0.01229  | 0.01227  | 0.00002 | 0.0003% |
| 11   | 5.27308 | 5.27272 | 5.27272 | -0.00035 | -0.00035 | 0.00000 | 0.0000% |
| 12   | 5.27308 | 5.26900 | 5.26899 | -0.00408 | -0.00408 | 0.00000 | 0.0000% |
| 13   | 5.27308 | 5.27229 | 5.27229 | -0.00079 | -0.00079 | 0.00000 | 0.0000% |
| 14   | 5.27308 | 5.27366 | 5.27366 | 0.00059  | 0.00058  | 0.00000 | 0.0000% |

(iii)

| Test | (1)     | (2)     | (3)     | (4)      | (5)      | (6)      | (7)      |
|------|---------|---------|---------|----------|----------|----------|----------|
| 1    | 8.02286 | 7.96338 | 7.96339 | -0.05949 | -0.05948 | -0.00001 | -0.0001% |
| 2    | 8.02286 | 7.91255 | 7.91229 | -0.11032 | -0.11057 | 0.00026  | 0.0032%  |
| 3    | 8.02286 | 7.94724 | 7.94699 | -0.07563 | -0.07588 | 0.00025  | 0.0032%  |
| 4    | 8.02286 | 8.01762 | 8.01760 | -0.00524 | -0.00526 | 0.00002  | 0.0002%  |
| 5    | 8.02286 | 7.99065 | 7.99060 | -0.03222 | -0.03226 | 0.00005  | 0.0006%  |
| 6    | 8.02286 | 7.99425 | 7.99422 | -0.02862 | -0.02865 | 0.00003  | 0.0004%  |
| 7    | 8.02286 | 7.99790 | 7.99786 | -0.02496 | -0.02500 | 0.00004  | 0.0005%  |
| 8    | 8.02286 | 8.06138 | 8.06121 | 0.03852  | 0.03834  | 0.00018  | 0.0022%  |
| 9    | 8.02286 | 7.95332 | 7.95346 | -0.06954 | -0.06940 | -0.00014 | -0.0017% |
| 10   | 8.02286 | 8.04057 | 8.04056 | 0.01771  | 0.01769  | 0.00002  | 0.0002%  |
| 11   | 8.02286 | 8.02243 | 8.02243 | -0.00044 | -0.00044 | 0.00000  | 0.0000%  |
| 12   | 8.02286 | 8.01674 | 8.01674 | -0.00613 | -0.00613 | 0.00000  | 0.0000%  |
| 13   | 8.02286 | 8.02078 | 8.02078 | -0.00208 | -0.00208 | 0.00000  | 0.0000%  |
| 14   | 8.02286 | 8.02395 | 8.02395 | 0.00108  | 0.00108  | 0.00000  | 0.0000%  |

**Table 5.28** (continued) – Investigation of the accuracy of Approximation 2 for England and Wales male data – age in 2005 (iv) 65, (v) 75 – column (1) = 'deterministic' mean annuity value at end of 2006, (2) = linear approximation to conditional mean annuity value at end of 2006, (3) = actual conditional mean annuity value at end of 2006, (4) = (2) – (1), (5) = (3) – (1), (6) = absolute error in approximation of (3) by (2), (7) = percentage error in approximation of (3) by (2)

(iv)

| Test | (1)      | (2)      | (3)      | (4)      | (5)      | (6)      | (7)      |
|------|----------|----------|----------|----------|----------|----------|----------|
| 1    | 12.24807 | 12.16089 | 12.16101 | -0.08718 | -0.08706 | -0.00012 | -0.0010% |
| 2    | 12.24807 | 12.09734 | 12.09727 | -0.15073 | -0.15080 | 0.00007  | 0.0006%  |
| 3    | 12.24807 | 12.13780 | 12.13748 | -0.11027 | -0.11059 | 0.00032  | 0.0026%  |
| 4    | 12.24807 | 12.25522 | 12.25522 | 0.00715  | 0.00715  | 0.00000  | 0.0000%  |
| 5    | 12.24807 | 12.19716 | 12.19710 | -0.05091 | -0.05097 | 0.00006  | 0.0005%  |
| 6    | 12.24807 | 12.21214 | 12.21209 | -0.03593 | -0.03598 | 0.00005  | 0.0004%  |
| 7    | 12.24807 | 12.22993 | 12.22992 | -0.01814 | -0.01816 | 0.00002  | 0.0001%  |
| 8    | 12.24807 | 12.28888 | 12.28871 | 0.04081  | 0.04063  | 0.00017  | 0.0014%  |
| 9    | 12.24807 | 12.17213 | 12.17255 | -0.07594 | -0.07552 | -0.00042 | -0.0035% |
| 10   | 12.24807 | 12.26986 | 12.26986 | 0.02179  | 0.02178  | 0.00001  | 0.0001%  |
| 11   | 12.24807 | 12.24771 | 12.24771 | -0.00036 | -0.00036 | 0.00000  | 0.0000%  |
| 12   | 12.24807 | 12.24028 | 12.24028 | -0.00779 | -0.00779 | 0.00000  | 0.0000%  |
| 13   | 12.24807 | 12.24339 | 12.24339 | -0.00468 | -0.00468 | 0.00000  | 0.0000%  |
| 14   | 12.24807 | 12.24827 | 12.24827 | 0.00020  | 0.00020  | 0.00000  | 0.0000%  |

(v)

| Test | (1)     | (2)     | (3)     | (4)      | (5)      | (6)      | (7)      |
|------|---------|---------|---------|----------|----------|----------|----------|
| 1    | 8.49986 | 8.38437 | 8.38479 | -0.11549 | -0.11507 | -0.00042 | -0.0050% |
| 2    | 8.49986 | 8.32729 | 8.32791 | -0.17258 | -0.17196 | -0.00062 | -0.0075% |
| 3    | 8.49986 | 8.37758 | 8.37758 | -0.12228 | -0.12229 | 0.00000  | 0.0000%  |
| 4    | 8.49986 | 8.53047 | 8.53050 | 0.03061  | 0.03063  | -0.00003 | -0.0003% |
| 5    | 8.49986 | 8.43822 | 8.43823 | -0.06164 | -0.06164 | 0.00000  | 0.0000%  |
| 6    | 8.49986 | 8.47137 | 8.47135 | -0.02849 | -0.02852 | 0.00002  | 0.0003%  |
| 7    | 8.49986 | 8.50118 | 8.50118 | 0.00132  | 0.00131  | 0.00000  | 0.0000%  |
| 8    | 8.49986 | 8.51804 | 8.51798 | 0.01817  | 0.01812  | 0.00005  | 0.0006%  |
| 9    | 8.49986 | 8.39840 | 8.39925 | -0.10147 | -0.10062 | -0.00085 | -0.0101% |
| 10   | 8.49986 | 8.52151 | 8.52151 | 0.02164  | 0.02165  | -0.00001 | -0.0001% |
| 11   | 8.49986 | 8.49967 | 8.49967 | -0.00020 | -0.00020 | 0.00000  | 0.0000%  |
| 12   | 8.49986 | 8.49204 | 8.49204 | -0.00783 | -0.00782 | 0.00000  | 0.0000%  |
| 13   | 8.49986 | 8.49230 | 8.49230 | -0.00756 | -0.00756 | 0.00000  | 0.0000%  |
| 14   | 8.49986 | 8.49635 | 8.49635 | -0.00351 | -0.00351 | 0.00000  | 0.0000%  |

**Table 5.29** – Investigation of the accuracy of Approximation 2 for England and Wales female data – age in 2005 (i) 35, (ii) 45, (iii) 55 – column (1) = 'deterministic' mean annuity value at end of 2006, (2) = linear approximation to conditional mean annuity value at end of 2006, (3) = actual conditional mean annuity value at end of 2006, (4) = (2) - (1), (5) = (3) - (1), (6) = absolute error in approximation of (3) by (2), (7) = percentage error in approximation of (3) by (2)

| 1  | ٠ | ١. |
|----|---|----|
| 1  | ı | ۱  |
| ١. | L |    |
| •  |   |    |

| Test | (1)     | (2)     | (3)     | (4)      | (5)      | (6)     | (7)     |
|------|---------|---------|---------|----------|----------|---------|---------|
| 1    | 3.71396 | 3.68773 | 3.68766 | -0.02623 | -0.02630 | 0.00006 | 0.0017% |
| 2    | 3.71396 | 3.66715 | 3.66689 | -0.04681 | -0.04707 | 0.00026 | 0.0070% |
| 3    | 3.71396 | 3.68868 | 3.68859 | -0.02528 | -0.02537 | 0.00009 | 0.0025% |
| 4    | 3.71396 | 3.70832 | 3.70832 | -0.00563 | -0.00564 | 0.00000 | 0.0001% |
| 5    | 3.71396 | 3.70467 | 3.70466 | -0.00929 | -0.00930 | 0.00002 | 0.0005% |
| 6    | 3.71396 | 3.70277 | 3.70275 | -0.01119 | -0.01121 | 0.00002 | 0.0007% |
| 7    | 3.71396 | 3.69852 | 3.69850 | -0.01544 | -0.01546 | 0.00003 | 0.0007% |
| 8    | 3.71396 | 3.72908 | 3.72901 | 0.01512  | 0.01505  | 0.00007 | 0.0019% |
| 9    | 3.71396 | 3.66623 | 3.66538 | -0.04773 | -0.04858 | 0.00085 | 0.0233% |
| 10   | 3.71396 | 3.72176 | 3.72173 | 0.00780  | 0.00777  | 0.00003 | 0.0008% |
| 11   | 3.71396 | 3.71350 | 3.71350 | -0.00046 | -0.00046 | 0.00000 | 0.0000% |
| 12   | 3.71396 | 3.71108 | 3.71108 | -0.00287 | -0.00288 | 0.00000 | 0.0000% |
| 13   | 3.71396 | 3.71352 | 3.71352 | -0.00044 | -0.00044 | 0.00000 | 0.0000% |
| 14   | 3.71396 | 3.71410 | 3.71410 | 0.00014  | 0.00014  | 0.00000 | 0.0000% |

(ii)

| Test | (1)     | (2)     | (3)     | (4)      | (5)      | (6)     | (7)     |
|------|---------|---------|---------|----------|----------|---------|---------|
| 1    | 5.70037 | 5.66011 | 5.66004 | -0.04027 | -0.04034 | 0.00007 | 0.0012% |
| 2    | 5.70037 | 5.62787 | 5.62752 | -0.07251 | -0.07285 | 0.00034 | 0.0061% |
| 3    | 5.70037 | 5.65840 | 5.65825 | -0.04197 | -0.04212 | 0.00015 | 0.0027% |
| 4    | 5.70037 | 5.69207 | 5.69206 | -0.00831 | -0.00832 | 0.00001 | 0.0002% |
| 5    | 5.70037 | 5.68508 | 5.68505 | -0.01529 | -0.01532 | 0.00003 | 0.0005% |
| 6    | 5.70037 | 5.67817 | 5.67816 | -0.02221 | -0.02222 | 0.00001 | 0.0002% |
| 7    | 5.70037 | 5.67736 | 5.67732 | -0.02302 | -0.02305 | 0.00004 | 0.0007% |
| 8    | 5.70037 | 5.72308 | 5.72297 | 0.02271  | 0.02259  | 0.00011 | 0.0020% |
| 9    | 5.70037 | 5.64150 | 5.64091 | -0.05888 | -0.05947 | 0.00059 | 0.0105% |
| 10   | 5.70037 | 5.71060 | 5.71057 | 0.01023  | 0.01020  | 0.00003 | 0.0005% |
| 11   | 5.70037 | 5.69976 | 5.69976 | -0.00062 | -0.00062 | 0.00000 | 0.0000% |
| 12   | 5.70037 | 5.69601 | 5.69601 | -0.00437 | -0.00437 | 0.00000 | 0.0000% |
| 13   | 5.70037 | 5.69915 | 5.69915 | -0.00122 | -0.00122 | 0.00000 | 0.0000% |
| 14   | 5.70037 | 5.70074 | 5.70074 | 0.00037  | 0.00037  | 0.00000 | 0.0000% |

(iii)

| Test | (1)     | (2)     | (3)     | (4)      | (5)      | (6)     | (7)     |
|------|---------|---------|---------|----------|----------|---------|---------|
| 1    | 8.75036 | 8.68769 | 8.68763 | -0.06268 | -0.06273 | 0.00006 | 0.0007% |
| 2    | 8.75036 | 8.63810 | 8.63768 | -0.11226 | -0.11268 | 0.00042 | 0.0049% |
| 3    | 8.75036 | 8.68048 | 8.68023 | -0.06988 | -0.07013 | 0.00025 | 0.0029% |
| 4    | 8.75036 | 8.73936 | 8.73934 | -0.01101 | -0.01102 | 0.00002 | 0.0002% |
| 5    | 8.75036 | 8.72508 | 8.72504 | -0.02528 | -0.02532 | 0.00004 | 0.0005% |
| 6    | 8.75036 | 8.70649 | 8.70647 | -0.04388 | -0.04389 | 0.00002 | 0.0002% |
| 7    | 8.75036 | 8.71739 | 8.71734 | -0.03297 | -0.03302 | 0.00005 | 0.0006% |
| 8    | 8.75036 | 8.78207 | 8.78192 | 0.03171  | 0.03155  | 0.00016 | 0.0018% |
| 9    | 8.75036 | 8.68809 | 8.68791 | -0.06227 | -0.06245 | 0.00018 | 0.0021% |
| 10   | 8.75036 | 8.76208 | 8.76205 | 0.01171  | 0.01169  | 0.00002 | 0.0003% |
| 11   | 8.75036 | 8.74958 | 8.74958 | -0.00078 | -0.00078 | 0.00000 | 0.0000% |
| 12   | 8.75036 | 8.74388 | 8.74387 | -0.00649 | -0.00649 | 0.00000 | 0.0000% |
| 13   | 8.75036 | 8.74728 | 8.74728 | -0.00309 | -0.00309 | 0.00000 | 0.0000% |
| 14   | 8.75036 | 8.75094 | 8.75094 | 0.00058  | 0.00058  | 0.00000 | 0.0000% |

**Table 5.29** (continued) – Investigation of the accuracy of Approximation 2 for England and Wales female data – age in 2005 (iv) 65, (v) 75 – column (1) = 'deterministic' mean annuity value at end of 2006, (2) = linear approximation to conditional mean annuity value at end of 2006, (3) = actual conditional mean annuity value at end of 2006, (4) = (2) – (1), (5) = (3) – (1), (6) = absolute error in approximation of (3) by (2), (7) = percentage error in approximation of (3) by (2)

(iv)

| Test | (1)      | (2)      | (3)      | (4)      | (5)      | (6)      | (7)      |
|------|----------|----------|----------|----------|----------|----------|----------|
| 1    | 13.30001 | 13.20689 | 13.20689 | -0.09312 | -0.09312 | 0.00000  | 0.0000%  |
| 2    | 13.30001 | 13.14210 | 13.14176 | -0.15790 | -0.15825 | 0.00035  | 0.0026%  |
| 3    | 13.30001 | 13.19748 | 13.19717 | -0.10252 | -0.10284 | 0.00032  | 0.0024%  |
| 4    | 13.30001 | 13.29238 | 13.29236 | -0.00763 | -0.00764 | 0.00001  | 0.0001%  |
| 5    | 13.30001 | 13.26237 | 13.26232 | -0.03764 | -0.03769 | 0.00005  | 0.0004%  |
| 6    | 13.30001 | 13.22828 | 13.22818 | -0.07173 | -0.07183 | 0.00010  | 0.0008%  |
| 7    | 13.30001 | 13.26219 | 13.26214 | -0.03782 | -0.03786 | 0.00005  | 0.0003%  |
| 8    | 13.30001 | 13.33162 | 13.33149 | 0.03161  | 0.03148  | 0.00013  | 0.0010%  |
| 9    | 13.30001 | 13.24541 | 13.24558 | -0.05459 | -0.05442 | -0.00017 | -0.0013% |
| 10   | 13.30001 | 13.30972 | 13.30972 | 0.00972  | 0.00971  | 0.00001  | 0.0001%  |
| 11   | 13.30001 | 13.29936 | 13.29936 | -0.00065 | -0.00065 | 0.00000  | 0.0000%  |
| 12   | 13.30001 | 13.29176 | 13.29176 | -0.00824 | -0.00824 | 0.00000  | 0.0000%  |
| 13   | 13.30001 | 13.29343 | 13.29343 | -0.00657 | -0.00657 | 0.00000  | 0.0000%  |
| 14   | 13.30001 | 13.29962 | 13.29962 | -0.00038 | -0.00038 | 0.00000  | 0.0000%  |

(v)

| Test | (1)     | (2)     | (3)     | (4)      | (5)      | (6)      | (7)      |
|------|---------|---------|---------|----------|----------|----------|----------|
| 1    | 9.40787 | 9.28182 | 9.28209 | -0.12606 | -0.12579 | -0.00027 | -0.0029% |
| 2    | 9.40787 | 9.21288 | 9.21324 | -0.19499 | -0.19463 | -0.00036 | -0.0039% |
| 3    | 9.40787 | 9.28254 | 9.28248 | -0.12533 | -0.12540 | 0.00006  | 0.0007%  |
| 4    | 9.40787 | 9.41208 | 9.41208 | 0.00421  | 0.00421  | 0.00000  | 0.0000%  |
| 5    | 9.40787 | 9.36082 | 9.36081 | -0.04706 | -0.04706 | 0.00001  | 0.0001%  |
| 6    | 9.40787 | 9.31408 | 9.31402 | -0.09379 | -0.09385 | 0.00006  | 0.0007%  |
| 7    | 9.40787 | 9.37452 | 9.37452 | -0.03335 | -0.03336 | 0.00000  | 0.0000%  |
| 8    | 9.40787 | 9.42309 | 9.42305 | 0.01522  | 0.01518  | 0.00004  | 0.0004%  |
| 9    | 9.40787 | 9.35774 | 9.35802 | -0.05013 | -0.04985 | -0.00028 | -0.0030% |
| 10   | 9.40787 | 9.41204 | 9.41204 | 0.00417  | 0.00417  | 0.00000  | 0.0000%  |
| 11   | 9.40787 | 9.40751 | 9.40751 | -0.00037 | -0.00037 | 0.00000  | 0.0000%  |
| 12   | 9.40787 | 9.39914 | 9.39914 | -0.00873 | -0.00873 | 0.00000  | 0.0000%  |
| 13   | 9.40787 | 9.39715 | 9.39715 | -0.01072 | -0.01072 | 0.00000  | 0.0000%  |
| 14   | 9.40787 | 9.40452 | 9.40452 | -0.00335 | -0.00335 | 0.00000  | 0.0000%  |



Year, t

**Figure 5.1** – Projection of the future  $\kappa^{(0)}$  parameters – solid curve = median, dashed curves = 95% prediction limits (2.5<sup>th</sup> and 97.5<sup>th</sup> percentiles) – (i) CMI data, (ii) England and Wales male data, (iii) England and Wales female data



**Figure 5.2** – Projection of the future  $\kappa^{(3)}$  parameters – solid curve = median, dashed curves = 95% prediction limits (2.5<sup>th</sup> and 97.5<sup>th</sup> percentiles) – (i) CMI data, (ii) England and Wales male data, (iii) England and Wales female data



Year, t

**Figure 5.3** – Projection of the future  $\kappa^{(4)}$  parameters – solid curve = median, dashed curves = 95% prediction limits (2.5<sup>th</sup> and 97.5<sup>th</sup> percentiles) – (i) CMI data, (ii) England and Wales male data, (iii) England and Wales female data



Year, t

**Figure 5.4** – Projection of the future  $\kappa^{(5)}$  parameters – solid curve = median, dashed curves = 95% prediction limits (2.5<sup>th</sup> and 97.5<sup>th</sup> percentiles) – (i) CMI data, (ii) England and Wales male data, (iii) England and Wales female data



Year of birth, c

**Figure 5.5** – Projection of the future γ parameters – solid curve = median, dashed curves = 95% prediction limits (2.5<sup>th</sup> and 97.5<sup>th</sup> percentiles) – (i) CMI data, (ii) England and Wales male data, (iii) England and Wales female data



Year

**Figure 5.6** – Extensions of the graphs in Figure 1.2 to future years, using a deterministic projection with all the future innovation terms set to zero – (i) CMI data, (ii) England and Wales male data, (iii) England and Wales female data

**Figure 5.7** – Projected forces of mortality for CMI data compared with previously published projection bases – age (i) 30, (ii) 50, (iii) 70, (iv) 90. Black solid curve = median, black dashed curves = 2.5<sup>th</sup> and 97.5<sup>th</sup> percentiles, blue solid curve = Medium Cohort, green solid curve = Medium Cohort subject to a 1% underpin, red solid curve = Long Cohort.



Figure 5.8 - Projected forces of mortality for England and Wales male data compared with GAD projection bases – age (i) 30, (ii) 50, (iii) 70, (iv) 89. Black solid curve = median, black dashed curves = 2.5<sup>th</sup> and 97.5<sup>th</sup> percentiles, blue solid curve = LLE, green solid curve = Principal, red solid curve = HLE.



(ii)

**Figure 5.9** – Projected forces of mortality for England and Wales female data compared with GAD projection bases – age (i) 30, (ii) 50, (iii) 70, (iv) 89. Black solid curve = median, black dashed curves = 2.5<sup>th</sup> and 97.5<sup>th</sup> percentiles, blue solid curve = LLE, green solid curve = Principal, red solid curve = HLE.







-4

-6

-8

Year





-8

Year





**Figure 5.13 –** Projected forces of mortality for CMI data under the low improvement assumption compared with previously published projection bases – age in 2005 (i) 35, (ii) 45, (iii) 55, (iv) 65, (v) 75. Black solid curve = median, black dashed curves = 2.5<sup>th</sup> and 97.5<sup>th</sup> percentiles, blue solid curve = Medium Cohort, green solid curve = Medium Cohort subject to a 1% underpin, red solid curve = Long Cohort.



**Figure 5.14** – Projected forces of mortality for CMI data under the high improvement assumption compared with previously published projection bases – age in 2005 (i) 35, (ii) 45, (iii) 55, (iv) 65, (v) 75. Black solid curve = median, black dashed curves = 2.5<sup>th</sup> and 97.5<sup>th</sup> percentiles, blue solid curve = Medium Cohort, green solid curve = Medium Cohort subject to a 1% underpin, red solid curve = Long Cohort.



ကု

80

90

100

Age

110

**Figure 5.15 –** Projected forces of mortality for England and Wales male data under the low improvement assumption compared with GAD projection bases – age in 2005 (i) 35, (ii) 45, (iii) 55, (iv) 65, (v) 75. Black solid curve = median, black dashed curves = 2.5<sup>th</sup> and 97.5<sup>th</sup> percentiles, blue solid curve = LLE, green solid curve = Principal, red solid curve = HLE.



Age

**Figure 5.16 –** Projected forces of mortality for England and Wales male data under the high improvement assumption compared with GAD projection bases – age in 2005 (i) 35, (ii) 45, (ii) 55, (iv) 65, (v) 75. Black solid curve = median, black dashed curves = 2.5<sup>th</sup> and 97.5<sup>th</sup> percentiles, blue solid curve = LLE, green solid curve = Principal, red solid curve = HLE.



Aae

**Figure 5.17** – Projected forces of mortality for England and Wales female data under the low improvement assumption compared with GAD projection bases – age in 2005 (i) 35, (ii) 45, (iii) 55, (iv) 65, (v) 75. Black solid curve = median, black dashed curves = 2.5<sup>th</sup> and 97.5<sup>th</sup> percentiles, blue solid curve = LLE, green solid curve = Principal, red solid curve = HLE.



Age

**Figure 5.18** – Projected forces of mortality for England and Wales female data under the high improvement assumption compared with GAD projection bases – age in 2005 (i) 35, (ii) 45, (iii) 55, (iv) 65, (v) 75. Black solid curve = median, black dashed curves = 2.5<sup>th</sup> and 97.5<sup>th</sup> percentiles, blue solid curve = LLE, green solid curve = Principal, red solid curve = HLE.



ကု

Aae

**Figure 5.19** – Projected probabilities of survival for CMI data under the low improvement assumption compared with previously published projection bases – age in 2005 (i) 35, (ii) 45, (iii) 55, (iv) 65, (v) 75. Black solid curve = median, black dashed curves =  $2.5^{\text{th}}$  and  $97.5^{\text{th}}$  percentiles, blue solid curve = Medium Cohort, green solid curve = Medium Cohort subject to a 1% underpin, red solid curve = Long Cohort.



**Figure 5.20** – Projected probabilities of survival for CMI data under the high improvement assumption compared with previously published projection bases – age in 2005 (i) 35, (ii) 45, (iii) 55, (iv) 65, (v) 75. Black solid curve = median, black dashed curves =  $2.5^{\text{th}}$  and  $97.5^{\text{th}}$  percentiles, blue solid curve = Medium Cohort, green solid curve = Medium Cohort subject to a 1% underpin, red solid curve = Long Cohort.



**Figure 5.21 –** Projected probabilities of survival for England and Wales male data under the low improvement assumption compared with GAD projection bases – age in 2005 (i) 35, (ii) 45, (iii) 55, (iv) 65, (v) 75. Black solid curve = median, black dashed curves = 2.5<sup>th</sup> and 97.5<sup>th</sup> percentiles, blue solid curve = LLE, green solid curve = Principal, red solid curve = HLE.



**Figure 5.22** – Projected probabilities of survival for England and Wales male data under the high improvement assumption compared with GAD projection bases – age in 2005 (i) 35, (ii) 45, (iii) 55, (iv) 65, (v) 75. Black solid curve = median, black dashed curves = 2.5<sup>th</sup> and 97.5<sup>th</sup> percentiles, blue solid curve = LLE, green solid curve = Principal, red solid curve = HLE.



Figure 5.23 – Projected probabilities of survival for England and Wales female data under the low improvement assumption compared with GAD projection bases – age in 2005 (i) 35, (ii) 45, (iii) 55, (iv) 65, (v) 75. Black solid curve = median, black dashed curves = 2.5<sup>th</sup> and 97.5<sup>th</sup> percentiles, blue solid curve = LLE, green solid curve = Principal, red solid curve = HLE.



Figure 5.24 – Projected probabilities of survival for England and Wales female data under the high improvement assumption compared with GAD projection bases – age in 2005 (i) 35, (ii) 45, (iii) 55, (iv) 65, (v) 75. Black solid curve = median, black dashed curves = 2.5<sup>th</sup> and 97.5<sup>th</sup> percentiles, blue solid curve = LLE, green solid curve = Principal, red solid curve = HLE.



**Figure 5.25** – Projected future sizes of a fund set up in 2005 equal to the mean annuity value plus ICA capital for CMI data – age in 2005 (i) 35, (ii) 45, (iii) 55, (iv) 65, (v) 75. Black solid curve = median, black dashed curves = 2.5<sup>th</sup> and 97.5<sup>th</sup> percentiles, red solid curve = 1 - percentile specified in Table 5.18.







**Figure 5.27 –** Projected future sizes of a fund set up in 2005 equal to the mean annuity value plus ICA capital for England and Wales female data – age in 2005 (i) 35, (ii) 45, (iii) 55, (iv) 65, (v) 75. Black solid curve = median, black dashed curves = 2.5<sup>th</sup> and 97.5<sup>th</sup> percentiles, red solid curve = 1 - percentile specified in Table 5.18.





**Figure 5.28** – Scatter diagrams to illustrate the correlation between the values of C(x) at different ages x in 2005 for CMI data – (i) 35 v 45, (ii) 35 v 55, (iii) 35 v 65, (iv) 35 v 75



**Figure 5.28** (continued) – Scatter diagrams to illustrate the correlation between the values of C(x) at different ages x in 2005 for CMI data – (v) 45 v 55, (vi) 45 v 65, (vii) 45 v 75, (viii) 55 v 65, (ix) 55 v 75, (x) 65 v 75



**Figure 5.29** – Scatter diagrams to illustrate the correlation between the values of C(x) at different ages x in 2005 for England and Wales male data – (i) 35 v 45, (ii) 35 v 55, (iii) 35 v 65, (iv) 35 v 75


**Figure 5.29** (continued) – Scatter diagrams to illustrate the correlation between the values of C(x) at different ages x in 2005 for England and Wales male data – (v) 45 v 55, (vi) 45 v 65, (vii) 45 v 75, (viii) 55 v 65, (ix) 55 v 75, (x) 65 v 75



**Figure 5.30** – Scatter diagrams to illustrate the correlation between the values of C(x) at different ages x in 2005 for England and Wales female data – (i) 35 v 45, (ii) 35 v 55, (iii) 35 v 65, (iv) 35 v 75



**Figure 5.30** (continued) – Scatter diagrams to illustrate the correlation between the values of C(x) at different ages x in 2005 for England and Wales female data – (v) 45 v 55, (vi) 45 v 65, (vii) 45 v 75, (viii) 55 v 65, (ix) 55 v 75, (x) 65 v 75







**Figure 5.32** – Illustration of the accuracy of Approximation 1 for England and Wales male data – columns (4) and (5) are as in Table 5.25 – age in 2005 (i) 35, (ii) 45, (iii) 55, (iv) 65, (v) 75



**Figure 5.33** – Illustration of the accuracy of Approximation 1 for England and Wales female data – columns (4) and (5) are as in Table 5.26 – age in 2005 (i) 35, (ii) 45, (iii) 55, (iv) 65, (v) 75







**Figure 5.35** – Illustration of the accuracy of Approximation 2 for England and Wales male data – columns (4) and (5) are as in Table 5.28 – age in 2005 (i) 35, (ii) 45, (iii) 55, (iv) 65, (v) 75



**Figure 5.36** – Illustration of the accuracy of Approximation 2 for England and Wales female data – columns (4) and (5) are as in Table 5.29 – age in 2005 (i) 35, (ii) 45, (iii) 55, (iv) 65, (v) 75

# 6: Conclusions

In Chapter 2, after fitting a number of Gompertz-Makeham models with time-dependent coefficients, the kappa parameters, to CMI data, England and Wales male data and England and Wales female data, we concluded that the GM(1,3) model gave a satisfactory fit to all three data sets and that satisfactory explanations could be given for the shapes of the resulting parameter graphs. However, some of the explanations depended upon cohort effects, and there were also cohort effects visible in the residuals, particularly for the England and Wales data sets. As a result, it was decided to introduce a further parameter  $\gamma_c$  depending on year of birth *c* into the model, applied to the force of mortality under the GM(1,3) model as a multiplicative factor.

Chapter 3 estimated  $\gamma_c$ , for each year of birth *c*, as the ratio of the actual number of deaths for year of birth *c* to the expected number of deaths under the model not including gamma parameters. For England and Wales data, the improvement in fit achieved by introducing the gamma parameters was found to be very significant. For CMI data, the improvement was more marginal – it appeared that this was because most of the cohort effects had already been captured implicitly via the shapes of the kappa parameter graphs and also because the smaller volume of the CMI data made any remaining cohort effects less statistically significant. Nevertheless, the similarities between the shapes of the gamma parameter graphs obtained for CMI data and for England and Wales data provided justification for introducing the gamma parameters for CMI data as well. Further improvement in fit to all three data sets, in a pure maximum likelihood sense, could have been achieved by re-estimating the kappa parameters following the estimation of the gamma parameters, but this was not implemented because it was found that the gamma parameters then reflected effects that were clearly not genuine cohort effects.

Chapter 4 fitted univariate time series models to the estimates of both the kappa and the gamma parameters. For the  $\kappa^{(3)}$  parameters, representing the general level of mortality in each calendar year, there was found to be a significant downward trend, which was greatest for CMI data and least for England and Wales female data, and an ARIMA(0,1,1) model was used. For the other parameters, an AR(1) model was used. In the case of the  $\kappa^{(0)}$  parameters, it was considered appropriate to use standard techniques to estimate the parameters of this AR(1) model. In the case of the  $\kappa^{(4)}$  and  $\kappa^{(5)}$  parameters, consideration of the impact of cohort effects was used to suggest a long-term mean value which was not necessarily similar to what would have been obtained using standard techniques. As the procedure we used to estimate the  $\gamma$  parameters was such that their logarithms could be away from zero only over a limited period, an AR(1) model with a zero mean value was fitted to the logarithms. The reason for fitting a time series model to the logarithms, rather than to the  $\gamma$  parameters themselves, was to eliminate the possibility of negative  $\gamma$  parameters.

In Chapter 5, we generated 500 scenarios for future values of the parameters for each data set, allowing for the observed correlations between the different kappa series. These scenarios were then used to calculate distributions of deferred and immediate annuity values for various values of the age in 2005 (the latest year in the data), using a deterministic interest rate. The annuity values were highest for CMI data and lowest for England and Wales male data, confirming that the mortality differential between assured lives and the general population is greater than that between males and females. It was necessary to extrapolate the 2005 mortality curve above the highest age in the data, and two different methods of doing this were considered. It was also necessary to make an assumption about future mortality improvements at ages above the highest age in the data, and two different assumptions were again considered to give an indication of the range of potential outcomes. Some comparisons were presented of the annuity values against previously published projections, starting from the 2005 mortality curves we fitted. In most scenarios, the model of this thesis gave higher annuity values than the previously published projections for lives aged 65 and 75 in 2005 in CMI data, but lower values for England and Wales data for both males and females. The model of this thesis generally gave higher future mortality improvements for each generation than the previously published projections at ages below the highest age of the data, as evidenced by the trend for the annuity values from the model of this thesis to

increase relative to the values from the previously published projections as the age in 2005 decreased below 65.

In Chapter 5 we also applied the model to ICA capital calculations for deferred and immediate annuities. The ICA capital values we arrived at were significantly lower than those given by the 'rule of thumb' of taking the 95<sup>th</sup> percentile of the distribution of the annuity values over the full outstanding lifetime of the annuitants, particularly for groups of annuitants with a long future life expectancy. This possibly justifies the use of a lower percentile than the 95<sup>th</sup> percentile, but we listed three factors an insurance company should consider before deciding to use a lower percentile. We also calculated diversification benefits in terms of ICA capital for an insurer with a portfolio of annuitants of different ages rather than a single age, and we found that the benefits were most significant for CMI data and least significant for England and Wales male data.

A number of possible topics for further research can be identified:

- Further investigation into the cause of the 1919-20 discontinuity in the gamma parameters would be beneficial. We followed Renshaw and Haberman (2006) in attributing it to the 1919 influenza epidemic but it is not clear that this would have led to reduced mortality for lives born in 1919 and to increased mortality for lives born in 1920, rather than affecting mortality in some other way.
- It would be valuable to develop mortality models which place cohort effects on an equal footing with period effects and which therefore provide an unbiased assessment of whether particular features of the data are consequences of period or cohort effects. Once such a model has been developed, the problems referred to in Chapter 3 with estimating both the period and cohort parameters in a single iterative procedure can be expected to be resolved. In developing such a model, it would be necessary to avoid the convergence and robustness issues that have been identified with the Renshaw and Haberman (2006) model. One possible line of investigation would be to impose smoothness of the age, period and/or cohort parameters of the Renshaw and Haberman model using P-splines, to avoid differences between successive parameter estimates which the data do not provide statistically significant evidence to justify. It could be argued that the age parameters are the ones for which the use of P-splines is most appropriate, because of the strong prior belief that mortality should be an increasing function of age rather than fluctuating from age to age and because time series methods are then still available for forecasting future period and cohort effects.
- It would be valuable to calculate stochastic projections of future mortality rates taking account of both stochastic fluctuations and parameter risk, and hence to quantify the likely financial impact of parameter estimation errors. Only stochastic fluctuations have been considered in this thesis.
- It would be valuable to develop mortality models incorporating shocks that occur in a particular year or for lives born in a particular year, in order to allow for features such as the 1919 influenza epidemic without large residuals being required.
- Developing a Bayesian procedure whereby the estimates of the gamma parameters, and of the parameters governing both the kappa and gamma time series processes, are updated as new data become available would be of value. Simulation from the joint posterior distribution would be carried out by Markov chain Monte Carlo, as in Czado *et al.* (2005).

# **References**

Baxter S., Should projections of mortality improvements be subject to a minimum value? Presentation given to Institute of Actuaries on 26 February 2007.

Brouhns N., Denuit M., Vermunt J. K. (2002), A Poisson log-bilinear regression approach to the construction of projected lifetables, *Insurance: Mathematics and Economics*, **31**, 373-393. Cairns A. J. G. (2000), A discussion of parameter and model uncertainty in insurance, *Insurance: Mathematics and Economics*, **27**, 313-330.

Cairns A. J. G., Blake D., Dowd K. (2006), A two-factor model for stochastic mortality with parameter uncertainty: theory and calibration, *Journal of Risk and Insurance*, **73**, 687-718. Cairns A. J. G., Blake D., Dowd K., Coughlan G. D., Epstein D., Khalaf-Allah M. (2008), Mortality density forecasts: an analysis of six stochastic mortality models, Pensions Institute Discussion Paper PI-0801.

Cairns A. J. G., Blake D., Dowd K., Coughlan G. D., Ong A., Epstein D., Balevich I. (2007), A quantitative comparison of stochastic mortality models using data from England & Wales and the United States, Pensions Institute Discussion Paper PI-0701.

CMI (1999), Standard tables of mortality based on the 1991-94 experiences, *Continuous Mortality Investigation Reports*, **17**, 1-228.

CMI (2002), An interim basis for adjusting the '92' series mortality projections for cohort effects, Working Paper 1.

CMI (2005), Projecting future mortality: Towards a proposal for a stochastic model, Working Paper 15.

CMI (2006), Stochastic projection methodologies: further progress and P-spline model features, example results and implications, Working Paper 20.

CMI (2007), Stochastic projection methodologies: Lee-Carter model features, example results and implications, Working Paper 25.

Currie I. D., Durban M., Eilers P. H. C. (2004), Smoothing and forecasting mortality rates, *Statistical Modelling*, **4**, 279-298.

Czado C., Delwarde A., Denuit M. (2005), Bayesian Poisson log-bilinear mortality projections, *Insurance: Mathematics and Economics*, **36**, 260-284.

de Boor C. (2001), A practical guide to splines: revised edition, Springer.

Forfar D. O., McCutcheon J. J., Wilkie A. D. (1988), On graduation by mathematical formula, *Journal of the Institute of Actuaries*, **115**, 1-149.

GAD (2006), National population projections 2004-based, Series PP2 No. 25.

Lee R. D., Carter L. R. (1992), Modeling and forecasting U.S. mortality, *Journal of the American Statistical Association*, **87**, 659-675.

LifeMetrics (2007), A toolkit for measuring and managing longevity and mortality risks, LifeMetrics Technical Document, Pension Advisory Group, JP Morgan.

Renshaw A. E., Haberman S. (2006), A cohort-based extension to the Lee-Carter model for mortality reduction factors, *Insurance: Mathematics and Economics*, **38**, 556-570.

Willets R. C. (2004), The cohort effect: insights and explanations, *British Actuarial Journal*, **10**, 833-877.

Willets R. C., Gallop A. P., Leandro P. A., Lu J. L. C., Macdonald A. S., Miller K. A., Richards S. J., Robjohns N., Ryan J. P., Waters H. R. (2004), Longevity in the 21st century, *British Actuarial Journal*, **10**, 685-832.

# Appendix A: Iterative scheme for estimating the parameters of Gompertz-Makeham models

We shall describe the iterative scheme by reference to the GM(2,2) case. No new principles are involved in fitting the GM(r,s) model for any other values of r and s.

The log-likelihood function is:

$$\ell = c - \sum_{x} \sum_{t} E_{xt} \{ \kappa_{t}^{(0)} + \kappa_{t}^{(1)} (x - \overline{x}) + \exp[\kappa_{t}^{(3)} + \kappa_{t}^{(4)} (x - \overline{x})] \}$$
  
+ 
$$\sum_{x} \sum_{t} D_{xt} \log \{ \kappa_{t}^{(0)} + \kappa_{t}^{(1)} (x - \overline{x}) + \exp[\kappa_{t}^{(3)} + \kappa_{t}^{(4)} (x - \overline{x})] \},$$

where c is a constant.

The relevant partial derivatives of the log-likelihood are:

$$\begin{split} &\frac{\partial\ell}{\partial\kappa_{i}^{(0)}} = -\sum_{x} E_{xi} + \sum_{x} D_{xi} \{\kappa_{i}^{(0)} + \kappa_{i}^{(1)}(x-\bar{x}) + \exp[\kappa_{i}^{(3)} + \kappa_{i}^{(4)}(x-\bar{x})]\}^{-1}, \\ &\frac{\partial^{2}\ell}{\partial(\kappa_{i}^{(0)})^{2}} = -\sum_{x} D_{xi} \{\kappa_{i}^{(0)} + \kappa_{i}^{(1)}(x-\bar{x}) + \exp[\kappa_{i}^{(3)} + \kappa_{i}^{(4)}(x-\bar{x})]\}^{-2}, \\ &\frac{\partial\ell}{\partial\kappa_{i}^{(1)}} = -\sum_{x} E_{xi}(x-\bar{x}) + \sum_{x} D_{xi}(x-\bar{x}) \{\kappa_{i}^{(0)} + \kappa_{i}^{(1)}(x-\bar{x}) + \exp[\kappa_{i}^{(3)} + \kappa_{i}^{(4)}(x-\bar{x})]\}^{-1}, \\ &\frac{\partial^{2}\ell}{\partial(\kappa_{i}^{(1)})^{2}} = -\sum_{x} D_{xi}(x-\bar{x})^{2} \{\kappa_{i}^{(0)} + \kappa_{i}^{(1)}(x-\bar{x}) + \exp[\kappa_{i}^{(3)} + \kappa_{i}^{(4)}(x-\bar{x})]\}^{-2}, \\ &\frac{\partial\ell}{\partial(\kappa_{i}^{(1)})^{2}} = -\sum_{x} E_{xi} \exp[\kappa_{i}^{(3)} + \kappa_{i}^{(4)}(x-\bar{x})] \\ &+ \sum_{x} D_{xi} \exp[\kappa_{i}^{(3)} + \kappa_{i}^{(4)}(x-\bar{x})] \{\kappa_{i}^{(0)} + \kappa_{i}^{(1)}(x-\bar{x}) + \exp[\kappa_{i}^{(3)} + \kappa_{i}^{(4)}(x-\bar{x})]\}^{-1}, \\ &\frac{\partial^{2}\ell}{\partial(\kappa_{i}^{(3)})^{2}} = -\sum_{x} E_{xi} \exp[\kappa_{i}^{(3)} + \kappa_{i}^{(4)}(x-\bar{x})] \\ &+ \sum_{x} D_{xi} \exp[\kappa_{i}^{(3)} + \kappa_{i}^{(4)}(x-\bar{x})] \{\kappa_{i}^{(0)} + \kappa_{i}^{(1)}(x-\bar{x}) + \exp[\kappa_{i}^{(3)} + \kappa_{i}^{(4)}(x-\bar{x})]\}^{-1}, \\ &\frac{\partial^{2}\ell}{\partial(\kappa_{i}^{(4)})^{2}} = -\sum_{x} E_{xi} \exp[\kappa_{i}^{(3)} + \kappa_{i}^{(4)}(x-\bar{x})] \{\kappa_{i}^{(0)} + \kappa_{i}^{(1)}(x-\bar{x}) + \exp[\kappa_{i}^{(3)} + \kappa_{i}^{(4)}(x-\bar{x})]\}^{-2}, \\ &\frac{\partial\ell}{\partial\kappa_{i}^{(4)}} = -\sum_{x} E_{xi}(x-\bar{x})\exp[\kappa_{i}^{(3)} + \kappa_{i}^{(4)}(x-\bar{x})] \{\kappa_{i}^{(0)} + \kappa_{i}^{(1)}(x-\bar{x}) + \exp[\kappa_{i}^{(3)} + \kappa_{i}^{(4)}(x-\bar{x})]\}^{-1}, \\ &\frac{\partial^{2}\ell}{\partial(\kappa_{i}^{(4)})^{2}} = -\sum_{x} E_{xi}(x-\bar{x})^{2}\exp[\kappa_{i}^{(3)} + \kappa_{i}^{(4)}(x-\bar{x})] \{\kappa_{i}^{(0)} + \kappa_{i}^{(1)}(x-\bar{x}) + \exp[\kappa_{i}^{(3)} + \kappa_{i}^{(4)}(x-\bar{x})]\}^{-1}, \\ &\frac{\partial^{2}\ell}{\partial(\kappa_{i}^{(4)})^{2}} = -\sum_{x} E_{xi}(x-\bar{x})^{2}\exp[\kappa_{i}^{(3)} + \kappa_{i}^{(4)}(x-\bar{x})] \{\kappa_{i}^{(0)} + \kappa_{i}^{(1)}(x-\bar{x}) + \exp[\kappa_{i}^{(3)} + \kappa_{i}^{(4)}(x-\bar{x})]\}^{-1}, \\ &\frac{\partial^{2}\ell}{\partial(\kappa_{i}^{(4)})^{2}} = -\sum_{x} E_{xi}(x-\bar{x})^{2}\exp[\kappa_{i}^{(3)} + \kappa_{i}^{(4)}(x-\bar{x})] \{\kappa_{i}^{(0)} + \kappa_{i}^{(1)}(x-\bar{x}) + \exp[\kappa_{i}^{(3)} + \kappa_{i}^{(4)}(x-\bar{x})]\}^{-1}, \\ &\sum_{x} D_{xi}(x-\bar{x})^{2}\exp[\kappa_{i}^{(3)} + \kappa_{i}^{(4)}(x-\bar{x})] \{\kappa_{i}^{(0)} + \kappa_{i}^{(1)}(x-\bar{x}) + \exp[\kappa_{i}^{(3)} + \kappa_{i}^{(4)}(x-\bar{x})]\}^{-1}, \\ &\sum_{x} D_{xi}(x-\bar{x})^{2}\exp[2\kappa_{i}^$$

Suppose that 4(*n*-1) iterations have been completed, where *n* is a positive integer, and that the estimates of the kappa parameters so obtained are  $\kappa_t^{(i)} = \alpha_t^{(i)}$ . Note that the  $\alpha_t^{(i)}$  depend on *n*. The revised estimates of the kappa parameters after the  $(4n-3)^{\text{th}}$  iteration,  $\kappa_t^{(i)} = \beta_t^{(i)}$ , are then defined as follows, the partial derivatives being evaluated at  $\kappa_t^{(i)} = \alpha_t^{(i)}$ :

$$\beta_{t}^{(0)} = \alpha_{t}^{(0)} - \frac{\frac{\partial \ell}{\partial \kappa_{t}^{(0)}}}{\frac{\partial^{2} \ell}{\partial (\kappa_{t}^{(0)})^{2}}},$$
$$\beta_{t}^{(1)} = \alpha_{t}^{(1)},$$
$$\beta_{t}^{(3)} = \alpha_{t}^{(3)},$$
$$\beta_{t}^{(4)} = \alpha_{t}^{(4)}.$$

The revised estimates of the kappa parameters after the (4*n*-2)<sup>th</sup> iteration,  $\kappa_t^{(i)} = \gamma_t^{(i)}$ , are defined as follows, the partial derivatives being evaluated at  $\kappa_t^{(i)} = \beta_t^{(i)}$ :

$$\begin{split} \boldsymbol{\gamma}_{t}^{(0)} &= \boldsymbol{\beta}_{t}^{(0)}, \\ \boldsymbol{\gamma}_{t}^{(1)} &= \boldsymbol{\beta}_{t}^{(1)} - \frac{\frac{\partial \ell}{\partial \boldsymbol{\kappa}_{t}^{(1)}}}{\frac{\partial^{2} \ell}{\partial (\boldsymbol{\kappa}_{t}^{(1)})^{2}}} \\ \boldsymbol{\gamma}_{t}^{(3)} &= \boldsymbol{\beta}_{t}^{(3)}, \\ \boldsymbol{\gamma}_{t}^{(4)} &= \boldsymbol{\beta}_{t}^{(4)}. \end{split}$$

The revised estimates of the kappa parameters after the (4*n*-1)<sup>th</sup> iteration,  $\kappa_t^{(i)} = \delta_t^{(i)}$ , are defined as follows, the partial derivatives being evaluated at  $\kappa_t^{(i)} = \gamma_t^{(i)}$ :

$$\begin{split} & \boldsymbol{\delta}_{t}^{(0)} = \boldsymbol{\gamma}_{t}^{(0)}, \\ & \boldsymbol{\delta}_{t}^{(1)} = \boldsymbol{\gamma}_{t}^{(1)}, \\ & \boldsymbol{\delta}_{t}^{(3)} = \boldsymbol{\gamma}_{t}^{(3)} - \frac{\frac{\partial \ell}{\partial \boldsymbol{\kappa}_{t}^{(3)}}}{\frac{\partial^{2} \ell}{\partial (\boldsymbol{\kappa}_{t}^{(3)})^{2}}}, \\ & \boldsymbol{\delta}_{t}^{(4)} = \boldsymbol{\gamma}_{t}^{(4)}. \end{split}$$

The revised estimates of the kappa parameters after the  $(4n)^{\text{th}}$  iteration,  $\kappa_t^{(i)} = \varepsilon_t^{(i)}$ , are defined as follows, the partial derivatives being evaluated at  $\kappa_t^{(i)} = \delta_t^{(i)}$ :

$$\varepsilon_t^{(0)} = \delta_t^{(0)},$$

$$\varepsilon_t^{(1)} = \delta_t^{(1)},$$

$$\varepsilon_t^{(3)} = \delta_t^{(3)},$$

$$\varepsilon_t^{(4)} = \delta_t^{(4)} - \frac{\frac{\partial \ell}{\partial \kappa_t^{(4)}}}{\frac{\partial^2 \ell}{\partial (\kappa_t^{(4)})^2}}.$$

The criteria used to determine when to stop the iterations are described in the main text.

# Appendix B: Investigations into estimating both the kappa and gamma parameters by a single iterative procedure

Under the model we are considering, he force of mortality at age x in calendar year t is:

$$\mu_{xt} = \gamma_{t-x} \{ \kappa_t^{(0)} + \exp[\kappa_t^{(3)} + \kappa_t^{(4)}(x-\overline{x}) + \kappa_t^{(5)}((x-\overline{x})^2 - \hat{\sigma}_x^2)] \}$$

The log-likelihood function is:

$$\ell = a - \sum_{x} \sum_{t} E_{xt} \gamma_{t-x} \{ \kappa_{t}^{(0)} + \exp[\kappa_{t}^{(3)} + \kappa_{t}^{(4)}(x - \overline{x}) + \kappa_{t}^{(5)}((x - \overline{x})^{2} - \hat{\sigma}_{x}^{2})] \}$$
  
+ 
$$\sum_{x} \sum_{t} D_{xt} \log \gamma_{t-x} + \sum_{x} \sum_{t} D_{xt} \log \{ \kappa_{t}^{(0)} + \exp[\kappa_{t}^{(3)} + \kappa_{t}^{(4)}(x - \overline{x}) + \kappa_{t}^{(5)}((x - \overline{x})^{2} - \hat{\sigma}_{x}^{2})] \}$$

where a is a constant.

We shall consider two different iterative procedures for maximising this log-likelihood function, which we shall refer to as Methods A and B. In both procedures, the starting values of the kappa parameters were those estimated from the GM(1,3) model before introducing gamma parameters and the starting values of the gamma parameters were all set to 1. We shall show the results of applying Methods A and B to England and Wales male data only – similar conclusions were reached for the other two data sets. In the England and Wales male data set we consider, years of birth observed for less than five years in the data have been excluded, as was done in Chapter 3.

In both Method A and Method B, each time the gamma parameters were updated, it was done by the Newton-Raphson method, leaving the kappa parameters constant. Similarly, each time one of the series of kappa parameters was updated, it was done by the Newton-Raphson method, leaving the other series of kappa parameters and the gamma parameters constant. Thus the procedure was analogous to that described in Appendix A. The difference between Methods A and B was in the order in which the different series were updated. In Method A, between each update of the gamma parameters, all the kappa series were updated once. In Method B, the kappa series were updated repeatedly until a complete loop of updating the kappa series, from one step of updating the  $\kappa^{(0)}$  parameters to another, changed none of the kappa parameters by more than 10<sup>-6</sup>. Method A was stopped when a complete loop from one step of updating the gamma parameters to another changed none of the kappa or gamma parameters by more than 10<sup>-6</sup>. Method B was found to be slower to converge and was stopped when such a loop changed none of the kappa or gamma parameters by more than  $10^{-4}$ . Thus it is questionable whether the iterations have been allowed to run for long enough in the implementation of Method B. However, we know that we started the iterations with reasonable values of the parameters, as the starting values were the parameter estimates under the GM(1,3) model before introducing gamma parameters. Therefore, if we find that Method B fails to produce sensible parameter values by the time none of the kappa or gamma parameters change by more than 10<sup>-4</sup>, then it is unlikely that it will produce sensible parameter values if the iterations are continued any further.

Figures B.1 and B.2 show graphs of the parameter estimates under Methods A and B respectively. These figures should be compared with Figure 3.2 for the kappa parameters and with Figure 3.7(ii) for the gamma parameters. We see that there are issues with the fitted gamma parameters in that, in both Figure B.1(v) and Figure B.2(v), the gamma parameters at the extremities of the range of years of birth are over three times greater than at the centre of the range. It does not seem plausible that such variations in the gamma parameters are serving to widen the range of shapes of the mortality curve as a function of age available in each calendar year, which was previously restricted by the GM(1,3) structure.

At the earliest years of birth, the lives concerned only appear in the data towards the top of the age range and in the earliest calendar years of the data set. Both Figure B.1 and B.2

show low values of the  $\kappa^{(4)}$  and  $\kappa^{(5)}$  parameters in these early years. Thus the high values of the gamma parameters are being compensated for by these low  $\kappa^{(4)}$  and  $\kappa^{(5)}$  parameters to give mortality rates of the correct magnitude.

At the latest years of birth, the lives concerned only appear in the data towards the bottom of the age range and in the latest calendar years of the data set. Both Figure B.1 and Figure B.2 show low values of the  $\kappa^{(0)}$  and  $\kappa^{(3)}$  parameters in these later years. Thus the high values of the gamma parameters are being compensated for by these low  $\kappa^{(0)}$  and  $\kappa^{(3)}$  parameters to give mortality rates of the correct magnitude.

For intermediate years of birth, the gamma parameters are significantly less than 1. Mortality of the correct magnitude is achieved via high values of the  $\kappa^{(0)}$  and  $\kappa^{(3)}$  parameters in the early years, when these lives are towards the bottom of the age range, and via high values of the  $\kappa^{(4)}$  and  $\kappa^{(5)}$  parameters in the later years, when these lives are towards the top of the age range.

Thus the reasons for the shapes of the graphs in Figures B.1 and B.2 have little to do with the underlying factors affecting mortality and naïve extrapolation of these graphs to produce future projections is unlikely to give sensible results. We would be more likely to obtain sensible results if we were adding a cohort effect to a model that already allowed for a wider range of shapes of the mortality curve as a function of age, as introducing the cohort effect would then no longer be indirectly performing the function of extending this range.

We should comment on the differences between Figures B.1 and B.2. We might have expected that both figures would be identical because both iterations should have converged to the same solution. However, we mentioned earlier that the iterations for Method B were stopped as soon as none of the parameters moved by more than 10<sup>-4</sup>, which may have been too early to obtain an accurate solution from Method B. An alternative possibility is that the likelihood function has multiple maxima, but we do not have sufficient evidence to reach this conclusion. There is no reason to doubt the conclusion that it is not appropriate to calculate future projections by extrapolating parameter graphs obtained by either Method A or Method B.

**Figure B.1** – Maximum likelihood parameter estimates for the GM(1,3) model extended to incorporate a cohort effect with the kappa parameters re-estimated by Method A – England and Wales male data –  $\mu_{xt} = \gamma_{t-x} \{ \kappa_t^{(0)} + \exp[\kappa_t^{(3)} + \kappa_t^{(4)}(x-\overline{x}) + \kappa_t^{(5)}((x-\overline{x})^2 - \hat{\sigma}_x^2)] \}$  – (i)  $\kappa_t^{(0)}$ , (ii)  $\kappa_t^{(3)}$ , (iii)  $\kappa_t^{(4)}$ , (iv)  $\kappa_t^{(5)}$ , (v)  $\gamma_c$ 



198

**Figure B.2** – Maximum likelihood parameter estimates for the GM(1,3) model extended to incorporate a cohort effect with the kappa parameters re-estimated by Method B – England and Wales male data –  $\mu_{xt} = \gamma_{t-x} \{ \kappa_t^{(0)} + \exp[\kappa_t^{(3)} + \kappa_t^{(4)}(x-\overline{x}) + \kappa_t^{(5)}((x-\overline{x})^2 - \hat{\sigma}_x^2)] \}$  – (i)  $\kappa_t^{(0)}$ , (ii)  $\kappa_t^{(3)}$ , (iii)  $\kappa_t^{(4)}$ , (iv)  $\kappa_t^{(5)}$ , (v)  $\gamma_c$ 



### Appendix C: Time series analysis

### C.1 First-order autoregressive and moving average processes

A time series process  $\{X_t\}$  is defined to be stationary if the distribution of  $X_t$  is independent of t.

The *autocorrelation at lag* k of a stationary time series process  $\{X_t\}$  is the coefficient of correlation between  $X_t$  and  $X_{t-k}$  as *t* varies,  $\rho(X_t, X_{t-k})$ .

As stated in Section 4.1, a first-order autoregressive process, or AR(1) process,  $\{X_t\}$  is defined by:

$$X_t = \mu + \alpha (X_{t-1} - \mu) + e_t,$$

where the  $e_t$  are independent N(0, $\sigma^2$ ) random variables and  $\alpha$ ,  $\mu$  and  $\sigma^2$  are parameters to be estimated, with  $|\alpha| < 1$ . The condition  $|\alpha| < 1$  is imposed to ensure that the process is stationary. If this condition is not imposed, then the absolute values of the  $X_t$  will in general increase without limit as *t* increases.

Let  $\{X_t\}$  be an AR(1) process, and let *k* be a positive integer. Then for each *t*, by repeated substitution of the defining equation into itself:

$$X_{t} - \mu = \alpha^{k} (X_{t-k} - \mu) + \sum_{i=0}^{k-1} \alpha^{i} e_{t-i}.$$

Taking covariances of both sides with  $X_{t-k}$  and observing that the  $e_u$  for u > t - k are uncorrelated with  $X_{t-k}$ , we see that the covariance between  $X_t$  and  $X_{t-k}$  is:

$$\operatorname{cov}(X_{t,},X_{t-k}) = \alpha^k \operatorname{var}(X_{t-k}).$$

The autocorrelation of  $\{X_t\}$  at lag k is:

$$\rho(X_t, X_{t-k}) = \frac{\operatorname{cov}(X_t, X_{t-k})}{\sqrt{\operatorname{var}(X_t)\operatorname{var}(X_{t-k})}} = \frac{\alpha^k \operatorname{var}(X_{t-k})}{\sqrt{\operatorname{var}(X_t)\operatorname{var}(X_{t-k})}}.$$

However, as the process  $\{X_t\}$  is stationary,  $var(X_{t-k})$  is independent of *k*. We therefore conclude that:

$$\rho(X_t, X_{t-k}) = \alpha^k.$$

Thus if  $0 < \alpha < 1$ , then the autocorrelation function of  $\{X_t\}$  is positive at all lags and decays exponentially. This is consistent with the remark we made in Section 4.2 that a positive decaying autocorrelation function is characteristic of an AR(1) process. If  $-1 < \alpha < 0$ , then the autocorrelation function of  $\{X_t\}$  alternates between positive and negative values, with the magnitude of the values decaying exponentially.

We also stated in Section 4.1 that a *first-order moving average process*, or MA(1) process,  $\{X_t\}$  is defined by:

$$X_t = \mu + e_t + \beta e_{t-1},$$

where the  $e_t$  are independent N(0, $\sigma^2$ ) random variables and  $\beta$ ,  $\mu$  and  $\sigma^2$  are parameters to be estimated. It is clear from the definition that {*X*<sub>*t*</sub>} is stationary.

Let  $\{X_t\}$  be an MA(1) process. The variance of  $X_t$  is:

$$var(X_t) = var(\mu + e_t + \beta e_{t-1})$$
  
= var(e\_t) + \beta^2 var(e\_{t-1}), since the e\_t are independent  
= (1 + \beta^2)\sigma^2.

The covariance between  $X_t$  and  $X_{t-1}$  is:

$$cov(X_{t}, X_{t-1}) = cov(\mu + e_t + \beta e_{t-1}, \mu + e_{t-1} + \beta e_{t-2})$$
  
=  $\beta var(e_{t-1})$ , since the  $e_t$  are independent  
=  $\beta \sigma^2$ .

The autocorrelation of  $\{X_t\}$  at lag 1 is therefore:

$$\rho(X_t, X_{t-1}) = \frac{\beta \sigma^2}{\sqrt{(1+\beta^2)\sigma^2(1+\beta^2)\sigma^2}} = \frac{\beta}{1+\beta^2}.$$

For k > 1, the covariance between  $X_t$  and  $X_{t-k}$  is:

 $\begin{array}{ll} \operatorname{cov}(X_{t,}X_{t-k}) &= \operatorname{cov}(\mu + e_t + \beta e_{t-1}, \mu + e_{t-k} + \beta e_{t-k-1}) \\ &= 0, \text{ since the } e_t \text{ are independent.} \end{array}$ 

The autocorrelation of  $\{X_t\}$  at lag k is therefore also zero.

These autocorrelations are consistent with the remark we made in Section 4.3 that a characteristic feature of an MA(1) process is an autocorrelation at lag 1 which does not persist at subsequent lags.

#### C.2 Sample autocorrelation coefficient

Suppose we have N ordered observations,  $x_1, \ldots, x_N$ , from a time series process  $\{X_t\}$ . Let:

$$\overline{x} = \frac{1}{N} \sum_{i=1}^{N} x_i.$$

For j = 1,...,N - 1, the sample autocorrelation coefficient at lag j,  $r_j$ , is defined as:

$$r_{j} = \frac{\frac{1}{N-j} \sum_{i=1}^{N-j} (x_{i} - \overline{x})(x_{i+j} - \overline{x})}{\frac{1}{N} \sum_{i=1}^{N} (x_{i} - \overline{x})^{2}}.$$

It can be shown that if  $\{X_t\}$  is a pure white noise process, *i.e.* the  $X_t$  are independent identically distributed normal random variables, then  $r_j$  is approximately normally distributed with mean 0 and variance 1 / N, for all j. We use this result to assess the significance of autocorrelations when fitting time series models in Chapter 4.

#### **C.3 Prediction intervals**

Suppose that the values  $X_u = x_u$  for  $u \le t$  of an AR(1) process  $\{X_t\}$  are known. We shall construct a prediction interval for the values  $X_{t+k}$  for k > 0. This prediction interval is used in Section 5.2.

We have:

$$X_{t+k} - \mu = \alpha^{k} (x_{t} - \mu) + \sum_{i=0}^{k-1} \alpha^{i} e_{t+k-i}.$$

Thus  $X_{t+k}$  is normally distributed with mean  $\mu + \alpha^{k}(x_{t} - \mu)$  and variance:

$$\operatorname{var}\left(\sum_{i=0}^{k-1} \alpha^{i} e_{t+k-i}\right) = \sum_{i=0}^{k-1} \alpha^{2i} \operatorname{var}(e_{t+k-i}) = \sigma^{2} \sum_{i=0}^{k-1} \alpha^{2i} = \frac{\sigma^{2}(1-\alpha^{2k})}{1-\alpha^{2}}.$$

A 95% prediction interval for  $X_{t+k}$ , based on the 2.5<sup>th</sup> and 97.5<sup>th</sup> percentiles of the distribution of  $X_{t+k}$ , is therefore:

$$\begin{bmatrix} \mu + \alpha^{k} (x_{t} - \mu) - \Phi^{-1}(0.975) \sqrt{\frac{\sigma^{2}(1 - \alpha^{2k})}{1 - \alpha^{2}}}, \\ \mu + \alpha^{k} (x_{t} - \mu) + \Phi^{-1}(0.975) \sqrt{\frac{\sigma^{2}(1 - \alpha^{2k})}{1 - \alpha^{2}}} \end{bmatrix},$$

where  $\Phi$  is the N(0,1) distribution function.

In Chapter 5 we also use a prediction interval for future values of an ARIMA(0,1,1) process, *i.e.* a process  $\{X_t\}$  such that  $\{X_t - X_{t-1}\}$  is an MA(1) process. To this end, let  $\{X_t\}$  be an ARIMA(0,1,1) process and suppose the values  $X_u = x_u$  for  $u \le t$  are known. For each positive integer *n*, we have:

$$X_{t+n} - X_{t+n-1} = \mu + e_{t+n} + \beta e_{t+n-1}$$

where the  $e_u$  are independent N(0, $\sigma^2$ ) random variables and  $\beta$ ,  $\mu$  and  $\sigma^2$  are parameters. The value of  $e_t$  is assumed to be known.

If *k* is a fixed positive integer, then summing the above equations for  $n \le k$  gives:

$$X_{t+k} = x_t + k\mu + e_{t+k} + (1+\beta) \sum_{i=1}^{k-1} e_{t+k-i} + \beta e_t.$$

Thus  $X_{t+k}$  is normally distributed with mean  $x_t + k\mu + \beta e_t$  and variance:

$$\operatorname{var}\left(e_{t+k} + (1+\beta)\sum_{i=1}^{k-1} e_{t+k-i}\right) = \operatorname{var}(e_{t+k}) + (1+\beta)^{2}\sum_{i=1}^{k-1} \operatorname{var}(e_{t+k-i}) = \sigma^{2}(1+(k-1)(1+\beta)^{2}).$$

A 95% prediction interval for  $X_{t+k}$ , based on the 2.5<sup>th</sup> and 97.5<sup>th</sup> percentiles of the distribution of  $X_{t+k}$ , is therefore:

$$[x_t + k\mu + \beta e_t - \Phi^{-1}(0.975)\sqrt{\sigma^2(1 + (k-1)(1+\beta)^2)}, x_t + k\mu + \beta e_t + \Phi^{-1}(0.975)\sqrt{\sigma^2(1 + (k-1)(1+\beta)^2)}]$$

### Appendix D: The value of an annuity after one year

As per Section 5.6, consider an annuitant who had just attained the age label x at the start of calendar year t, where calendar year t - 1 is the last year of the data, and has now survived to the end of calendar year t. The (random) value of the liabilities in respect of this annuitant, L(t + 1), is:

$$\begin{split} &\sum_{i=x_{1}-x-1}^{\infty} v^{i} \exp\left(-\sum_{j=1}^{i} \mu_{x+j,t+j}\right) \\ &= \sum_{i=x_{1}-x-1}^{a-x-1} v^{i} \exp\left(-\gamma_{t-x} \sum_{j=1}^{i} \left[\kappa_{t+j}^{(0)} + \exp\left\{\kappa_{t+j}^{(3)} + \kappa_{t+j}^{(4)} \left(x+j-\overline{x}\right) + \kappa_{t+j}^{(5)} \left(\left(x+j-\overline{x}\right)^{2} - \hat{\sigma}_{x}^{2}\right)\right\}\right]\right) \\ &+ \sum_{i=a-x}^{\infty} v^{i} \exp\left(-\gamma_{t-x} \sum_{j=1}^{a-x-1} \left[\kappa_{t+j}^{(0)} + \exp\left\{\kappa_{t+j}^{(3)} + \kappa_{t+j}^{(4)} \left(x+j-\overline{x}\right) + \kappa_{t+j}^{(5)} \left(\left(x+j-\overline{x}\right)^{2} - \hat{\sigma}_{x}^{2}\right)\right)\right]\right) \\ &- \gamma_{t-x} \sum_{j=a-x}^{i} \frac{\mu_{a,t+j}^{0} \mu_{x+j,t-1}^{0}}{\mu_{a,t-1}^{0}} \end{split}$$

from the high improvement assumption, where v = 1 / 1.045 and *a* is the maximum age of the data set.  $x_1$  is the age label at which the annuity commences for a deferred annuity, and is equal to x + 1 for an immediate annuity.

Under the log-linear extrapolation method, for j > a - x,  $\frac{\mu_{x+j,t-1}^0}{\mu_{a,t-1}^0} = \exp(\lambda(x+j-a))$ , where

 $\lambda$  is determined so as to make  $\mu_{x,t-1}^0$  a differentiable function of x at x = a. Therefore:

$$\begin{split} &L(t+1) \\ &= \sum_{i=x_{1}-x-1}^{a-x-1} v^{i} \exp\left(-\gamma_{t-x} \sum_{j=1}^{i} \left[\kappa_{t+j}^{(0)} + \exp\left\{\kappa_{t+j}^{(3)} + \kappa_{t+j}^{(4)}(x+j-\bar{x}) + \kappa_{t+j}^{(5)}((x+j-\bar{x})^{2} - \hat{\sigma}_{x}^{2})\right\}\right]\right) \\ &+ \sum_{i=a-x}^{\infty} v^{i} \exp\left\{-\gamma_{t-x} \sum_{j=1}^{a-x-1} \left[\kappa_{t+j}^{(0)} + \exp\left\{\kappa_{t+j}^{(3)} + \kappa_{t+j}^{(4)}(x+j-\bar{x}) + \kappa_{t+j}^{(5)}((x+j-\bar{x})^{2} - \hat{\sigma}_{x}^{2})\right\}\right] \\ &- \gamma_{t-x} \sum_{j=a-x}^{i} \exp(\lambda(x+j-a)) \left[\kappa_{t+j}^{(0)} - \gamma_{t-x} \sum_{j=a-x}^{i} \exp\{\kappa_{t+j}^{(3)} + \kappa_{t+j}^{(4)}(a-\bar{x}) + \kappa_{t+j}^{(5)}((a-\bar{x})^{2} - \hat{\sigma}_{x}^{2})\right\}\right] \\ &= \sum_{i=x_{1}-x-1}^{\infty} v^{i} \exp\left(-\gamma_{t-x} \sum_{j=1}^{i} A(x+j,t+j)\right), \end{split}$$

where:

$$A(x+j,t+j) = \kappa_{t+j}^{(0)} + \exp\{\kappa_{t+j}^{(3)} + \kappa_{t+j}^{(4)}(x+j-\overline{x}) + \kappa_{t+j}^{(5)}((x+j-\overline{x})^2 - \hat{\sigma}_x^2)\}, j < a - x$$
  
=  $\exp(\lambda(x+j-a))[\kappa_{t+j}^{(0)} + \exp\{\kappa_{t+j}^{(3)} + \kappa_{t+j}^{(4)}(a-\overline{x}) + \kappa_{t+j}^{(5)}((a-\overline{x})^2 - \hat{\sigma}_x^2)\}], j \ge a - x.$ 

Now by repeated application of the defining equations of the time series models fitted in Chapter 4, we have that for j > 0:

$$\begin{split} \kappa_{t+j}^{(i)} &= \mu^{(i)} + (\alpha^{(i)})^{j+1} (\kappa_{t-1}^{(i)} - \mu^{(i)}) + \sigma^{(i)} \sum_{k=0}^{j-1} (\alpha^{(i)})^k Z_{t+j-k}^{(i)} + \sigma^{(i)} (\alpha^{(i)})^j z_t^{(i)}, i = 0, 4, 5, \\ \kappa_{t+j}^{(3)} &= \kappa_{t-1}^{(3)} + (j+1)\mu^{(3)} + \sigma^{(3)} Z_{t+j}^{(3)} + \sigma^{(3)} (1 + \beta^{(3)}) \sum_{k=1}^{j-1} Z_{t+j-k}^{(3)} + \sigma^{(3)} (1 + \beta^{(3)}) z_t^{(3)} \\ &+ \sigma^{(3)} \beta^{(3)} z_{t-1}^{(3)}. \end{split}$$

Note that a small letter *z* has been used for the innovations for year *t* as well as for year t - 1 here as these are assumed to be known by the end of year *t*.

Thus for j < a - x,

$$\begin{split} &A(x+j,t+j) = \mu^{(0)} + (\alpha^{(0)})^{j+1} (\kappa_{t-1}^{(0)} - \mu^{(0)}) + \sigma^{(0)} \sum_{k=0}^{j-1} (\alpha^{(0)})^{k} Z_{t+j-k}^{(0)} + \sigma^{(0)} (\alpha^{(0)})^{j} z_{t}^{(0)} \\ &= \left\{ \begin{aligned} \kappa_{t-1}^{(3)} + (j+1)\mu^{(3)} + \sigma^{(3)} Z_{t+j}^{(3)} + \sigma^{(3)} (1+\beta^{(3)}) \sum_{k=1}^{j-1} Z_{t+j-k}^{(3)} + \sigma^{(3)} (1+\beta^{(3)}) z_{t}^{(3)} \\ &+ \sigma^{(3)} \beta^{(3)} z_{t-1}^{(3)} \\ &+ \left[ \mu^{(4)} + (\alpha^{(4)})^{j+1} (\kappa_{t-1}^{(4)} - \mu^{(4)}) + \sigma^{(4)} \sum_{k=0}^{j-1} (\alpha^{(4)})^{k} Z_{t+j-k}^{(4)} \\ &+ \sigma^{(4)} (\alpha^{(4)})^{j} z_{t}^{(4)} \\ &+ \left[ \mu^{(5)} + (\alpha^{(5)})^{j+1} (\kappa_{t-1}^{(5)} - \mu^{(5)}) + \sigma^{(5)} \sum_{k=0}^{j-1} (\alpha^{(5)})^{k} Z_{t+j-k}^{(5)} \\ &+ \sigma^{(5)} (\alpha^{(5)})^{j} z_{t}^{(5)} \end{aligned} \right] \\ &= \mu^{(0)} + (\alpha^{(0)})^{j+1} (\kappa_{t-1}^{(0)} - \mu^{(0)}) + \sigma^{(0)} \sum_{k=0}^{j-1} (\alpha^{(0)})^{k} Z_{t+j-k}^{(0)} + \sigma^{(0)} (\alpha^{(0)})^{j} z_{t}^{(0)} \\ &+ [1 + \sigma^{(3)} (1 + \beta^{(3)}) z_{t}^{(3)} + \sigma^{(4)} (\alpha^{(4)})^{j} z_{t}^{(4)} (x+j-\bar{x}) \\ &+ \sigma^{(5)} (\alpha^{(5)})^{j} z_{t}^{(5)} ((x+j-\bar{x})^{2} - \hat{\sigma}_{x}^{2}) ] \end{aligned} \\ &= \exp \begin{cases} \kappa_{t-1}^{(3)} + (j+1)\mu^{(3)} + \sigma^{(3)} Z_{t+j}^{(3)} + \sigma^{(3)} (1 + \beta^{(3)}) \sum_{k=1}^{j-1} Z_{t+j-k}^{(3)} + \sigma^{(3)} \beta^{(3)} z_{t-1}^{(3)} \\ &+ \left[ \mu^{(4)} + (\alpha^{(4)})^{j+1} (\kappa_{t-1}^{(4)} - \mu^{(4)}) + \sigma^{(4)} \sum_{k=0}^{j-1} (\alpha^{(5)})^{k} Z_{t+j-k}^{(5)} \right] ((x+j-\bar{x})^{2} - \hat{\sigma}_{x}^{2}) \end{cases} \end{cases}$$

after linearisation in the  $z_t^{(l)}$ .

Define:

$$\overline{\kappa}_{t+j}^{(i)} = \mu^{(i)} + (\alpha^{(i)})^{j+1} (\kappa_{t-1}^{(i)} - \mu^{(i)}) + \sigma^{(i)} \sum_{k=0}^{j-1} (\alpha^{(i)})^k Z_{t+j-k}^{(i)}, i = 0, 4, 5,$$
  
$$\overline{\kappa}_{t+j}^{(3)} = \kappa_{t-1}^{(3)} + (j+1)\mu^{(3)} + \sigma^{(3)} Z_{t+j}^{(3)} + \sigma^{(3)} (1+\beta^{(3)}) \sum_{k=1}^{j-1} Z_{t+j-k}^{(3)} + \sigma^{(3)} \beta^{(3)} z_{t-1}^{(3)}.$$

Equivalent definitions are:

$$\overline{\kappa}_{t+j}^{(i)} = \mu^{(i)} + (\alpha^{(i)})^{j} (\overline{\kappa}_{t}^{(i)} - \mu^{(i)}) + \sigma^{(i)} \sum_{k=0}^{j-1} (\alpha^{(i)})^{k} Z_{t+j-k}^{(i)}, i = 0, 4, 5,$$
  
$$\overline{\kappa}_{t+j}^{(3)} = \overline{\kappa}_{t}^{(3)} + j\mu^{(3)} + \sigma^{(3)} Z_{t+j}^{(3)} + \sigma^{(3)} (1 + \beta^{(3)}) \sum_{k=1}^{j-1} Z_{t+j-k}^{(3)}.$$

Thus the  $\overline{\kappa}_{t+j}^{(l)}$  follow the same time series processes as the  $\kappa_{t+j}^{(l)}$ , but starting in year *t* from the  $\overline{\kappa}_{t}^{(l)}$ , as defined in Section 5.6, rather than in year *t* – 1 from the  $\kappa_{t-1}^{(l)}$ , and with the innovation term  $Z_t^{(3)}$  used in the MA(1) model set equal to zero. We then have:

$$\begin{split} A(x+j,t+j) &= \overline{\kappa}_{t+j}^{(0)} + \sigma^{(0)} (\alpha^{(0)})^{j} z_{t}^{(0)} + [1+\sigma^{(3)} (1+\beta^{(3)}) z_{t}^{(3)} \\ &+ \sigma^{(4)} (\alpha^{(4)})^{j} z_{t}^{(4)} (x+j-\overline{x}) + \sigma^{(5)} (\alpha^{(5)})^{j} z_{t}^{(5)} ((x+j-\overline{x})^{2} - \hat{\sigma}_{x}^{2})] \\ \exp \left\{ \overline{\kappa}_{t+j}^{(3)} + \overline{\kappa}_{t+j}^{(4)} (x+j-\overline{x}) + \overline{\kappa}_{t+j}^{(5)} ((x+j-\overline{x})^{2} - \hat{\sigma}_{x}^{2}) \right\} \\ &= \overline{\kappa}_{t+j}^{(0)} + \sigma^{(0)} (\alpha^{(0)})^{j} z_{t}^{(0)} + [1+\sigma^{(3)} (1+\beta^{(3)}) z_{t}^{(3)} \\ &+ \sigma^{(4)} (\alpha^{(4)})^{j} z_{t}^{(4)} (x+j-\overline{x}) + \sigma^{(5)} (\alpha^{(5)})^{j} z_{t}^{(5)} ((x+j-\overline{x})^{2} - \hat{\sigma}_{x}^{2})] \overline{\mu}_{\exp}(x+j,t+j) \\ &= \overline{\kappa}_{t+j}^{(0)} + \overline{\mu}_{\exp}(x+j,t+j) + \sigma^{(0)} (\alpha^{(0)})^{j} z_{t}^{(0)} + [\sigma^{(3)} (1+\beta^{(3)}) z_{t}^{(3)} \\ &+ \sigma^{(4)} (\alpha^{(4)})^{j} z_{t}^{(4)} (x+j-\overline{x}) + \sigma^{(5)} (\alpha^{(5)})^{j} z_{t}^{(5)} ((x+j-\overline{x})^{2} - \hat{\sigma}_{x}^{2})] \overline{\mu}_{\exp}(x+j,t+j). \end{split}$$

For  $j \ge a - x$ ,

$$A(x + j, t + j) = \exp(\lambda(x + j - a)) \times \left[ \mu^{(0)} + (\alpha^{(0)})^{j+1} (\kappa_{t-1}^{(0)} - \mu^{(0)}) + \sigma^{(0)} \sum_{k=0}^{j-1} (\alpha^{(0)})^{k} Z_{t+j-k}^{(0)} + \sigma^{(0)} (\alpha^{(0)})^{j} z_{t}^{(0)} \right] \\ + \left[ \kappa_{t-1}^{(3)} + (j+1)\mu^{(3)} + \sigma^{(3)} Z_{t+j}^{(3)} + \sigma^{(3)} (1 + \beta^{(3)}) \sum_{k=1}^{j-1} Z_{t+j-k}^{(3)} + \sigma^{(3)} (1 + \beta^{(3)}) z_{t}^{(3)} \right] \\ + \sigma^{(3)} \beta^{(3)} z_{t-1}^{(3)} \\ + \sigma^{(3)} \beta^{(3)} z_{t-1}^{(3)} \\ + \left[ \mu^{(4)} + (\alpha^{(4)})^{j+1} (\kappa_{t-1}^{(4)} - \mu^{(4)}) + \sigma^{(4)} \sum_{k=0}^{j-1} (\alpha^{(4)})^{k} Z_{t+j-k}^{(4)} \right] (a - \bar{x}) \\ + \left[ \mu^{(5)} + (\alpha^{(5)})^{j+1} (\kappa_{t-1}^{(5)} - \mu^{(5)}) + \sigma^{(5)} \sum_{k=0}^{j-1} (\alpha^{(5)})^{k} Z_{t+j-k}^{(5)} \right] ((a - \bar{x})^{2} - \hat{\sigma}_{x}^{2}) \\ + \sigma^{(5)} (\alpha^{(5)})^{j} z_{t}^{(5)} \end{array} \right]$$

$$= \exp(\lambda(x+j-\alpha)) \times \left[ \mu^{(0)} + (\alpha^{(0)})^{j+1} (\kappa^{(0)}_{t-1} - \mu^{(0)}) + \sigma^{(0)} \sum_{k=0}^{j-1} (\alpha^{(0)})^k Z^{(0)}_{t+j-k} + \sigma^{(0)} (\alpha^{(0)})^j z^{(0)}_t \right] + \frac{1}{2} \left[ \left( \alpha^{(0)} + \alpha^{(0)} \right)^{j-1} (\alpha^{(0)})^{j-1} z^{(0)}_t \right] + \frac{1}{2} \left[ \left( \alpha^{(0)} + \alpha^{(0)} \right)^{j-1} (\alpha^{(0)})^{j-1} (\alpha^{(0)})^{j-1} z^{(0)}_t \right] + \frac{1}{2} \left[ \left( \alpha^{(0)} + \alpha^{(0)} \right)^{j-1} (\alpha^{(0)})^{j-1} (\alpha^{(0)})^{j-1} z^{(0)}_t \right] + \frac{1}{2} \left[ \left( \alpha^{(0)} + \alpha^{(0)} \right)^{j-1} (\alpha^{(0)})^{j-1} (\alpha^{(0)})^{$$

$$+ \exp(\lambda(x+j-a)) \times \\ [1+\sigma^{(3)}(1+\beta^{(3)})z_{t}^{(3)} + \sigma^{(4)}(\alpha^{(4)})^{j}z_{t}^{(4)}(a-\bar{x}) + \sigma^{(5)}(\alpha^{(5)})^{j}z_{t}^{(5)}((a-\bar{x})^{2} - \hat{\sigma}_{x}^{2})] \\ = \left\{ \kappa_{t-1}^{(3)} + (j+1)\mu^{(3)} + \sigma^{(3)}Z_{t+j}^{(3)} + \sigma^{(3)}(1+\beta^{(3)})\sum_{k=1}^{j-1}Z_{t+j-k}^{(3)} + \sigma^{(3)}\beta^{(3)}z_{t-1}^{(3)} \\ + \left[ \mu^{(4)} + (\alpha^{(4)})^{j+1}(\kappa_{t-1}^{(4)} - \mu^{(4)}) + \sigma^{(4)}\sum_{k=0}^{j-1}(\alpha^{(4)})^{k}Z_{t+j-k}^{(4)} \right] (a-\bar{x}) \\ + \left[ \mu^{(5)} + (\alpha^{(5)})^{j+1}(\kappa_{t-1}^{(5)} - \mu^{(5)}) + \sigma^{(5)}\sum_{k=0}^{j-1}(\alpha^{(5)})^{k}Z_{t+j-k}^{(5)} \right] ((a-\bar{x})^{2} - \hat{\sigma}_{x}^{2}) \right\}$$

after linearisation in the  $z_t^{(l)}$ 

$$= \exp(\lambda(x+j-a)) \left[ \overline{\kappa}_{t+j}^{(0)} + \sigma^{(0)} (\alpha^{(0)})^{j} z_{t}^{(0)} \right] + \exp(\lambda(x+j-a)) \times \\ \left[ 1 + \sigma^{(3)} (1+\beta^{(3)}) z_{t}^{(3)} + \sigma^{(4)} (\alpha^{(4)})^{j} z_{t}^{(4)} (a-\overline{x}) + \sigma^{(5)} (\alpha^{(5)})^{j} z_{t}^{(5)} ((a-\overline{x})^{2} - \hat{\sigma}_{x}^{2}) \right] \\ \exp\left\{ \overline{\kappa}_{t+j}^{(3)} + \overline{\kappa}_{t+j}^{(4)} (a-\overline{x}) + \overline{\kappa}_{t+j}^{(5)} ((a-\overline{x})^{2} - \hat{\sigma}_{x}^{2}) \right\} \\ = \exp(\lambda(x+j-a)) \left[ \overline{\kappa}_{t+j}^{(0)} + \sigma^{(0)} (\alpha^{(0)})^{j} z_{t}^{(0)} \right] \\ + \left[ 1 + \sigma^{(3)} (1+\beta^{(3)}) z_{t}^{(3)} + \sigma^{(4)} (\alpha^{(4)})^{j} z_{t}^{(4)} (a-\overline{x}) + \sigma^{(5)} (\alpha^{(5)})^{j} z_{t}^{(5)} ((a-\overline{x})^{2} - \hat{\sigma}_{x}^{2}) \right] \\ \overline{\mu}_{exp}(x+j,t+j)$$

where we define, for  $j \ge a - x$  ,

$$\overline{\mu}_{\exp}(x+j,t+j) = \exp(\lambda(x+j-a))\exp\{\overline{\kappa}_{t+j}^{(3)} + \overline{\kappa}_{t+j}^{(4)}(a-\overline{x}) + \overline{\kappa}_{t+j}^{(5)}((a-\overline{x})^2 - \hat{\sigma}_x^2)\}$$

Thus:

$$\begin{aligned} A(x+j,t+j) &= \exp(\lambda(x+j-a))\overline{\kappa}_{t+j}^{(0)} + \overline{\mu}_{\exp}(x+j,t+j) \\ &+ \exp(\lambda(x+j-a))\sigma^{(0)}(\alpha^{(0)})^{j} z_{t}^{(0)} \\ &+ [\sigma^{(3)}(1+\beta^{(3)})z_{t}^{(3)} + \sigma^{(4)}(\alpha^{(4)})^{j} z_{t}^{(4)}(a-\overline{x}) + \sigma^{(5)}(\alpha^{(5)})^{j} z_{t}^{(5)}((a-\overline{x})^{2} - \hat{\sigma}_{x}^{2})] \\ \overline{\mu}_{\exp}(x+j,t+j). \end{aligned}$$

Substituting these expressions for A(x + j, t + j) into our formula for L(t + 1) gives L(t + 1) as:

$$\sum_{i=x_{1}-x-1}^{a-x-1} v^{i} \exp \left\{-\gamma_{t-x} \sum_{j=1}^{i} \left\{ \overline{\kappa}_{t+j}^{(0)} + \overline{\mu}_{exp} (x+j,t+j) + \sigma^{(0)} (\alpha^{(0)})^{j} z_{t}^{(0)} + [\sigma^{(3)} (1+\beta^{(3)}) z_{t}^{(3)} + \sigma^{(4)} (\alpha^{(4)})^{j} z_{t}^{(4)} (x+j-\overline{x}) + \sigma^{(5)} (\alpha^{(5)})^{j} z_{t}^{(5)} ((x+j-\overline{x})^{2} - \hat{\sigma}_{x}^{2})] \overline{\mu}_{exp} (x+j,t+j) \right\} \right\} +$$

$$\begin{split} &+\sum_{i=a=x}^{n} v^{i} \exp\left\{ -\frac{\left[ \overline{k}_{i=j}^{(n)} + \overline{\mu}_{axp} (x+j,t+j) + \sigma^{(0)} (\alpha^{(0)})^{j} z_{i}^{(0)} + [\sigma^{(3)} (1+\beta^{(3)}) z_{i}^{(3)} \right] + \sigma^{(4)} (\alpha^{(4)})^{j} z_{i}^{(4)} (x+j-\overline{x})}{+ \sigma^{(5)} (\alpha^{(5)})^{j} z_{i}^{(5)} ((x+j-\overline{x})^{2} - \hat{\sigma}_{x}^{2}) \right] \overline{\mu}_{axp} (x+j,t+j)} \right\} \\ &+ \frac{1}{\sigma^{(2)}} \left\{ \sum_{j=a=x}^{i} \sum_{j=x}^{i} \sum_$$

where we define, for  $j \ge a - x$ ,

$$\overline{\mu}_{x+j,t+j} = \gamma_{t-x} [\exp(\lambda(x+j-a))\overline{\kappa}_{t+j}^{(0)} + \overline{\mu}_{\exp}(x+j,t+j)].$$

Thus:

$$\begin{split} L(t+1) &= \sum_{i=k_{i}=-1}^{a-1} v^{i} \exp \left\{ -\sum_{j=l}^{i} \overline{\mu}_{x+j,i+j}^{-1} - \frac{\gamma_{r-x}\sigma^{(0)}z_{i}^{(0)}\alpha^{(0)}(1-(\alpha^{(0)})^{i})}{1-\alpha^{(0)}} - \gamma_{r-x}\sigma^{(1)}z_{i}^{(1)} \sum_{j=l}^{i} \overline{\mu}_{\exp}(x+j,t+j) \\ -\gamma_{r-x}\sigma^{(1)}z_{i}^{(2)} \sum_{j=l}^{i} \overline{\mu}_{\exp}(x+j,t+j) \\ -\gamma_{r-x}\sigma^{(1)}z_{i}^{(2)} \sum_{j=l}^{i} (\alpha^{(1)})^{j} (x+j-\overline{x})\overline{\mu}_{\exp}(x+j,t+j) \\ -\gamma_{r-x}\sigma^{(0)}z_{i}^{(0)} \alpha^{(0)} (1-(\alpha^{(0)})^{a-x-1}) \\ -\gamma_{r-x}\sigma^{(0)}z_{i}^{(0)} \alpha^{(0)} (\alpha^{(0)})^{a-x} (1-(\alpha^{(0)})^{a-x-1}) \\ -\gamma_{r-x}\sigma^{(0)}z_{i}^{(1)} (\alpha^{(1)})^{j} (x+j-\overline{x})\overline{\mu}_{\exp}(x+j,t+j) \\ -\gamma_{r-x}\sigma^{(3)} (1+\beta^{(3)})z_{i}^{(2)} \sum_{j=l}^{i} \overline{\mu}_{\exp}(x+j,t+j) \\ -\gamma_{r-x}\sigma^{(3)} (1+\beta^{(3)})z_{i}^{(3)} \sum_{j=l}^{i} \alpha^{(4)} j^{j} \overline{\mu}_{\exp}(x+j,t+j) \\ -\gamma_{r-x}\sigma^{(3)} z_{i}^{(4)} (a-\overline{x}) \sum_{j=l}^{i} (\alpha^{(4)})^{j} \overline{\mu}_{\exp}(x+j,t+j) \\ -\gamma_{r-x}\sigma^{(3)} z_{i}^{(5)} \sum_{j=l}^{a-x-1} (\alpha^{(3)})^{j} ((x+j-\overline{x})^{2}-\sigma_{x}^{2}) \overline{\mu}_{\exp}(x+j,t+j) \\ -\gamma_{r-x}\sigma^{(3)} z_{i}^{(5)} \sum_{j=l}^{a-x-1} (\alpha^{(3)})^{j} (x+j-\overline{x})^{2} - \sigma_{x}^{2}) \overline{\mu}_{\exp}(x+j,t+j) \\ + \frac{\gamma_{r-x}\sigma^{(3)} z_{i}^{(5)} (a-\overline{x})^{2} - \sigma_{x}^{2}} \sum_{j=a-x}^{a-x-1} v^{i} \exp\left(-\sum_{j=l}^{i} \overline{\mu}_{x+j,i+j}\right) \\ + \frac{\gamma_{r-x}\sigma^{(0)} z_{i}^{(0)}} \sum_{i=k_{i}-x-1}^{a-x-1} (v\alpha^{(0)})^{i} \exp\left(-\sum_{j=l}^{i} \overline{\mu}_{x+j,i+j}\right) \\ -\gamma_{r-x}\sigma^{(3)} (1+\beta^{(3)}) z_{i}^{(3)} \sum_{i=x_{i}-x-1}^{\infty} (v\alpha^{(0)}e^{A})^{i} \exp\left(-\sum_{j=l}^{i} \overline{\mu}_{x+j,i+j}\right) \\ + \frac{\gamma_{r-x}\sigma^{(0)} z_{i}^{(0)} \left(\frac{\alpha^{(0)}(1-(\alpha^{(0)})^{a-x-1})}{1-\alpha^{(0)}} + \frac{(\alpha^{(0)})^{a-x-1}}{1-\alpha^{(0)}e^{A}}\right) \sum_{i=w-x}^{\infty} v^{i} \exp\left(-\sum_{j=l}^{i} \overline{\mu}_{x+j,i+j}\right) \\ + \frac{\gamma_{r-x}\sigma^{(0)} z_{i}^{(0)} \left(\frac{\alpha^{(0)}(1-(\alpha^{(0)})^{a-x-1})}{1-\alpha^{(0)}} + \frac{(\alpha^{(0)})^{a-x-1}}{1-\alpha^{(0)}e^{A}}\right) \sum_{i=w-x}^{\infty} v^{i} \exp\left(-\sum_{j=l}^{i} \overline{\mu}_{x+j,i+j}\right) \\ -\gamma_{r-x}\sigma^{(3)} (1+\beta^{(3)}) z_{i}^{(3)} \sum_{i=x_{i}-x-1}^{\infty} v^{i} \left(\sum_{j=l}^{i} \overline{\mu}_{\exp}(x+j,t+j)\right) \exp\left(-\sum_{j=l}^{i} \overline{\mu}_{x+j,i+j}\right) \\ -\gamma_{r-x}\sigma^{(4)} z_{i}^{(4)} \sum_{i=x_{i}-x-1}^{\infty} v^{i} \left(\sum_{j=l}^{i} \overline{\mu}_{x+j,i+j}\right) - \frac{\gamma_{r-x}\sigma^{(4)} z_{i}^{(4)} \sum_{i=x_{i}-x-1}^{\infty} v^{i} \left(\sum_{j=l}^{i} \overline{\mu}_{\exp}(x+j,t+j)\right) - \frac{\gamma_{r-x}\sigma^{(4)} z_{i}^{(4)} \sum_{i=x_{i}-x-1}^{\infty} v$$

$$-\gamma_{t-x}\sigma^{(4)}z_{t}^{(4)}\sum_{i=a-x}^{\infty}v^{i} \begin{pmatrix} \sum_{j=1}^{a-x-1}(\alpha^{(4)})^{j}(x+j-\bar{x})\overline{\mu}_{\exp}(x+j,t+j) \\ +(a-\bar{x})\sum_{j=a-x}^{i}(\alpha^{(4)})^{j}\overline{\mu}_{\exp}(x+j,t+j) \end{pmatrix}$$

$$\exp\left(-\sum_{j=1}^{i}\overline{\mu}_{x+j,t+j}\right)$$

$$-\gamma_{t-x}\sigma^{(5)}z_{t}^{(5)}\sum_{i=x_{1}-x-1}^{a-x-1}v^{i}\left(\sum_{j=1}^{i}(\alpha^{(5)})^{j}((x+j-\bar{x})^{2}-\hat{\sigma}_{x}^{2})\overline{\mu}_{\exp}(x+j,t+j)\right)$$

$$-\gamma_{t-x}\sigma^{(5)}z_{t}^{(5)}\sum_{i=a-x}^{\infty}v^{i}\left(\sum_{j=1}^{a-x-1}(\alpha^{(5)})^{j}((x+j-\bar{x})^{2}-\hat{\sigma}_{x}^{2})\overline{\mu}_{\exp}(x+j,t+j)\right)$$

$$+\left((a-\bar{x})^{2}-\hat{\sigma}_{x}^{2}\sum_{j=a-x}(\alpha^{(5)})^{j}\overline{\mu}_{\exp}(x+j,t+j)\right)$$

$$\exp\left(-\sum_{j=1}^{i}\overline{\mu}_{x+j,t+j}\right)$$

after linearisation in the  $z_t^{(l)}$ .

Note that the summation terms over the index *i* are all independent of the  $z_t^{(l)}$  and have the form of annuity values. Some of the annuities have payments that vary over time, and others are valued at revised interest rates. We now introduce the following notation for the summation terms over the index *i*:

$$\begin{aligned} FutAnnVal(x,t) &= \sum_{i=x_{1}-x-1}^{\infty} v^{i} \exp\left(-\sum_{j=1}^{i} \overline{\mu}_{x+j,t+j}\right), \\ FutAnnVal\_Maxa(x,t) &= \sum_{i=x_{1}-x-1}^{a-x-1} v^{i} \exp\left(-\sum_{j=1}^{i} \overline{\mu}_{x+j,t+j}\right), \\ FutAnnVal\_Maxa\_alpha0(x,t) &= \sum_{i=x_{1}-x-1}^{a-x-1} (v\alpha^{(0)})^{i} \exp\left(-\sum_{j=1}^{i} \overline{\mu}_{x+j,t+j}\right), \\ FutAnnVal\_Mina\_alpha0\_lambda(x,t) &= \sum_{i=a-x}^{\infty} (v\alpha^{(0)}e^{\lambda})^{i} \exp\left(-\sum_{j=1}^{i} \overline{\mu}_{x+j,t+j}\right), \\ FutAnnVal\_Varying3(x,t) &= \sum_{i=x_{1}-x-1}^{\infty} v^{i} \left(\sum_{j=1}^{i} \overline{\mu}_{\exp}(x+j,t+j)\right) \exp\left(-\sum_{j=1}^{i} \overline{\mu}_{x+j,t+j}\right), \\ FutAnnVal\_Varying4(x,t) &= \sum_{i=x_{1}-x-1}^{a-x-1} v^{i} \left(\sum_{j=1}^{i} (\alpha^{(4)})^{j} (x+j-\overline{x}) \overline{\mu}_{\exp}(x+j,t+j)\right) + \\ \end{aligned}$$

$$+ \sum_{i=a-x}^{\infty} \frac{v^{i} \left( \sum_{j=1}^{a-x-1} (\alpha^{(4)})^{j} (x+j-\bar{x}) \overline{\mu}_{exp} (x+j,t+j) + (a-\bar{x}) \sum_{j=a-x}^{i} (\alpha^{(4)})^{j} \overline{\mu}_{exp} (x+j,t+j) \right)}{\exp \left( -\sum_{j=1}^{i} \overline{\mu}_{x+j,t+j} \right)} ,$$

$$FutAnnVal\_Varying5(x,t) = \sum_{i=x_{1}-x-1}^{a-x-1} \frac{v^{i} \left( \sum_{j=1}^{i} (\alpha^{(5)})^{j} ((x+j-\bar{x})^{2} - \hat{\sigma}_{x}^{2}) \overline{\mu}_{exp} (x+j,t+j) \right)}{\exp \left( -\sum_{j=1}^{i} \overline{\mu}_{x+j,t+j} \right)}$$

$$+ \sum_{i=a-x}^{\infty} \frac{v^{i} \left( \sum_{j=1}^{a-x-1} (\alpha^{(5)})^{j} ((x+j-\bar{x})^{2} - \hat{\sigma}_{x}^{2}) \overline{\mu}_{exp} (x+j,t+j) \right)}{\exp \left( -\sum_{j=1}^{i} \overline{\mu}_{x+j,t+j} \right)} \right)$$

$$\exp \left( -\sum_{j=1}^{i} \overline{\mu}_{x+j,t+j} \right)$$

Then:

$$\begin{split} L(t+1) &= FutAnnVal(x,t) - \frac{\gamma_{t-x}\sigma^{(0)}\alpha^{(0)}z_{t}^{(0)}}{1-\alpha^{(0)}}FutAnnVal\_Maxa(x,t) \\ &+ \frac{\gamma_{t-x}\sigma^{(0)}\alpha^{(0)}z_{t}^{(0)}}{1-\alpha^{(0)}}FutAnnVal\_Maxa\_alpha0(x,t) \\ &- \gamma_{t-x}\sigma^{(0)}z_{t}^{(0)} \left(\frac{\alpha^{(0)}(1-(\alpha^{(0)})^{a-x-1})}{1-\alpha^{(0)}} + \frac{(\alpha^{(0)})^{a-x}}{1-\alpha^{(0)}e^{\lambda}}\right) \\ [FutAnnVal(x,t) - FutAnnVal\_Maxa(x,t)] \\ &+ \frac{\gamma_{t-x}\sigma^{(0)}\alpha^{(0)}e^{-(a-x-1)\lambda}z_{t}^{(0)}}{1-\alpha^{(0)}e^{\lambda}}FutAnnVal\_Mina\_alpha0\_lambda(x,t) \\ &- \gamma_{t-x}\sigma^{(3)}(1+\beta^{(3)})z_{t}^{(3)}FutAnnVal\_Varying3(x,t) \\ &- \gamma_{t-x}\sigma^{(4)}z_{t}^{(4)}FutAnnVal\_Varying4(x,t) - \gamma_{t-x}\sigma^{(5)}z_{t}^{(5)}FutAnnVal\_Varying5(x,t) \\ &= FutAnnVal(x,t) + \frac{\gamma_{t-x}\sigma^{(0)}(\alpha^{(0)})^{a-x+1}(e^{\lambda}-1)z_{t}^{(0)}}{(1-\alpha^{(0)})(1-\alpha^{(0)}e^{\lambda})}FutAnnVal\_Maxa(x,t) \\ &+ \frac{\gamma_{t-x}\sigma^{(0)}\alpha^{(0)}z_{t}^{(0)}}{1-\alpha^{(0)}}FutAnnVal\_Maxa\_alpha0(x,t) - \\ &- \gamma_{t-x}\sigma^{(0)}z_{t}^{(0)}\left(\frac{\alpha^{(0)}(1-(\alpha^{(0)})^{a-x-1})}{1-\alpha^{(0)}} + \frac{(\alpha^{(0)})^{a-x}}{1-\alpha^{(0)}e^{\lambda}}\right)FutAnnVal(x,t) \\ &+ \frac{\gamma_{t-x}\sigma^{(0)}\alpha^{(0)}e^{-(a-x-1)\lambda}z_{t}^{(0)}}{1-\alpha^{(0)}e^{\lambda}}FutAnnVal\_Mina\_alpha0\_lambda(x,t) \\ &- \gamma_{t-x}\sigma^{(0)}z_{t}^{(0)}\left(\frac{\alpha^{(0)}(1-(\alpha^{(0)})^{a-x-1})}{1-\alpha^{(0)}} + \frac{(\alpha^{(0)})^{a-x}}{1-\alpha^{(0)}e^{\lambda}}\right)FutAnnVal(x,t) \\ &+ \frac{\gamma_{t-x}\sigma^{(0)}\alpha^{(0)}e^{-(a-x-1)\lambda}z_{t}^{(0)}}{1-\alpha^{(0)}e^{\lambda}}FutAnnVal\_Mina\_alpha0\_lambda(x,t) \\ &- \gamma_{t-x}\sigma^{(3)}(1+\beta^{(3)})z_{t}^{(3)}FutAnnVal\_Varying3(x,t) \\ &- \gamma_{t-x}\sigma^{(4)}z_{t}^{(4)}FutAnnVal\_Varying4(x,t) - \gamma_{t-x}\sigma^{(5)}z_{t}^{(5)}FutAnnVal\_Varying5(x,t). \end{split}$$